

Storage Automated Diagnostic Environment User's Guide

Sun Microsystems, Inc. 4150 Network Circle Santa Clara, CA 95054 U.S.A. 650-960-1300

Part No. 816-3142-10 February, 2002 Revision A

Send comments about this document to: docfeedback@sun.com

Copyright 2002 Sun Microsystems, Inc., 4150 Network Circle, Santa Clara, CA 95054 U.S.A. All rights reserved.

This product or document is distributed under licenses restricting its use, copying, distribution, and decompilation. No part of this product or document may be reproduced in any form by any means without prior written authorization of Sun and its licensors, if any. Third-party software, including font technology, is copyrighted and licensed from Sun suppliers.

Parts of the product may be derived from Berkeley BSD systems, licensed from the University of California. UNIX is a registered trademark in the U.S. and other countries, exclusively licensed through X/Open Company, Ltd.

Sun, Sun Microsystems, the Sun logo, AnswerBook2, Sun StorEdge, StorTools, and Solaris are trademarks, registered trademarks, or service marks of Sun Microsystems, Inc. in the U.S. and other countries. All SPARC trademarks are used under license and are trademarks or registered trademarks of SPARC International, Inc. in the U.S. and other countries. Products bearing SPARC trademarks are based upon an architecture developed by Sun Microsystems, Inc.

The OPEN LOOK and Sun[™] Graphical User Interface was developed by Sun Microsystems, Inc. for its users and licensees. Sun acknowledges the pioneering efforts of Xerox in researching and developing the concept of visual or graphical user interfaces for the computer industry. Sun holds a non-exclusive license from Xerox to the Xerox Graphical User Interface, which license also covers Sun's licensees who implement OPEN LOOK GUIs and otherwise comply with Sun's written license agreements.

Federal Acquisitions: Commercial Software—Government Users Subject to Standard License Terms and Conditions.

DOCUMENTATION IS PROVIDED "AS IS" AND ALL EXPRESS OR IMPLIED CONDITIONS, REPRESENTATIONS AND WARRANTIES, INCLUDING ANY IMPLIED WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE OR NON-INFRINGEMENT, ARE DISCLAIMED, EXCEPT TO THE EXTENT THAT SUCH DISCLAIMERS ARE HELD TO BE LEGALLY INVALID.

Copyright 2002 Sun Microsystems, Inc., 4150 Network Circle, Santa Clara, CA 95054 Etats-Unis. Tous droits réservés.

Ce produit ou document est distribué avec des licences qui en restreignent l'utilisation, la copie, la distribution, et la décompilation. Aucune partie de ce produit ou document ne peut être reproduite sous aucune forme, par quelque moyen que ce soit, sans l'autorisation préalable et écrite de Sun et de ses bailleurs de licence, s'il y en a. Le logiciel détenu par des tiers, et qui comprend la technologie relative aux polices de caractères, est protégé par un copyright et licencié par des fournisseurs de Sun.

Des parties de ce produit pourront être dérivées des systèmes Berkeley BSD licenciés par l'Université de Californie. UNIX est une marque déposée aux Etats-Unis et dans d'autres pays et licenciée exclusivement par X/Open Company, Ltd.

Sun, Sun Microsystems, le logo Sun, AnswerBook2, Sun StorEdge, StorTools, et Solaris sont des marques de fabrique ou des marques déposées, ou marques de service, de Sun Microsystems, Inc. aux Etats-Unis et dans d'autres pays. Toutes les marques SPARC sont utilisées sous licence et sont des marques de fabrique ou des marques déposées de SPARC International, Inc. aux Etats-Unis et dans d'autres pays. Les produits portant les marques SPARC sont basés sur une architecture développée par Sun Microsystems, Inc.

L'interface d'utilisation graphique OPEN LOOK et Sun™ a été développée par Sun Microsystems, Inc. pour ses utilisateurs et licenciés. Sun reconnaît les efforts de pionniers de Xerox pour la recherche et le développement du concept des interfaces d'utilisation visuelle ou graphique pour l'industrie de l'informatique. Sun détient une licence non exclusive de Xerox sur l'interface d'utilisation graphique Xerox, cette licence couvrant également les licenciés de Sun qui mettent en place l'interface d'utilisation graphique OPEN LOOK et qui en outre se conforment aux licences écrites de Sun.

LA DOCUMENTATION EST FOURNIE "EN L'ETAT" ET TOUTES AUTRES CONDITIONS, DECLARATIONS ET GARANTIES EXPRESSES OU TACITES SONT FORMELLEMENT EXCLUES, DANS LA MESURE AUTORISEE PAR LA LOI APPLICABLE, Y COMPRIS NOTAMMENT TOUTE GARANTIE IMPLICITE RELATIVE A LA QUALITE MARCHANDE, A L'APTITUDE A UNE UTILISATION PARTICULIERE OU A L'ABSENCE DE CONTREFAÇON.







Contents

Typographic Conventions xiii

- ▼ To Access Online Sun Documentation xvi
 - ▼ To Access Brocade documentation xvii

1. Overview 1

What is the Storage Automated Diagnostic Environment? 1

Changes from Previous Release 2

Products Not Supported 3

How the Storage Automated Diagnostic Environment Works 3

Monitoring Devices Using the Storage Automated Diagnostic Environment 4

Information Deltas 4

Storage Automated Diagnostic Environment Functions 5

Storage Automated Diagnostic Environment Agent Functionality 5

Storage Automated Diagnostic Environment Diagnostic Functionality 6

Tasks You Can Perform Using the Storage Automated Diagnostic Environment 7

2. Installing and Starting SUNWstade 9

Installation Checklist 10

Host Requirements 12

Example Configurations 12

Installing SUNWstade 13

Installation Notes 13

- ▼ To Install SUNWstade 13
- ▼ To Verify the Installation 16
- ▼ To Install a Patch 17

Sun StorEdge 3900 and 6900 Series Configuration 18

Sun StorEdge 3900 and 6900 Series Upgrades 18

Starting the Storage Automated Diagnostic Environment 21

- ▼ To Set the Environment Variables 21
- ▼ To Edit the .conf Files 22
- ▼ To Run the ras_install Script 22
 - ▼ To Start the Storage Automated Diagnostic Environment Services 23 Installing Distributed Agents (Slaves) 24

Prerequisites 24

▼ To Set Up a Slave 25

Upgrading or Removing SUNWstade 26

- ▼ To Upgrade the Storage Automated Diagnostic Environment Package 26
- ▼ To Remove the SUNWstade Package 26

The Storage Automated Diagnostic Environment Configuration File 27 Starting SUNWstade 28

 To Start the Storage Automated Diagnostic Environment 29 Product Abbreviations 30

3. Maintaining the Storage Automated Diagnostic Environment 31

▼ To Access the Maintenance Window 31

General Maintenance 35

▼ To Maintain Site Information 35

- ▼ To Maintain Hosts 36
 - ▼ To Update the Master Configuration 38
 - ▼ To Add a New Slave 40
 - ▼ To Push Slave Configs 43

Device Discovery 43

- ▼ To Access the Device Discovery Window 44
- ▼ To Discover a Sun StorEdge 3900 or 6900 Series Cabinet 45
- ▼ To Search the LAN for Sun StorEdge Devices Using the Subnet Button 46
- ▼ To Set Up Sun StorEdge T3 and T3+ Array Message Monitoring 46
- ▼ To Maintain Devices 47
 - ▼ To Manually Add a Device 48
 - ▼ To Update a Device 51
 - ▼ To Delete a Device 52

Renaming a Device 52

Customizing Email Deliveries 53

- ▼ To Set Up Local Email and Pager Notifications 53
- ▼ To Send a Customizable Subset of the Event-Driven Messages From the Host 55

Using Providers 56

- ▼ To Access the Notification Provider Maintenance Window 56
- ▼ To Set Up Providers 57

HTTP Provider 58 NSM Provider 58 NetConnect Provider 59 SRS Provider 59 SNMP Traps 60 RSS Provider 61 Push Configs 61

- ▼ To Start or Stop Agent(s) 62
 - ▼ To Access the Start/Stop Agents Window 62
 - ▼ To Deactivate Monitoring on a Device-by-Device Basis 63
- ▼ To Send Test Email 64
 - ▼ To Access the Test Email Window 64
- ▼ To Review the Configuration 65
- ▼ To Change Configuration Options 66

Topology Maintenance 67

- ▼ To Access the Topology Window 67
- ▼ To Create a Topology Snapshot 68
- ▼ To Merge Topologies 69
- To Display Topology History 70
 Grouping 71
 - ▼ To Create and Maintain Groups 71
 - ▼ To Add Devices to a Particular Group 73

4. Monitoring 75

▼ To Access the Monitor Page 75

Monitoring Devices 76

▼ To View Instrumentation Reports 76

Monitoring Topology 79

- ▼ To Access the Topologies Page 79
 - ▼ To Display Host-Based Topology 80

Monitoring Logs 84

- ▼ To Access the Log Page 84
- ▼ To Review the Most Recent Entries on a Host 85
- ▼ To View an Event Log for a Host 86

- ▼ To View Alert Logs 88
- ▼ To Display Agent Errors 89

5. Diagnostics 91

Selecting a User Interface 91 Diagnostics Test Rules 92 Running Diagnostic Tests From the Command Line 92 Standard Command-Line Arguments 93 Running Diagnostic Tests From the GUI Window 94 To Access the Diagnostic Tests 94 Test from Topology 97 To View the Topology 97 Notes 98 Test from List 98 To Invoke the Test Option Pane for a Particular Device 99 Test Manager 100 To Archive or Delete Tests 101 Diagnostic Tests Available in the Storage Automated Diagnostic Environment 102 Test Defaults 103 Default Options 103 Storage Automated Diagnostic Environment Tests 104 Sun StorEdge A3500FC Array Test (a3500fctest) 104 a3500fctest Test Options 106 Sun StorEdge A5000 and A5200 Array Enclosure Test (a5ksestest) 107 a5ksestest Test Options 108 Sun StorEdge A5000 and A5200 Array Test (a5ktest) 109 a5ktest Subtests 109

Sun StorEdge FC Tape Test (fctapetest) 111 fctapetest Test Requirements 111 fctapetest Test Options 112 Sun StorEdge PCI Fibre Channel-100 Host Adapter Test (ifptest) 113 ifptest Test Options 114 Sun StorEdge PCI Dual Fibre Channel Host Adapter Board Test (qlctest) 115 Sun StorEdge SBus Fibre Channel-100 Host Adapter Board Test (socaltest) 118 Brocade Silkworm Test (brocadetest) 121 brocadetest (1M) Test Options 122 Sun StorEdge Network FC Switch-8 and Switch-16 Switch Test (switchtest) 123 switchtest Test Options 124 Sun StorEdge T3 and T3+ Array Test (t3ofdg) 125 t3ofdg Test Options 126 Sun StorEdge T3 and T3+ Array Test (t3test) 127 t3test Subtests 127 t3test Test Options 129 Sun StorEdge T3 and T3+ Array Test (t3volverify) 130 t3volverify Test Options 131 Virtualization Engine Diagnostic Test (vediag) 132 Sun StorEdge VELUN Test (veluntest) 134 veluntest Test Options 135 Storage Automated Diagnostic Environment Tools 136 To Access the Storage Automated Diagnostic Environment Diagnostics ▼ Tools 136

Revision Checking 136

▼ To Run Revision Checking Report 137

▼ To View the Revision Checking Report 137

Fibre Channel Counters Check 138

▼ To Check Fibre Channel Counters 138

6. Reports 139

▼ To Access the Reports Page 139

Agent Statistics 140

To Check Storage Automated Diagnostic Environment Statistics 140

Thresholds List 141

Switch Traffic 141

Switch FC Graph 142

Switch FC Data 143

7. Utilities 145

System Utilities 145

▼ To Access the System Utilities Page 146

System Time-Outs 147

▼ To Change System Time-Out Settings 147

Erase Cache 148

▼ To Erase a Device's Cache 148

Run Agent 149

▼ To Run the Agent Manually 149

Email Configuration 150

▼ To Email Configuration Information 150

Recover Configuration 151

Password Maintenance 152

▼ To Change the Security Password 152

Update User Roles 153

Clear Login 154

Sun StorEdge[™] 3900 and 6900 Series Configuration 155

▼ To Access the Sun StorEdge 3900 and 6900 Series Configuration Window 155

Sun StorEdge 3900 and 6900 Series Configuration Options 157

Sun StorEdge T3+ Array View/Configure 157

Sun StorEdge Network FC Switch-8 and Switch-16 Switch View/ Configure 160

Virtualization Engines 161

Log Functions and Sun StorEdge 3900 and 6900 Series Maintenance 162

8. Storage Automated Diagnostic Environment Help 165

Site Map 165

▼ To Access the Online Help 166

Storage Automated Diagnostic Environment Event Grid 167

▼ To Customize an Event Report 167

Storage Automated Diagnostic Environment Architecture 169

Utilities 172

Storage Automated Diagnostic Environment Diagnostics Tests 172

Diagnostics Utilities 173

Storage Automated Diagnostic Environment Agent Utilities 174

Preface

The *Storage Automated Diagnostic Environment User's Guide* describes how to use the Storage Automated Diagnostic Environment graphical user interface (GUI) to collect data about the condition of various storage devices. In addition, this guide describes the tests that provide diagnostics for the following Sun products:

- Sun StorEdgeTM T3 and T3+ arrays
- Sun StorEdge A5000 and A5200 array
- Sun StorEdge A3500FC array
- Sun StorEdge network FC switch-8 and switch-16 switches
- Brocade Silkworm switches
- Fibre Channel tape
- Virtualization engine

This guide is written for system administrators and support personnel who are already familiar with Sun disk array and storage area network (SAN) products.

How This Book Is Organized

This book contains the following topics:

Chapter 1 provides an overview and general operating instructions for the Storage Automated Diagnostic Environment.

Chapter 2 contains detailed installation and configuration information for the Storage Automated Diagnostic Environment.

Chapter 3 explains the Maintenance functions that are necessary for setting up the Storage Automated Diagnostic Environment for the first time, or for making necessary changes, to ensure proper monitoring and notification.

Chapter 4 explains the Monitoring functions that you can perform using the Storage Automated Diagnostic Environment, including monitoring devices, monitoring topology, and monitoring logs.

Chapter 5 discusses the functionality of Storage Automated Diagnostic Environment diagnostic Tests from the GUI and the diagnostic tools available.

Chapter 6 discusses the various reports and lists associated with the Storage Automated Diagnostic Environment.

Chapter 7 explains the system tools and utilities specific to performing administrative tasks, all of which are optional. This chapter includes the Sun StorEdge 3900 and 6900 series utilities, which enable the user to manage the components for the Sun StorEdge 3900 and 6900 series Solutions.

Chapter 8 briefly describes the online help associated with the Storage Automated Diagnostic Environment, including utilities, man pages, the event grid, and the GUI online help.

Using UNIX Commands

This document might not contain information on basic $\text{UNIX}^{\textcircled{B}}$ commands and procedures such as shutting down the system, booting the system, and configuring devices.

See one or more of the following for this information:

- Solaris Handbook for Sun Peripherals
- AnswerBook2TM online documentation for the SolarisTM operating environment
- Other software documentation that you received with your system

Typographic Conventions

Typeface	Meaning	Examples
AaBbCc123	The names of commands, files, and directories; on-screen computer output	Edit your.login file. Use ls -a to list all files. % You have mail.
AaBbCc123	What you type, when contrasted with on-screen computer output	% su Password:
AaBbCc123	Book titles, new words or terms, words to be emphasized	Read Chapter 2 in the <i>User's Guide.</i> These are called <i>class</i> options. You <i>must</i> be superuser to do this.
	Command-line variable; replace with a real name or value	To delete a file, type rm <i>filename</i> .

Shell Prompts

Shell	Prompt
C shell	machine_name%
C shell superuser	machine_name#
Bourne shell and Korn shell	\$
Bourne shell and Korn shell superuser	#

Related Documentation

Product	Title	Part Number
Sun StorEdge T3 and T3+ arrays	 Sun StorEdge T3 and T3+ Array Start Here Sun StorEdge T3 and T3+ Array Installation, Operation, and Service Manual Sun StorEdge T3 and T3+ Array Administrator's Guide Sun StorEdge T3 and T3+ Array Configuration Guide Sun StorEdge T3 and T3+ Array Site Preparation Guide Sun StorEdge T3 and T3+ Field Service Manual Sun StorEdge T3 and T3+ Array Release Notes 	816-0772-10 816-0773-10 816-0776-10 816-0777-10 816-0778-10 816-0779-10 816-0781-12
Sun StorEdge PCI FC-100 host adapter	• Sun StorEdge PCI FC-100 Host Adapter Installation Manual	805-3682-10
Sun StorEdge SBus FC-100 host adapter board	• Sun StorEdge SBus FC-100 Host Adapter Installation and Service Manual	802-7572-11
Sun StorEdge PCI Dual Fibre Channel host adapter board	 Sun StorEdge PCI Dual Fibre Channel Host Adapter Product Notes Sun StorEdge PCI Dual Fibre Channel Host Adapter Installation Guide 	806-5857-10 806-5600-10
Sun StorEdge A5000 array	 Sun StorEdge A5000 User's Guide Sun StorEdge A5000 Release Notes Sun StorEdge A5000 Product Notes Sun StorEdge A5000 Configuration Guide Sun StorEdge A5000 Installation and Documentation Guide 	806-1946-10 806-1947-10 805-1018-13 805-0264-15 805-1903-15
Sun StorEdge A3500/A3500FC array	 Sun StorEdge A3500/A3500FC Configuration Guide Sun StorEdge A3500/A3500FC Controller Module Guide Sun StorEdge A3500/A3500FC Task Map 	805-4981-13 805-4980-11 805-4982-11
Sun StorEdge network FC switch- 8 and switch-16	 Sun StorEdge Network FC Switch-8 and Switch-16 Installation and Configuration Guide Sun StorEdge Network FC Switch-8 and Switch-16 Field Troubleshooting Guide Sun StorEdge Network FC Switch-8 and Switch-16 Release Notes 	806-6922-10 816-0252-10 816-0842-10

Product	Title	Part Number
RAID Manager 6.22	 RAID Manager 6.22 User's Guide RAID Manager 6.22 Release Notes 	806-0478-10 806-3721-10
Solaris	• Solaris Handbook for Sun Peripherals	806-2210-10
Sun StorEdge 3900 and 6900 Series	 Sun StorEdge 3900 and 6900 Series Reference Manual Sun StorEdge 3900 and 6900 Series Site Preparation Guide Sun StorEdge 3900 and 6900 Regulatory and Safety Compliance Manual Sun StorEdge 3900 and 6900 Series Hardware Installation and Service Manual 	816-3245-10 816-3242-10 816-3243-10 816-3244-10
	• Sun StorEdge 3900 and 6900 Series Release Notes	816-3247-10

▼ To Access Online Sun Documentation

A broad selection of Sun system documentation is located at:

http://www.sun.com/products-n-solutions/hardware/docs

A complete set of Solaris documentation and many other titles are located at:

http://docs.sun.com

Brocade Documentation

You can locate the following Brocade documentation on a special website provided by Brocade.

- Brocade Silkworm[®] 2400 Hardware Reference Manual
- Brocade Silkworm[®] 2800 Hardware Reference Manual
- Brocade Fabric OSTM Hardware Reference Manual
- Brocade Fabric OSTM Release Notes
- Brocade QuickLoop User's Guide
- Brocade WebTools User's Guide
- Brocade Zoning User's Guide
- Sun StorEdge SAN Release 2.0 Installation, and Configuration Guide, part number 806-6922-10**
- Sun StorEdge SAN Release 2.0 Release Notes, part number 806-6924-10**

** The Sun StorEdge switch documents are referenced for overall configuration guidelines and Operating System level and patch revision information.

▼ To Access Brocade documentation

The URL for the Brocade site is site is http://www.brocade.com.

To access the Silkworm series hardware and software documentation, from the Brocade website:

- 1. Click the Partners link.
- 2. Click the Partner Login link.
- 3. Enter the Login: Sun. Enter the password: silkworm.

Sun Welcomes Your Comments

Sun is interested in improving its documentation and welcomes your comments and suggestions. You can email your comments to Sun at:

docfeedback@sun.com

Please include the part number (*816-3142-10*) of the document in the subject line of your email.

Overview

What is the Storage Automated Diagnostic Environment?

The Storage Automated Diagnostic Environment is a host-based online health and diagnostic monitoring tool for storage area network (SAN) and direct-attached storage (DAS) devices. It can be configured to monitor on a 24-hour basis, collecting information that enhances the reliability, availability, and serviceability (RAS) of the storage devices.

The Storage Automated Diagnostic Environment offers the following features:

- A common web-based user interface for device monitoring and diagnostics
- Distributed test invocation by means of lists or topology
- Topology grouping for multi-level hosts and components
- Alternate master support for redundancy
- Revision checking
- Support for the service processor and virtualization engine components of Sun StorEdge 3900 and 6900 series offerings
- Remote notification through SRS, SRS/NetConnect, RSS, HTTP, NSM, and SMTP Providers, or email
- Support for storage area networks (SANs)

Changes from Previous Release

TABLE 1-1 and TABLE 1-2 list the changes from the previous release of Network Storage Agent 2.1.

8	
Features	Function
Diagnostic graph view	Graphical topology view of devices for testing
Diagnostic list view	Device listing with ability to invoke tests
Diagnostic invocation	Test invocation modules
Diagnostic output view	Test status and output
Diagnostic archive view	Archive for test status/output
Storage Automated Diagnostic Environment Link test	Link test for isolation of passive devices
Test Process Manager	Distributed test manager
Discover Sun StorEdge 3900 and 6900 Series	Discovery of Sun StorEdge 3900 and 6900 series
Topology	Topology comparison
Revision checking	Distributed revision checking for patches and device firmware
Sun StorEdge T3 and T3+ array out-of-band tests	Sun StorEdge T3 and T3+ array <code>t3volverify(1M)</code> and <code>t3ofdg(1M)</code> tests
Revision checking	Revision checking, configuration checking, and device testing
NetConnect update	Update for NetConnect Provider
Grouping in graph view	Grouping in graph view to create rack views
Alternate master	Alternate master for redundant control and monitoring
Virtualization Engine support	Instrumentation, health, and diagnostics for the virtualization engine.
Sun StorEdge 3900 and 6900 Series Configuration Utilities	Configure, modify, restore, and verify Sun StorEdge 3900 and 6900 Series components.
Event Grid	Shows all the actionable and non-actionable events the Storage Automated Diagnostic Environment generates.
Out-of-band topology	Topology merging options
Brocade	Support for Brocade Silkworm switches in a SAN environment

TABLE 1-1 Storage Automated Diagnostic Environment Additions

Products Not Supported

Products not supported with the Storage Automated Diagnostic Environment include the following:

- Sun StorEdge A2 array and Sun StorEdge D2 disk array
- Sun StorEdge Internal Fibre Channel Disk
- Sun Fire 880 FC-AL Disk
- Sun StorEdge A1000 array

How the Storage Automated Diagnostic Environment Works

The Storage Automated Diagnostic Environment uses a variety of system tools to monitor various devices. These system tools are executed by device-specific modules within the Storage Automated Diagnostic Environment. All information generated by the Storage Automated Diagnostic Environment is standardized to the common information model (CIM).

The Storage Automated Diagnostic Environment program executes at fixed intervals from the cron(1M) facility and relies on a configuration file describing each device that needs to be monitored. Whenever the devices can be discovered automatically, manual configuration entries are not required. The tasks required to configure the Storage Automated Diagnostic Environment are simplified by the use of a web browser-based GUI.

The Storage Automated Diagnostic Environment captures instrumentation data from, or associated with, the storage devices and applies rules to convert this into a set of events. These events contain information that characterizes the operational behavior of the device. Some of these events can represent conditions that require service action, in which case the event is tagged as an alert. The Storage Automated Diagnostic Environment sends alerts and events to various recipients through a set of notification facilities, such as email or email targeted at a pager. In addition, the Storage Automated Diagnostic Environment can send events as a telemetry stream through the providers to a secure central repository at Sun. This enables the information to be used for product improvement and enables Sun trained personnel to be more effective in providing service, both proactive and reactive.

Monitoring Devices Using the Storage Automated Diagnostic Environment

Monitoring varies from device to device but usually consists of the following methods. The Storage Automated Diagnostic Environment typically:

- 1. Finds and monitors message log files for the device and reports new, relevant entries
- 2. Executes commands to probe the device for status, state, and statistics information
- 3. Probes the device remotely for information, for devices that support remote access through a management path.

To minimize email transmission, the Storage Automated Diagnostic Environment keeps a state database on the local host running the agent. This database keeps state information from one execution to the next. When applicable, information is compared and only the differences are sent.

Information Deltas

In general, information events are divided into two categories:

- Delta: Information events in this category are never sent in their entirety; only changes are sent. This category includes device configuration, device state, and system configuration events.
- Nondelta: This category includes message log events, device statistics, and other information events. These events are never compared with previous values because there are too many changes to track. Therefore, the events in a nondelta category are sent in their entirety. Often, events in this category follow a transmission schedule; for example, device statistics for a Sun StorEdge T3 and T3+ array could be reported every seven days.

Because the Storage Automated Diagnostic Environment is executed from a cron and relies on the Internet services daemon (inetd(1M)) for communication and for the management console, space and resource requirements are kept at a minimum.

Storage Automated Diagnostic Environment Functions

For each device, the Storage Automated Diagnostic Environment performs the following functions:

1. Sends the information, by way of a discovery event, to the system administrator and/or the Network Storage Command Center (NSCC) through an interface with the transport mechanisms.

Note – The first access to a device yields a discovery event that collects all the information about that device, plus other events for other preconfigured devices, that may be generated by health monitors.

- 2. Reads the proper /var/adm/messages files, finds relevant entries, and reports them as events through the local email notification mechanism, if configured.
- 3. Connects to Sun StorEdge T3 and T3+ array storage devices directly through inband data paths and out-of-band management paths.
- 4. Reads the device's configuration and state information, stores it locally in the cache, compares the results of the last run, and transmits the differences.
- 5. Reads threshold information and reports errors when the frequency threshold reaches predefined levels.

Storage Automated Diagnostic Environment Agent Functionality

The Storage Automated Diagnostic Environment remotely monitors Sun network storage devices. The Storage Automated Diagnostic Environment can monitor host message files for errors, or connect directly through the "in-band" data path or "outof-band" management path of Sun StorEdge devices, in order to obtain status information about each device being monitored. To facilitate monitoring of this environment, the Storage Automated Diagnostic Environment can be configured as a set of distributed agents, in a master/slave relationship, spread across a series of servers. A single agent is given the role of master and acts as the collection point for events originating in the other slave agents. The master provides the web server interface.

You can use the Storage Automated Diagnostic Environment to monitor the following types of Sun storage products:

- Sun StorEdge T3 and T3+ arrays
- Sun StorEdge A5000 and A5200 arrays
- Sun StorEdge A3500FC arrays
- Sun StorEdge network FC switch-8 and switch-16 switches
- Brocade Silkworm switches
- Virtualization engines
- Fibre Channel tape
- Sun StorEdge 3900 and 6900 Series Solutions

Storage Automated Diagnostic Environment Diagnostic Functionality

Diagnostic tests have been integrated into the Storage Automated Diagnostic Environment for device diagnostics and field replaceable unit (FRU) isolation. Each test can be run individually from the command line or from the Storage Automated Diagnostic Environment user interface.

The following tests are described in Chapter 5, "Diagnostics."

- Sun StorEdge A3500FC Array Test (a3500fctest)
- Sun StorEdge A5000 Array Enclosure Test (a5ksestest)
- Sun StorEdge A5000 Array Test (a5ktest)
- Sun StorEdge FC Tape Test (fctapetest)
- Sun StorEdge PCI FC-100 Host Adapter Board Test (ifptest)
- Sun StorEdge PCI Dual Fibre Channel Host Adapter Board Test (qlctest)
- Sun StorEdge SBus FC-100 Host Adapter Board Test (socaltest)
- Sun StorEdge network FC switch-8 and switch-16 Switch Test (switchtest)
- Sun StorEdge T3 and T3+ array Tests (t3ofdg, t3test, t3volverify)
- Virtualization Engine Tests (vediag, veluntest)
- Brocade Silkworm Test (brocadetest)

From the Storage Automated Diagnostic Environment user interface, you can select tests from the topology or from a list view. When the tests execute, the Storage Automated Diagnostic Environment initiates the test on the proper host. You can retrieve test results by using the Test Manager selection from the GUI.

Tasks You Can Perform Using the Storage Automated Diagnostic Environment

Through the Storage Automated Diagnostic Environment GUI, you can perform the tasks that appear in TABLE 1-2.

Task	Function
Maintain host information	 Supports the maintenance of host-specific information such as host name, location, and IP address. Enables you to manually add a new slave or an alternate master Tests the availability of the Storage Automated Diagnostic Environment on each host using the Ping Slaves function. Enables you to push slave configuration to the corresponding slaves.
Add or update the master configuration	 Supports the maintenance of customer information and master configuration information, including: Customer name, contract number, and location information Default local message files Device category selection Monitoring frequency
Discover devices	Requests that the Storage Automated Diagnostic Environment probe the environments for the desired device types
Maintain devices	Supports the maintenance of device-specific information.
Set up local email and pager notification	 Enables local notification information: : Enable specific events to be emailed to local administrators. Events can be categorized by device type, severity level, and event type. Events can also be summarized and sent to a pager's email address. Enables events to be automatically translated from their internal encoded format to a human-readable format
Set up a Provider	Relays events generated by health monitors.
Control agent activity	Temporarily stops the Storage Automated Diagnostic Environment from running on a selected host.
View instrumentation reports	Reviews all FRU-level information details of a selected device and its access components.
Create Topology snapshot	Creates and updates the topology view from the host.

TABLE 1-2 Functions of the Storage Automated Diagnostic Environment

TABLE 1-2	Functions of the Storage Automated Diagnostic Environment

Create Topology history	Displays archived topology information of the selected host or previously-configured multiple hosts.	
Verify the health of a SAN and display topology drawings of the Fibre Channel connections.	Collects counter information based on error messages and telemetry information. This information is used in the topology drawing to indicate link failures.	
Monitor devices, SANs and logs	Displays the most recent system errors saved by the Storage Automated Diagnostic Environment.	
Adjust the system functions	 Enables you to change time-out settings Enables you to display system errors Enables you to erase device cache Enables you to manually run the Storage Automated Diagnostic Environment Enables you to display thresholds Enables you to configure email addresses Enables you to change passwords 	
Run Diagnostic tests from the GUI window.	 Enables you to access diagnostic test from a topology view Enables you to access diagnostic test from a list Enables you to adjust the default settings for selected diagnostic tests Enables you to review, delete, or archive diagnostic tests. Enables you to review the results of old tests. 	
Run Diagnostic tests from a command line	Enables you to run diagnostic tests by specifying all test options in the form of command-line arguments	
Manage Sun StorEdge 3900 and 6900 series components	Enables you to review, configure, display, and manage the Sun StorEdge 3900 and 6900 series components (Sun StorEdge T3 and T3+ array, Sun StorEdge network FC switch-8 and switch-16 switches, and virtualization engine for the Sun StorEdge 6900 series solutions).	

Installing and Starting SUNWstade

This chapter presents instructions for installing, configuring, and starting SUNWstade on your system. It includes the following main topics:

- "Installation Checklist" on page 10
- "Host Requirements" on page 12
- "Installing SUNWstade" on page 13
- "Sun StorEdge 3900 and 6900 Series Configuration" on page 18
- "Starting the Storage Automated Diagnostic Environment" on page 21
- "Upgrading or Removing SUNWstade" on page 26
- "The Storage Automated Diagnostic Environment Configuration File" on page 27
- "Starting SUNWstade" on page 28

Installation Checklist

Use this checklist to install the Storage Automated Diagnostic Environment.

Step	Action
	Download the latest compressed SUNWstade package from the Sun Download Center.
1	Note: You must have root access to install SUNWstade.
0	Install the SUNWstade package by selecting it during the package add:
z	# pkgadd -d ./tmp
3	Once you have completely and successfully installed the SUNWstade package, run the agent installation script. Refer to "To Run the ras_install Script" on page 22.
4	Configure the Sun StorEdge T3 and T3+ Array and the Host. Refer to Chapter 7, "Utilities."
5	Access the GUI on the host where the master was installed. Use the server name and port 7654 from a browser to set up the rest of the configuration: http:// <master-server.domain>:7654</master-server.domain>
6	Log in to the Storage Automated Diagnostic Environment: • Login: ras • Default password: agent
7	Set the site information parameters. Note that fields with an asterisk (*) are mandatory. Refer to Chapter 3, "Maintenance."
8	Using Device Discovery, request the Storage Automated Diagnostic Environment to probe the environments for the desired device types. Refer to Chapter 3, "Maintenance."
9	Customize the generation of email using the General Maintenance: Email Notification: Add Notification functionality. Refer to Chapter 3, "Maintenance."
10	Send a test email and a short message to verify the master instance of the Storage Automated Diagnostic Environment's mailing capability.
11	Verify the component's revision using the Diagnose: Tools: Revision Check functionality. Refer to Chapter 5, "Diagnostics."

12	Create a topology snapshot using the Maintenance: Topology Maintenance functionality. Refer to Chapter 3, "Maintenance."
13	Push the configuration. This is done to ensure that all instances of the master agent are synchronized with every device the Storage Automated Diagnostic Environment is monitoring, or to update the slaves, if applicable. Refer to Chapter 3, "Maintenance."
14	Review the configuration to ensure that the Storage Automated Diagnostic Environment is not missing required configuration information necessary for proper monitoring. Refer to Chapter 3, "Maintenance."
15	Create a Snapshot Topology and merge topologies, if applicable. Refer to Chapter 3, "Maintenance."

Host Requirements

The SUNWstade package is installed on a host in the /opt/SUNWstade directory. Servers running in the Solaris operating environment and Sun StorEdge devices are supported. Install SUNWstade on a host that satisfies these conditions:

- The host has access to /var/adm/messages files, where device logs are sent.
- The host has access to the T3 and T3+ array message log, to which Sun StorEdge T3 and T3+ array device logs are sent. The name of the file appears on the Sun StorEdge T3 and T3+ array message log configuration screen for each host.
- The host has access to the Sun StorEdge T3 and T3+ arrays and/or the Sun StorEdge network FC switch-8 and switch-16 switches over TCP/IP.
- For SAN datapath monitoring, the host has access to the data path of the devices being monitored.
- The master host can run a browser to complete and maintain the monitored configuration.
- The service processor (which is defined as a *Solaris server host*) connects to Sun storage devices "out-of-band" through Ethernet.
- The host must have the SUNWsan package installed, along with the latest 111413-xx luxadm patch.

If the host satisfies only some of these conditions, you can install the agent on multiple hosts and configure the agents to complement each other. Note, however, that SUNWstade is not a relocatable package.

Example Configurations

- The Sun StorEdge T3 and T3+ Array has an "in-band" datapath attachment to one host but logs its messages to another management host.
- In a multipath environment, more than one host has access to the same device(s).

Installing SUNWstade

This section contains the following subsections:

- "Installation Notes" on page 13
- "To Install SUNWstade" on page 13
- "To Verify the Installation" on page 16
- "To Install a Patch" on page 17

Installation Notes

- After the package has been completely installed, execute the program /opt/SUNWstade/bin/ras_install to install the SUNWstade service and cron. Refer to "To Run the ras_install Script" on page 22.
- When installing a new version of, or a patch to, the Storage Automated Diagnostic Environment, stop the agents before installing the update (refer to "To Start or Stop Agent(s)" on page 63). Then, run ras_install after installing the update.
- After installing SUNWstade, set the environment variables PATH and MANPATH to include SUNWstade directories /opt/SUNWstade/bin and /opt/SUNWstade/man. Refer to "To Set the Environment Variables" on page 21.

▼ To Install SUNWstade

• Use the pkgadd(1M) command and answer the prompts as shown in CODE EXAMPLE 2-1.

```
# pkgadd -d .
The following packages are available:
  1 SUNWstade Storage Automated Diagnostic Environment (sparc) 2.0
Select package(s) you wish to process (or 'all' to process
all packages). (default: all) [?,??,g]: 1
 (various copright notices)
Do you accept the terms? [yes,no,view,?] yes
 Installation end.
Using </opt/SUNWstade> as the package base directory.
## Processing package information.
## Processing system information.
## Verifying disk space requirements.
## Checking for conflicts with packages already installed.
## Checking for setuid/setgid programs.
This package contains scripts which will be executed with super-user
permission during the process of installing this package.
Do you want to continue with the installation of \langle SUNWstade \rangle [y,n,?] y
Installing Storage Automated Diagnostic Environment as <SUNWstade>
## Installing part 1 of 1.
/opt/SUNWstade/CHANGES
/opt/SUNWstade/INSTALL
/opt/SUNWstade/README
/opt/SUNWstade/Scripts/Solutions/bin/t3 raid4
/opt/SUNWstade/Scripts/Solutions/bin/t3 raid5
/opt/SUNWstade/Scripts/Solutions/config
/opt/SUNWstade/StarterKit/TAPEAgent.pm
/opt/SUNWstade/StarterKit/WHATAgent.pm
/opt/SUNWstade/System/.#ST_Tests.1.16
/opt/SUNWstade/System/.#ST_Tests.1.4
/opt/SUNWstade/System/DeviceState.map
/opt/SUNWstade/System/DocSequence
. . .
(Continued on following page)
```

```
. . .
/opt/SUNWstade/snmp/mibs/IPV6-ICMP-MIB.txt
/opt/SUNWstade/snmp/mibs/IPV6-MIB.txt
/opt/SUNWstade/snmp/mibs/IPV6-TC-txt
/opt/SUNWstade/snmp/mibs/IPV6-TCP-MIB.txt
/opt/SUNWstade/snmp/mibs/IPV6-UDP-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMP-COMMUNITY-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMP-FRAMEWORK-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMP-MPD-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMP-NOTIFICATION-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMP-PROXY-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMP-TARGET-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMP-USER-BASED-SM-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMP-VIEW-BASED-ACM-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMPv2-CONF.txt
/opt/SUNWstade/snmp/mibs/SNMPv2-MIB.txt
/opt/SUNWstade/snmp/mibs/SNMPv2-SMI.txt
/opt/SUNWstade/snmp/mibs/SNMPv2-TC.txt
/opt/SUNWstade/snmp/mibs/SNMPv2-TM.txt
/opt/SUNWstade/snmp/mibs/t300.mib
/opt/SUNWstade/snmp/mibs/v2_2FA.mib
/opt/SUNWstade/snmp/mibs/v2_2FE.mib
/opt/SUNWstade/snmp/mibs/v2_2SW.txt
[ verifying class <none> ]
## Executing postinstall script.
libsnmp found!
 _____
After the package is completely installed,
execute the program '/opt/SUNWstade/bin/ras_install'
to install the rasagent inet service and cron.
_____
If you choose not to use cron this time, re-run
ras_install later to establish a cron entry.
_____
# /opt/SUNWstade/bin/ras_install
   **** Installing Storage Automated Diagnostic Environment and crons ****
? Are you installing a Master or a Slave? (Enter M=master or S=slave):M
(Default = M) *** Master Install ***
This script will now add the Storage Automated Diagnostic Environment service to the inetd
config file. When this script ends, go to the master agent IP-address, port 7654 with a
browser to complete the Storage Automated Diagnostic Environment configuration.
```

(Continued on following page)

//etc/services is now updated. - The inetd.conf entry for rashtp is already in /etc/inetd.conf. ? Do you want to C=start or P=stop Storage Automated Diagnostic Environment cron [C/P, default=C] : C ** Storage Automated Diagnostic Environment cron is now installed. - Resetting the inetd services to see the new rashtp service. - Testing access to Storage Automated Diagnostic Environment webserver, (this will timeout in 20 secs) ... Storage Automated Diagnostic Environment installed properly! To complete the Storage Automated Diagnostic Environment configuration, point your browser to http://<hostname>:7654. Use the browser only after the Storage Automated Diagnostic Environment has been installed on all master and slave hosts. #

▼ To Verify the Installation

• Use the pkginfo(1M) command:

pkginfo -1 SUNWstade

▼ To Install a Patch

- 4. Download the latest Storage Automated Diagnostic Environment patch xxxxx-xx from Sunsolve to a temporary workspace.
- 5. As superuser, use the patchadd (1M) command and answer the prompts as shown in CODE EXAMPLE 2-2.

CODE EXAMPLE 2-2 Patch Installation

cd /tmp

```
# uncompress xxxxx-xx.tar.Z
# tar xvf xxxxxx-xx.tar
# cd xxxxxx-xx
# patchadd .
# /opt/SUNWstade/bin/ras_install -options
```

Chapter 2 Installing and Starting SUNWstade 17

Sun StorEdge 3900 and 6900 Series Configuration

The Sun StorEdge 3900 and 6900 Series solutions must have the file /var/opt/SUNWstade/DATA/Solution.golden created on the Service Processor before the Storage Automated Diagnostic Environment GUI is launched. This task is performed by Manufacturing (OPS) at the factory.

Manufacturing must run /opt/SUNWstade/bin/config_solutions while building the cabinet at the factory. This creates the /var/opt/SUNWstade/DATA/Solution.golden file, which is used as a baseline, shown in FIGURE 2-1.

Sun StorEdge 3900 and 6900 Series Upgrades

After any upgrade or component change (including model number upgrades), you must again execute /opt/SUNWstade/bin/config_solution on the Service Processor, which enables proper discovery of the Sun StorEdge 3900 and 6900 Series Solutions. Anytime the /opt/SUNWstade/bin/config_solution file is executed, the Storage Automated Diagnostic Environment Master must rediscover this cabinet using instructions given from the Discovery menu, Solution option. Refer to "Device Discovery" on page 43 for more information on Discovery.

FIGURE 2-1 and FIGURE 2-2 show an example Solution.golden file and Solution Golden Snapshot.
```
$ config_solution -m 6910
sp0:
vla: -> found VE (vla / 90.0.0.20 / 2900006022004188)
vlb: -> found VE (vlb / 90.0.0.21 / 2900006022004193)
swla: -> found Switch (90.0.0.30 / 100000c0dd00b1f4)
swlb: -> found Switch (90.0.0.31 / 100000c0dd0057aa)
sw2a: -> found Switch (90.0.0.32 / 100000c0dd008467)
sw2b: -> found Switch (90.0.0.33 / 100000c0dd00b170)
t3b0: -> found T3 (90.0.0.40 / 50020f230000a2a6/50020f230000a0d0/
slr-mi.370-3990-02-e-f2.041638)
t3b1: -> found T3 (90.0.0.41 / 50020f23000068cc/50020f2300004331/
slr-mi.370-3990-01-e-f0.026828)
t3b2: -> Warning: cannot ping 90.0.0.42
t3b3: -> Warning: cannot ping 90.0.0.43
t3be0: -> Warning: cannot ping 90.0.0.50
t3bel: -> Warning: cannot ping 90.0.0.51
t3be2: -> Warning: cannot ping 90.0.0.52
t3be3: -> Warning: cannot ping 90.0.0.53
t3be4: -> Warning: cannot ping 90.0.0.54
_____
     Configuration Summary
-----
Model : 6910, Requested = 6910
| Switch: 4 * 8 ports
       : swla, swlb, sw2a, sw2b,
  T3s : 2
       : t3b0, t3b1,
  VEs : 2
: vla, vlb,
_____
NO ERRORS FOUND!
Snapshot /var/opt/SUNWstade/DATA/Solution.golden saved!
Done.
```

FIGURE 2-1 Example Solution.golden file

```
#Solution Golden Snapshot
created=2002-02-08 00:32:40
created2=Fri Feb 8 00:32:40 MST 2002
model=6910
[device1]
type=ve
class=switch.ve
ip=90.0.0.20
ipno=90.0.20
name=diag244-v1a
wwn=2900006022004188
login=v1
initiator=I00001
[device2]
type=ve
class=switch.ve
ip=90.0.0.21
ipno=90.0.0.21
name=diag244-v1b
wwn=2900006022004193
login=v1
initiator=I00002
... <snip>
[device8]
type=t3
class=storage.t3
ip=90.0.0.41
ipno=90.0.0.41
name=diag244-t3b1
wwn=50020f23000068cc
wwn2=50020f2300004331
key=slr-mi.370-3990-01-e-f0.026828
ctrl_model=375-0084-02-JJ11
```

FIGURE 2-2 Solution.golden Snapshot

Starting the Storage Automated Diagnostic Environment

Once you have successfully installed the Storage Automated Diagnostic Environment, but before you start it, there are several tasks you must perform. These tasks include setting the environment variables, editing the .conf file, and manually running ras_install, and are explained in the following subsections.

- "To Set the Environment Variables" on page 21
- "To Edit the .conf Files" on page 22
- "To Run the ras_install Script" on page 22

▼ To Set the Environment Variables

After installing the Storage Automated Diagnostic Environment, you must set the environment variables PATH and MANPATH to include the Storage Automated Diagnostic Environment directories /opt/SUNWstade/bin and /opt/SUNWstade/man.

1. For the Korn or Bourne shell, type the following:

```
$ PATH=/opt/SUNWstade/bin:$PATH
```

- \$ MANPATH=/opt/SUNWstade/man:\$MANPATH
- \$ export PATH MANPATH
- 2. For the C shell, type the following:

% setenv PATH /opt/SUNWstade/bin:\$PATH

% setenv MANPATH /opt/SUNWstade/man:\$MANPATH

▼ To Edit the .conf Files

Typically, if you are using NIS, NIS+, or DNS, you can prevent potential problems with ras_install and the inetd services by editing the /etc/nsswitch.conf file, prior to running the ras_install script.

1. While in the nsswitch.conf file in the /etc directory, reorder the entries for hosts and protocols to be files nis.

Depending on the complexity of your network setup, consider removing the [NOTFOUND=RETURN] message before the word files for all other entries.

2. Update the /etc/hosts file to include valid entries for *each system* with which the Storage Automated Diagnostic Environment will communicate. This includes proper aliases as well.

For example, although 10.0.0.10 msp0 is acceptable with standard Ethernet systems, you can change one entry or all of the host entries, by updating the /etc/hosts file, to 10.0.0.10 msp0.domain.name.com msp0.

Note – Use the /etc/resolv.conf file to find where ras_install is locating the domain name. You can determine if the domainname is an issue by reading the error messages or by looking in the rasagent.conf "hostname" field.

▼ To Run the ras_install Script

You must run ras_install manually upon completing the pkgadd -d .

Execute ras_install anytime after the initial execution and setup when you need to modify the cron or when you need to change the Master or Slave title of a Host agent.

 To Start the Storage Automated Diagnostic Environment Services

1. Run ras_install.

```
# cd /opt/SUNWstade/bin
# ./ras_install
```

2. Type M (master) to the following question:

```
**** Installing the Storage Automated Diagnostic Environment Server and
Crons ****
? Are you installing a Master or a Slave Agent?
```

```
(Enter M=master or S=slave):M
```

Note – M = master is the default.

The ras_install script sets up the host as a master, establishes a Storage Automated Diagnostic Environment entry in cron tab, and restarts the cron daemon. The ras_install script also alters the /etc/inetd.conf and the /etc/services files to enable the host to serve the GUI for configuring and setting up the Storage Automated Diagnostic Environment.

The /etc/services file is updated with the Storage Automated Diagnostic Environment HTTP port number (7654) to service requests from the slave agent and to open the GUI on that port.

3. When you run the ras_install script, a cron(1M) entry is added to the cron facility, based on your answer to the following question:

? Do you want to C=start or P=stop Storage Automated Diagnostic Environment cron [C/P, default=C] : C

Note – For testing purposes or upon initial configuration, you can skip the cron activation during the installation and start the cron later by rerunning the ras_install script.

The text of the cron entry that executes is as follows:

```
0,5,10,15,20,25,30,35,40,45,50,55 * * * * \
/opt/SUNWstade/bin/rasagent -c >> /opt/SUNWstade/log/cron.log 2>&1
```

The cron starts the rasagent program every five minutes. You can adjust the monitoring frequency if necessary, and periodically enable or disable the cron in order to execute the rasagent program.

Installing Distributed Agents (Slaves)

When a server satisfies all or only some of the host requirements to monitor specific storage devices, you can optionally install the Storage Automated Diagnostic Environment on multiple servers.

When you distribute the Storage Automated Diagnostic Environment over several systems, configure only one system to communicate with the providers. This way, the secondary agents send their findings to the primary host in the form of messages through the HTTP service activated on port 7654.

Prerequisites

To install a slave agent, you must know the IP address of the host to be configured as the master agent. If you have not yet installed the master agent, abort installation of the slave agent and install the Storage Automated Diagnostic Environment on the host to be configured as the master.

▼ To Set Up a Slave

1. Run ras_install and type S (slave) in response to the following question:

```
**** Installing the Storage Automated Diagnostic Environment Server and
Crons ****
? Are you installing a Master or a Slave Agent?
(Enter M=master or S=slave): S
```

2. Enter the IP address of the master agent.

Note – If this instance of the slave was previously configured with a master, the IP address or name of that master host is the default selection.

The Storage Automated Diagnostic Environment then verifies that the master host is reachable.

Upgrading or Removing SUNWstade

This section contains the following subsections:

- "To Upgrade the Storage Automated Diagnostic Environment Package" on page 26
- "To Remove the SUNWstade Package" on page 26

▼ To Upgrade the Storage Automated Diagnostic Environment Package

1. Remove the initial installation using the following command:

pkgrm SUNWstade

Removing the initial installation does not erase the previous configuration information. The cache and topology information of each device is also retained to maintain a consistent, historical view of the Sun StorEdge devices.

2. Reinstall the package using the following command:

```
# pkgadd -d .
```

3. Run ras_install to enable the cron and to configure the agent as master or slave. Upgrade the Master first.

▼ To Remove the SUNWstade Package

1. Remove the initial installation using the following command:

pkgrm SUNWstade

2. To completely remove the package for a clean install, remove the following directories once the pkgrm command has completed its run:

```
# rm -f /var/opt/SUNWstade
# rm -f /opt/SUNWstade
```

The Storage Automated Diagnostic Environment Configuration File

The Storage Automated Diagnostic Environment configuration information is located in the file /opt/SUNWstade/DATA/rasagent.conf. Information such as site location, hosts, devices, and local notifications that you enter into the Storage Automated Diagnostic Environment graphical user interface (GUI) is stored in this file. The file remains on the system between upgrades so that previously-entered configuration information is not lost.

```
GSV ACRONYM=SADE
GSV_DESC=Storage Automated Diagnostic Environment
GSV_NAME=Storage Automated Diagnostic Environment
GSV_PKG=SUNWstade
GSV_PKGDIR=SUNWstade
GSV_PORT=7654
active=N
admin_email=yourself@yoursite
categories=a5k|host|san|switch|t3|topo|
contract=87654321
cust_no=12345678
customer=Sun Microsystems
email=rasagent@nscc.central.sun.com
fc_frequency=0
frequency=10
gui_bandwidth=H
host_adapter=S
hostname=bradster.central.sun.com
local_event_log=Y
log_months=
logfile=/var/adm/messages
mailer=/usr/bin/mail
max_event_file_size=2
```

• • •

```
. . .
```

```
[device11]
active=Y
class=storage.t3
datahost=
host=crash3.central.sun.com
hostIpno=172.20.104.3
ip=172.20.104.9
ipno=172.20.104.9
key=. .
name=t3-104-9
type=t3
wwn=50020f2300000945
wwn2=50020f230000dfe
[notification1]
device=*
email=brad.derolf@sun.com
event=*
priority=*
type=Email
```

Starting SUNWstade

Once you have executed ras_install, you can launch the Storage Automated Diagnostic Environment graphical user interface (GUI) from a web browser.

The Storage Automated Diagnostic Environment GUI is a browser-based tool that enables you to maintain and tune the Storage Automated Diagnostic Environment functions. To maintain and tune the Storage Automated Diagnostic Environment, point the browser to the host with the master instance of Storage Automated Diagnostic Environment.

The following sections explain how to use the GUI functionality.

▼ To Start the Storage Automated Diagnostic Environment

1. Open a web browser and go to http://hostname:7654 where *hostname* is the IP address or the host name of the Master.

2. Log in to the Storage Automated Diagnostic Environment by typing:

- Login: ras
- Default password: agent

The Storage Automated Diagnostic Environment administration page is displayed.

SUN.	Storage Automated Diagnostic Environment Maintenance Monitor Diagnose Report Utilities H 2.0.6.009 bradster.central.sun.com
	Welcome to the Storage Automated Diagnostic Environment copyright © 2002 Sun Microsystems, Inc. All rights reserved. Use is subject to license terms. Third-party software, including font technology, is copyrighted and licensed from Sun suppliers. Portions may be derived from Berkley SDD systems, licensed from Use CAS sun, Sun Microsystems, the Sun logo and Sun Steffage are transformed for Berkley SDD systems, Inc. in the US. and other countries. Federal Acquisitions is Commercial Software - Government User's Subject to Standard License Terms and Conditions. Copyright © 2002 Sun Microsystems, Inc. Tous droits reserves. Distribute par des licenses qui en restreignent Hutilisation. Le logicid detenu par des tiers, et qui comprend la technologie rediative aux policies de coraterce, est protegte par un copyright et license par des fourniseurs de Sun. Des parties de ce produit pourront être derivees des systemes (MSD licenses par l'Université de Californies, Inc. Sun, Sun Microsystems, le logo Sun et Sun Storfdge sont des marques de trainique ou des marques deposites de Sun Microsystems, Inc is dus d'Astures pays.
	This site can be used to configure and maintain the agents. It can also be used to monitor and diagnose Sun storage products. See the <u>Help</u> page for more details.
	Summary
	Site Info: Sun Microsystems Network STORAGE
	Installation: 2 hosts, 10 devices
	Devices: 3 A5ks, 4 Switches, 3 T3s
	Notifications: Http, NetConnect
	Email: 1 (brad.derolf@sun.com)
	Last Event: 2002-02-24 17:25:09
	Last Alert: [No alerts]
	Basic Installation Steps [Basic Steps in popup]
	Review Review Discover Setup Create Review Start Site Info Hosts Devices Emails Notifications Topology Config. Agents

The Master Configuration window opens automatically the first time you initiate the GUI. The Master Configuration window can also appear if there are empty mandatory fields, which can occur after an upgrade.

Note – The Storage Automated Diagnostic Environment is always accessed on port 7654 of the *host* configured as the *master agent*. All of the agents communicate with one another through this port to synchronize their configurations.

Product Abbreviations

TABLE 2-1 contains a list of the trademarked Sun product names and the abbreviations used in the Storage Automated Diagnostic Environment.

 TABLE 2-1
 Product Names and Abbreviations

Trademarked Sun Product Name	Abbreviation
Sun StorEdge A5000 array	Sun A5000
Sun StorEdge A3500FC array	Sun A3500FC
Sun StorEdge T3 and T3+ array	Sun T3
FC-Tape	FC-Tape
Sun StorEdge network FC switch-8 and switch-16 switch	Sun Switch
Virtualization Engine	VE
Sun StorEdge 3900 and 6900 series	Sun 3900/6900

For readability purposes, some Sun products are referred to within the Storage Automated Diagnostic Environment GUI with the abbreviated name.

Maintaining the Storage Automated Diagnostic Environment

The Maintenance chapter includes all the functions that are necessary for setting up the Storage Automated Diagnostic Environment for the first time or to make changes, as necessary, to ensure proper monitoring, notification, and diagnosis.

▼ To Access the Maintenance Window

1. Click the Maintenance link in the upper right corner of the Storage Automated Diagnostic Environment main window.

Maintenance is divided into two sections: General Maintenance and Topology Maintenance, as shown in FIGURE 3-1.



FIGURE 3-1 Maintenance Window

To configure the Storage Automated Diagnostic Environment, use the General Maintenance functions and the information found in TABLE 3-1.

Maintenance

Administration of all the Storage Automated Diagnostic Environment instances can be done from one centralized web GUI. This interface is always accessed on **port 7654** of the **host** configured with the **Master** Agent. All the agents communicate with one another through this same port to sync their configurations.

For these and all functions, additional information can be found in the Help section.

<<- Use the buttons on the left to go to the appropriate "Maintenance" func

[Site Information]	Used to enter site information: address, contact name etc
[Maintain Hosts]	Used to configure the selected hosts' instance of Storage Automated Diagnostic Environment to point to log files, device categories and execution frequency.
[Discovery]	Can be used to automatically and/or manually find devices physically attached to the host system or are accessible on the ethernet network.
[Maintain Devices]	Maintain information about specific devices (for both Masters and Slaves) that were manually or automatically discovered in a previous session and are configured to be monitored.
[Email Notification]	Optionally used to customize the generation of local only emails to the system admin or to other administrators and/or service personnel. Local notification does not affect the main transmission functions of the Storage Automated Diagnostic Environment Agent.
[Providers]	This selection determines the means by which the Storage Automated Diagnostic Environment Agent will relay events generated by the Health Monitors.
[Push Configs]	Has the same function as the <i>Push Slave Configs</i> in Maintain Hosts. Basically this will sync all slaves with the current configuration. There is no confirmation.
[Start/Stop Agent]	Can be used to temporarily stop the Storage Automated Diagnostic Environment Agent from executing for a selected host. In cases where maintenance is being done on a device, this function will aid in preventing false error Alert events.
[Start/Stop Devices]	Allows for deactivating notifications on a device-by-device basis.
[Test Email]	Generate a canned email message to an email address(es) entered in the field supplied in the window. Multiple addresses can be entered delimitted by either a comma or a space.
[Review Config]	Execute a query on the configuration and report the results and make available pointers to areas that require review.
[Config Options]	General configuration options

Note – Once you have configured the Storage Automated Diagnostic Environment Master and Slave hosts, you should use the Maintenance section only to update email notifications or to update the monitored devices.

Function	Description		
Maintain Hosts	Ensures proper tracking of the RAS information for the devices being monitored.		
Discovery	Enables the Storage Automated Diagnostic Environment to probe the in-band data paths of the Sun StorEdge A5000 arrays, Sun StorEdge A3500FC arrays, and Fibre Channel tape, as well as the out-of-band management paths for the Sun StorEdge T3 and T3+ arrays, the Sun StorEdge network FC switch-8 and switch-16 switches, and the Brocade Silkworm switches.		
	Note: Agents on hosts with a T3 and T3+ message log file will automatically find only Sun StorEdge T3 and T3+ arrays that have made entries in that log file.		
Maintain Devices	Enables the Storage Automated Diagnostic Environment to manually add devices or delete devices or to change the agent's reference to the device(s) being monitored.		
Email Notification	Enables the configuration of certain types of events for specific device types, and sends an automatic email to a list of multiple users' email addresses.		
	This option can be fully customized to streamline notifications; for example, in addition to specifying email addresses, you can specify pager numbers.		
Providers	The selections you make here instruct the Storage Automated Diagnostic Environment to use the appropriate protocol to send the device data collected by the agent modules back to Sun.		
Push Configs	If you select this option, an update will occur, even when there is no information to be updated.		
Start/Stop Agent	Enables the Storage Automated Diagnostic Environment to start or stop agents from executing.		
	For a Slave agent, this disables monitoring of only those devices that the agent has been configured to monitor. If the Master agent is turned off, the Slaves continue to run. But, if any events occur, the configurations are not pushed to the Storage Automated Diagnostic Environment until the Master agent is restarted.		

 TABLE 3-1
 Maintenance Functions

Start/Stop Devices	Enables the Storage Automated Diagnostic Environment to start or stop the alert notification of an event for one or more selected devices.		
	This function does not stop the monitoring of the device and the interface to the provider.		
Test Email	Enables the Storage Automated Diagnostic Environment to generate a generic email and send it to the list of recipients configured in the Email Notification step.		
Review Config	Enables the Storage Automated Diagnostic Environment to verify all settings and display instructions for those that have been missed or for those that should be double-checked.		
Config Options	Returns the proper options, depending on the location and status of the device. Use this window to update existing configuration options.		

 TABLE 3-1
 Maintenance Functions

General Maintenance

This following sections discuss the general maintenance functions that you can perform using the Storage Automated Diagnostic Environment:

- "To Maintain Site Information" on page 35
- "To Maintain Hosts" on page 36
- "To Maintain Devices" on page 47
- "Customizing Email Deliveries" on page 53
- "Using Providers" on page 56
- "To Start or Stop Agent(s)" on page 62
- "To Send Test Email" on page 64
- "To Review the Configuration" on page 65
- "To Change Configuration Options" on page 66

▼ To Maintain Site Information

• From the General Maintenance window, select Site Information. The fields on this page must be completed before you proceed. The fields with an asterisk (*) are mandatory.

Customer Maintenance [Help]					
Enter Site information, fields marked with a red asterisk are mandatory. This page must be entered before doing anything else in the GUI.					
Customer Information					
Šun Microsystems	* is mandatory				
<u>12345678</u>					
<u></u> į87654321	Stock Symbol:				
Unit test of SAN	*				
ž2002 Storage Lane					
Ňetwork	*				
ĬSTORAGE	Zip Code: 80026 *				
<u>Į</u> USA	*				
John Smith	*				
303 272 1234	Email john.smith@sun.com				
	nce arked with a red asterisk are ma [Sun Microsystems [12345678 [07654321 [Unit test of SAN [2002 Storage Lane [Network [STORACE [USA John Smith] 303 272 1234				

TABLE 3-3 describes the fields on the the Storage Automated Diagnostic Environment Maintain Site Information window.

Field	Action
Customer Name	Type your customer name.
Customer No.	Type your company's customer number.
Contract No.	Type your company's contract number.
Stock Ticket	Enter your company's stock market symbol as a unique company identifier.
Site Information	Type your site name and address. Include a contact person's name and either an email address or a phone number.

▼ To Maintain Hosts

When the Storage Automated Diagnostic Environment package is installed on a host, it registers with the Master agent and an entry is added to this page.

The Maintain Host section contains the following subsections:

- "To Update the Master Configuration" on page 38
- "To Add a New Slave" on page 40
- "To Push Slave Configs" on page 43
- "Device Discovery" on page 43
- "To Set Up Sun StorEdge T3 and T3+ Array Message Monitoring" on page 46

Note – For this automatic registration to work, you must first install and run the Master agent. (A 'Y' will appear in the Active field of the Maintain Hosts window.) Once all hosts are installed, you can access this page to change the configuration of each host.

1. Click the Maintenance link on the Storage Automated Diagnostic Environment main window.

2. Click the Maintain Hosts link on the Maintenance menu.

The Maintain Hosts window is displayed.

Maintain Hosts [Help]						
This page contains the list of all monitored hosts in this installation.						
 [Add a new Slave] registers a new host with this installation. An agent must already be installed on this host prior to executing this function. [Ping Slaves] verifies that all slaves are available and get their hostid and agent version#. [Push Slave Configs] transmits the configuration (address, hosts, devices, emails) to all the Slave Hosts. 						
Existing Hosts [Update All Hosts]						
HostName	Type	HostID	Host IP	Last Push	Active	Ping / Hostid
bradster.central.sun.com	Master			N/A	Y	
<u>crash3.central.sun.com</u>	Alt-Master			02-22 14:30:28	Y	
Add a new Slave Ping Slaves PUSH Slave Configs						

TABLE 3-2 describes the fields on the Storage Automated Diagnostic Environment Maintain Hosts screen. The functionality is described in more detail in the following subsections.

Button	Function
Add a new Slave	Use when the Storage Automated Diagnostic Environment Agent does not automatically register. This can occur if a previously installed instance is temporarily deleted from the configuration. Note that the Slave is not active if manually added.
Ping Slaves	Verify that the Storage Automated Diagnostic Environment at each Host is up and running. The Storage Automated Diagnostic Environment pings the Slaves and requests each Slave's HostID to verify that the Host is up and that the Storage Automated Diagnostic Environment service is available.
Push Slave Configs	Once all Slaves have been updated, you can push the new configurations by using this button. Each instance of the Storage Automated Diagnostic Environment has identical copies of the configuration file, but the administration of the Storage Automated Diagnostic Environment Slaves is done from the Master.

▼ To Update the Master Configuration

• From the Maintain Hosts window, click an existing Host Name link and update the fields as required.

Master Configuration			[<u>Help</u>]		
Enter Local Agent Information					
Host Name:	diag176.central.sun.com				
Categories to Monitor :	□:Sun A3500FC □:Sun A5000 ■:Brocade Silkworm	⊠:Host ⊠:FC Counters ⊠:Sun Switch	□:Sun T3 □:FC-Tape □:VE		
Monitoring Frequency:	120 minutes 🗖				
Sun 3900/6900 Series?:					
Monitor InBand DataPath?:					
Show only monitored devices in topology graph:					
Enter Default Logfiles					
Message Log:	/var/adm/messages				
Sun T3 Message Log:	I				
Submit Back					

TABLE 3-3 describes the fields on the the Storage Automated Diagnostic Environment Master Configuration window that you can update.

Field	Required Action		
Categories to Monitor	Select from the available categories supported <i>with this instance</i> of the Storage Automated Diagnostic Environment. You must make at least one selection.		
	Categories from which to choose include: • Sun StorEdge A3500FC array • Sun StorEdge A5000 array • Host • FC Counters • Sun StorEdge network FC switch-8 and switch-16 switches • Brocade Silkworm switches • Sun StorEdge T3 and T3+ array • Fibre Channel Tape • Topology • Virtualization Engine		
Monitoring Frequency	Select a variable between 5 and 120 minutes. This frequency variable affects how often the Storage Automated Diagnostic Environment executes.		
	Monitoring frequency does not affect the cron. The cron will initiate the Storage Automated Diagnostic Environment Agent's execution, but it is the Storage Automated Diagnostic Environment Agent's task to query this variable and test whether it is time for the Storage Automated Diagnostic Environment Agent to run.		
Sun StorEdge 3900 and 6900 series	Select the Sun StorEdge 3900 and 6900 series checkbox if you are installing the the Storage Automated Diagnostic Environment on the service processor of a Sun StorEdge 3900 or 6900 Series solution.		
Monitor InBand Data Path	Select this checkbox to enable monitoring of the in-band data path.		
Show Only Monitored Devices in Topology	Select this checkbox to show only the devices in the Monitor Devices menu. With this feature, once a device is manually deleted, it will not show in the Topology.		
Message Log	The path for the message log (for example, /var/adm/messages), which displays the history of status messages.		
Sun StorEdge T3 and T3+ array Message Log	Type the path for the host's Sun StorEdge T3 and T3+ array Message Log. The path should be to where the T3 logs are being sent. This field is used to tell the Storage Automated Discovery Environment where to look for the log files; it does not alter the location to where the Sun StorEdge T3 and T3+ array actually sends the log files.		
	Note: You can check and verify the path by looking at the /etc/syslog.conf on the host.		

▼ To Add a New Slave

If an agent does not automatically register, you can add a new Slave manually.

To access the Maintain Hosts: Add New Slave window:

- 1. Click the Maintenance link on the the Storage Automated Diagnostic Environment window.
- 2. Click the Maintain Hosts link on the Maintenance menu.
- 3. Click the Add a new Slave button.

The Maintain Hosts: Add New Slave window is displayed.

Note – The Categories to Monitor choices indicate Sun Microsystems network storage devices that are supported with *this instance* of the Storage Automated Diagnostic Environment.

Maintain Hosts: Add Ne	w Slave		[<u>Help</u>]
Enter Host Information			
HostID :	I		
Host Name:	¥ iii		
Special Contract:	N		
Categories to Monitor:	□:Sun A3500FC □:Sun A5000 □:Brocade Silkworm	□:Host □:FC Counters □:Sun Switch	□:Sun T3 □:FC-Tape □:VE
Monitoring Frequency:	[Select] 🗆		
Alternate Master:			
Sun 3900/6900 Series?:			
Monitor InBand DataPath:			
Show only monitored devices in topology graph:			
Enter Logfiles Locations			
Message Log:	/var/adm/messages		
Sun T3 Message Log:	Ĭ		
	Add	Back	

- 4. Complete the fields using the descriptions shown in TABLE 3-4.
- 5. Click Add to add the new host.

Caution – If the system you are adding as a Slave is currently a Master, the following message is displayed:

WARNING: System is currently configured as a Master. Please use the Push Slave Configs section to change this system to a Slave.

6. To automatically update the Slave's configurations and to verify that all Slaves agree with the Master configurations, click Push Slave Configs from the Maintenance menu.

Notes:

• Enter an 8-digit HostID in the /usr/sbin/hostid directory. The Storage Automated Diagnostic Environment will not add a new Slave without it.

If you do not know the HostID, you can enter a dummy number and later correct it. Note, however, that the HostID cannot match any other previously accepted HostID.

The Host you add as a Slave must be accessible over the Ethernet from the Master and must, at a minimum, be able to respond to a ping.

 If you manually enter a Slave Host and no instance of the Storage Automated Diagnostic Environment is installed on that host, you must also manually enter the HostID for that Slave Host.

Otherwise, the Categories to Monitor field will not have any devices selected and the Storage Automated Diagnostic Environment will not be active.

The Active state will depend on the Slave configuration file (rasagent.conf.push) in the /var/opt/SUNWstade/DATA directory, and will also depend on whether the Active flag is set. If there is simply no Slave configuration file (for example, if the Host is a Master), the Active state will be set to OFF. TABLE 3-4 describes the fields on the Maintain Hosts: Add New Slave window.

Field	Required Input		
HostID	Type the 8-digit alphanumeric HostID, which is the same as the system ID (for example, 123x45y6) in the /usr/sbin/hostid directory. You can use the hostid command on the Host to obtain a valid HostID.		
	If the HostID is not available, you can enter a dummy number and later correct it. Note, however, that the HostID cannot match any other previously accepted HostID.		
Host Name	Type the name of the Host (hostname.domain).		
Categories to Monitor	Choose the device the Storage Automated Diagnostic Environment will monitor. The Categories to Monitor choices indicate Sun Microsystems Network Storage devices that are supported with <i>this instance</i> of the Storage Automated Diagnostic Environment.		
Monitoring Frequency	Select a variable between 5 and 120 minutes. This frequency variable affects how frequently the Storage Automated Diagnostic Environment executes. The default is 5 minutes.		
	Note: The gron continues to run every 5 minutes, but it checks this frequency setting before loading any modules.		
Alternate Master	The Alternate Master checkbox will be enabled the first time you add a new Slave. For the Slave to be an alternate Master, you must disable this checkbox.		
Sun StorEdge 3900 and 6900 series	Enable the Sun StorEdge 3900 and 6900 series button if the device or devices the Storage Automated Diagnostic Environment will monitor are part of a Sun StorEdge 3900 or 6900 series solution.		
Message Log	The path for the message log (the default is /var/adm/messages), which displays the history of the status messages.		
Sun StorEdge T3 and T3+ array Message Log	Type the path for the host's Sun StorEdge T3 and T3+ array Message Log. The path should be to where the Sun StorEdge T3 and T3+ array logs are being sent. This field is used to tell the Storage Automated Discovery Environment where to look for the log files; it does not alter the location to where the Sun StorEdge Sun StorEdge T3 and T3+ array actually sends the log files.		
	Note: You can check and verify the path by looking at the /etc/syslog.conf on the host.		
Monitor In-Band Data Path	Enable this button if you want to view devices on the topology of a Host that are accessible in-band by that host.		

 TABLE 3-4
 Maintain Hosts: Add New Slave Fields

▼ To Push Slave Configs

1. From the Maintain Hosts window, click Push Slave Configs.

The Push Slave Configs window is displayed.

2. From the Action pulldown menu, select Change Master to Slave.

Push Slave Configs	[<u>Help</u>]
 This page can be used to update the configuration file of a set of slaves. It can also be used to change the IP address that these slaves use to send information back to the mast slave that may be sending it's information to the wrong IP-address or when there is more that one subn master. If in Maintain Hosts, you manually added a Master to your list of slaves, you can also use this function these masters to slaves. This can be useful on installations where all hosts are automatically created as a 3900/6900 Series products). 	er. This allows to fix a et available to the to properly convert masters (like Sun
Select Action	
Push ConfigFile	
Select Slave(s)	
Crash3.central.sun.com	
Update Selected Slaves	Update All Slaves

- 3. From the New Master IP pulldown menu, select the IP of the Master that will control the new Slave.
- 4. Click Update Selected Slaves.

Device Discovery

You can request the Storage Automated Diagnostic Environment to probe the environments for the device types shown on the Device Discovery window. The Storage Automated Diagnostic Environment discovers storage devices in two ways:

- Merged host-centric discovery views
- Storage-centric topology

Note – A subnet discovery mechanism discovers Sun StorEdge T3 and T3+ array storage, virtualization devices, and switch devices for a particular subnet. For Sun StorEdge 3900 and 6900 series solutions, the Host's file validates the discovery results.

To Access the Device Discovery Window

1. Click the Maintenance link on the the Storage Automated Diagnostic Environment window.

2. Click the Discovery link on the Maintenance menu.

The Device Discovery window is displayed.

Devi	Device Discovery				
Device request • Sun S attachr	Device Discovery initiates the Storage Automated Diagnostic Environment Agent to query all configured hosts for devices of the requested type. This process may take a few moments, depending on the size of the monitored configuration. • Sun StorEdge A5000 and A3500FC arrays and Fibre Channel Tape devices are discovered using the datapath attachments to the host				
• Sun S is simil	 Sun StorEdge network FC switch-8 and switch-16 are automatically found by reading the /etc/fcswitch.conf file (format is similar to /etc/hosts). Additionally, the Subnet button will ask for an ip network prefix to scan for switches answering to SMMB environments. 				
 Sum StorEdge T3 and T3+ arrays are automatically found by scanning the T3 logfile on all hosts. Entries that contain a valid ip name or address are added to the configuration. The Sun StorEdge 3900/6900 Series button discovers Sun 3900/6900 Series devices using the //var/opt/DATA/Solution.golden file on the Service Processor. The file is created by executing config_solution (refer to the config_solution man page for more information). Sunet Discovery uses some to guery a suphert and can discover switches and Sun StorEdge T3 and T3+ arrays. 					
Select a Category					
	Sun A3500FC	<u>Sun A5000</u>	Sun Switch		
	Sun T3 FC-Tape VE				
	Sun 3900/6900 Series Subnet				

- The Storage Automated Diagnostic Environment discovers the Sun StorEdge A5000 arrays and Sun StorEdge A3500FC RAID controllers on the data paths of the Host to which they are physically attached.
- The Storage Automated Diagnostic Environment discovers the Sun StorEdge network FC switch-8 and switch-16 switch devices by means of fc_addr and by parsing the IP address from the /etc/fcswitch.conf file.
- The Storage Automated Diagnostic Environment discovers the Sun StorEdge Sun StorEdge T3 and T3+ arrays by filtering the Sun StorEdge T3 and T3+ array message log file (pre-configured in the Host configuration window) for valid Sun StorEdge T3 and T3+ array IP addresses.

Note – As long as there are valid Sun StorEdge T3 and T3+ array entries in this log file, the *auto-discovery* will find them and add them. You can avoid adding older, unwanted Sun StorEdge T3 and T3+ arrays by rolling over the log file, much the same way the system log rolls over. Reference the scripts /usr/lib/newsyslog and /etc/cron.d/logchecker.

In a cascaded switch-configured SAN, only the switch with the Host attachment (that is, the local switch) needs to be in the /etc/fcswitch.conf file. The Storage Automated Diagnostic Environment will discover the cascaded switch when the first switch is discovered.

To Discover a Sun StorEdge 3900 or 6900 Series Cabinet

Sun StorEdge 3900 series storage systems for direct attach storage (DAS) environments and Sun StorEdge 6900 series storage systems for storage area network (SAN) environments are complete, preconfigured storage solutions. End users have only restricted accessibility to the internal components of the subsystems, but they do have full utilization of its storage functionality.

Use the Solution button on the Device Discovery window to instruct the Storage Automated Diagnostic Environment to query for a Sun StorEdge 3900 or 6900 series cabinet. This function uses the local /etc/hosts file for discovery on the service processor.

Discover Sun StorEdge 3900/6	900 Series [Help]
Select the rack that you want discovered.	
Run on Host:	Local 🗖
Turn Monitoring On:	v
Confirm Sun 35	000/6900 Series Discovery Back

- 1. Select the Host from the Run on Host pull-down menu.
- 2. Enable the Turn Monitoring On checkbox.
- 3. Click Confirm Sun StorEdge 3900 and 6900 series Discovery.

Note – The Storage Automated Diagnostic Environment employs a direct Ethernet connection (out-of-band) to the Sun StorEdge T3 and T3+ arrays, virtualization engines, Sun StorEdge network FC switch-8 and switch-16 switch devices, and Brocade Silkworm switch devices.

You must manually add the Brocade Silkworms (refer to "To Manually Add a Device" on page 48), or search the LAN for a Brocade Silkworm device and then manually delete the device. Refer to "To Search the LAN for Sun StorEdge Devices Using the Subnet Button" on page 46 and "To Delete a Device" on page 52 for more information.

▼ To Search the LAN for Sun StorEdge Devices Using the Subnet Button

Use the Subnet button to search the LAN for Sun StorEdge network FC switch-8 and switch-16 switch devices, Brocade Silkworm devices, and Sun StorEdge T3 and T3+ arrays.

1. Enter the IP address of the subnet you want searched.

2. Enter the Subnet prefix for the search.

The Subnet functionality will search the subnet for valid devices.

The search results are sent back to the Master agent. The Master agent moves the recently discovered, valid devices that were previously in the subnet, into the main configuration file.

▼ To Set Up Sun StorEdge T3 and T3+ Array Message Monitoring

In order for the Storage Automated Diagnostic Environment software package to monitor messages from a Sun StorEdge T3 or T3+ array, you must mirror the Sun StorEdge T3 or T3+ array's /syslog to a Host with the Storage Automated Diagnostic Environment installed and configured to monitor Sun StorEdge T3 and T3+ arrays.

- 1. Refer to the Sun StorEdge T3 and T3+ Disk Array Installation, Operation, and Service Manual for procedures on how to set up the Sun StorEdge T3 and T3+ array and the Host to forward /syslog messages.
- 2. Refer to the section "To Maintain Hosts" on page 36 to configure each Host with the name given to the Sun StorEdge T3 and T3+ array message log file.

▼ To Maintain Devices

Use the Maintain Devices window to configure the Host to monitor each device. You can also use this window to change the name the storage device will use as a reference to that device through its email notifications.

The Maintain Devices section is divided into the following subsections:

- "To Manually Add a Device" on page 48
- "To Update a Device" on page 51
- "To Delete a Device" on page 52
- "Renaming a Device" on page 52

Be aware of the host's access method when making decisions about which devices the Storage Automated Diagnostic Environment will monitor:

 TABLE 3-5
 Device Configuration

Device Monitoring Method	Inband or Out-of-Band?
Monitor <i>Sun StorEdge A3500-FC arrays</i> from a Host that has an attachment through the Fibre Channel cables and can run the Sun StorEdge RAID Manager commands.	Inband
Monitor <i>Sun StorEdge A5000 arrays</i> from the Host that has an attachment through the Fibre Channel cables.	Inband
Monitor the Sun StorEdge network FC switch-8 and switch-16 switches on any Host on the subnet, because the probing is done over the network.	Out-of-Band
Monitor the <i>Brocade Silkworm switches</i> on any host on the subnet, because the probing is done over the network.	Out-of-Band
Monitor <i>Sun StorEdge T3 and T3+ arrays</i> from the Host that has access to the Sun StorEdge T3 and T3+ array message log file and has an Ethernet connection on the subnet.	Inband and Out-of-Band
Monitor <i>Fibre Channel tape devices</i> from the Host that has an attachment through the Fibre Channel cables.	Inband
Monitor Sun StorEdge 3900 and 6900 series cabinets from the service processor, which has Ethernet connection to the Sun StorEdge T3 and T3+ array, the Sun StorEdge network FC switch-8 and switch-16 switches, and the virtualization engine.	Out-of-Band

Note – The Primary Host column lists the registered Host that does the actual probing of the device.

The buttons indicate Sun network storage devices that are supported with this instance of the Storage Automated Diagnostic Environment.

- You can discover Sun StorEdge A5000 arrays and Sun StorEdge A3500FC arrays using the Device Discovery function, which is accessible from the Discovery page or through the Maintain Devices window. See "Device Discovery" on page 43.
- ▼ To Manually Add a Device
 - 1. Click the Maintenance link on the the Storage Automated Diagnostic Environment main window.
 - 2. Click the Maintain Devices link on the General Maintenance menu.

The Maintain Devices window is displayed.

				Cli	ick here to a manu	dd a devi 1ally
Maintai	n Devices				/	[<u>Help</u>]
This page di automaticall Use [Push S	splays all monitored de ly discover all devices by lave Configs] to upda	vices. Add or upo / category. te the configurat	date devices by clickin tion of your slave ager	g the appropriate link. L its.	Jse the [Discove	r] button to
Add a new	/ Device Manually					
				and a second s		
	Select: Sun A350	00FC <u>Sun A500</u>	0 <u>Sun Switch</u> <u>Sun T</u>	3 <u>FC-Tape VE</u> <u>Silkwo</u>	orm Switch	
List Existi	Select: <u>Sun A350</u> ng Devices	<u>00FC Sun A500</u>	<u>0 Sun Switch Sun T</u>	3 <u>FC-Tape</u> <u>VE</u> <u>Silkwo</u> Host:A	Il Hosts	– [0
List Existin <u>Primary</u> <u>Host</u>	Select: <u>Sun A350</u> ng Devices <u>Select-to-Update</u>	00FC Sun A500	0 <u>Sun Switch</u> <u>Sun T</u> IP Address	4 FC-Tape VE SIIKwc Host: Al	II Hosts Functions	F GO Monitor
List Existin <u>Primary</u> <u>Host</u> diag176	Select: <u>Sun A350</u> ng Devices <u>Select-to-Update</u> <u>Carson</u>	00FC <u>Sun A500</u> <u>Type</u> A5K	0 <u>Sun Switch</u> <u>Sun T</u> IP Address	a <u>FC-Tape VE Silkwo</u> Host:A WWN	II Hosts Functions	Monitor On
List Existin <u>Primary</u> <u>Host</u> diag176 diag176	Select: <u>Sun A350</u> ng Devices <u>Select-to-Update</u> <u>Carson</u> <u>George</u>	<u>DOFC</u> <u>Sun A500</u> <u>Туре</u> A5K A5K	0 Sun Switch Sun T IP Address	3 <u>FC-Tape VE Silkwc</u> Host:A WWN	Il Hosts Functions	Monitor On On
List Existi Primary Host diag176 diag176 diag176	Select: Sun A350 ng Devices Select-to-Update Carson Georae 172.20.67.182	ODEC Sun A500 Type A5K A5K BROCADE	0 Sun Switch Sun T IP Address	a FC-Tabe VE Silkwo Host: Al	Il Hosts Functions	Monitor On On On On
List Existi Primary Host diag176 diag176 diag176 diag176 diag176	Select: Sun A350 ng Devices Select-to-Update Carson George 172.20.67.182 172.20.67.183	DOFC Sun A500 Type A5K A5K BROCADE BROCADE	0 Sun Switch Sun T IP Address	a <u>FC-Tape VE Silkwa</u> Host: Al	Il Hosts Functions	Monitor On On On On On

3. From the Add a new Device Manually section of the Maintain Devices window, select the type of device you want to add to the Storage Automated Diagnostic Environment.

Maintain Devices : Add Sun T3	[<u>Help</u>]
Enter Device Information	
Device Name : I	
Telnet Password (optional):	
Primary Host : Local	
IP Name/Address :	
Add Back	

FIGURE 3-2 Adding a Sun StorEdge T3 and T3+ array manually

Complete the required fields using the descriptions shown in TABLE 3-6.

Device	Required Input		
Sun StorEdge A3500FC array	• Type the <i>device name</i> the Storage Automated Diagnostic Environment will use to identify the device.		
	Enter the serial number for the device.		
	 Select the Storage Automated Diagnostic Environment's <i>primary host</i> that will perform the actual monitoring of the device. Note: The default is the Host that automatically discovered the device. 		
Sun StorEdge A5000 array	• Type the <i>device name</i> the Storage Automated Diagnostic Environment will use to identify the device.		
	Select the Storage Automated Diagnostic Environment's <i>primary host</i> that will perform the actual monitoring of the device. Note: The default is the Host that automatically discovered the device.		
Sun StorEdge network FC switch-8 and switch-	• Type the <i>device name</i> the Storage Automated Diagnostic Environment will use to identify the device.		
16 switches	 Select the Storage Automated Diagnostic Environment's <i>primary host</i> that will perform the actual monitoring of the device. Note: The default is the Host that automatically discovered the device. 		
	• Type the IP address that identifies the Host to the network.		

TABLE 3-6Maintain Devices: Add a new Device Manually

TABLE 3-6	Maintain De	vices: Add	a new I	Device	Manually

Device	Required Input		
Brocade Silkworm switches	• Type the <i>device name</i> the Storage Automated Diagnostic Environment will use to identify the device.		
	 Select the Storage Automated Diagnostic Environment's primary host that will perform the actual monitoring of the device. Note: The default is the Host that automatically discovered the device. 		
	• Type the IP address that identifies the Host to the network.		
Sun StorEdge T3 and T3+ Array	• Type the <i>device name</i> the Storage Automated Diagnostic Environment will use to identify the device.		
	• Type the optional telnet password, which enables the Storage Automated Diagnostic Environment to log into the Sun StorEdge T3 and T3+ array device. Note that the user is not authorized to <i>change</i> an existing Sun StorEdge T3 and T3+ array password.		
	 Select the Storage Automated Diagnostic Environment's primary host that will perform the actual monitoring of the device. Note: The default is the Host that automatically discovered the device. 		
	• Type the IP address that identifies the Host to the network.		
FC Tape	• Type the <i>device name</i> the Storage Automated Diagnostic Environment will use to identify the device.		
	 Select the Storage Automated Diagnostic Environment's primary host that will perform the actual monitoring of the device. Note: The default is the Host that automatically discovered the device. 		
Virtualization Engine	• Select the Storage Automated Diagnostic Environment's <i>primary host</i> that will perform the actual monitoring of the device. Note: The default is the Host that automatically discovered the device.		
	Enable the Turn Monitoring On button.		

Note — To obtain the device name for the Sun StorEdge A3500FC array, use the RM6 command drvutil -1 *devName*, where *devName* is derived from the command /usr/lib/osa/bin/lad.

4. Once your list is complete, click Push Slave Configs.

The Push Slave Configs functionality is required to update a Slave's configurations after you have added or updated one or more devices. Refer to "To Push Slave Configs" on page 43 for more information.

▼ To Update a Device

- 1. From the Maintain Devices window, select the device you want to update from the name link under List Existing Devices.
- 2. Once you have corrected the necessary fields, click Update.

Note – After you have added a device or updated a device, refresh the system by clicking Re-Discover.

Maintain Devices : Update Sun T3				
Device Information				
Device Type :	t3	Monitoring: On		
Device Name :	Icrash3-008			
Telnet Password (optional):	₩ ₩ ₩ # # # #			
IP Name/Address :				
IP Number :				
Key :				
WWN:				
Primary Host :	crash3.central.sun.com 🗆			
	Update Delete Re-Discover Bac	K		

Note – Changing the name of the device changes only the *reference* to that device within the Storage Automated Diagnostic Environment. It does not affect the access or reference to that device within the operating system.

Note – If you change the configuration of a device (for example, if you change the chassis ID of a switch), you must *delete* that device using the functionality found in "To Delete a Device" on page 52.

The Storage Automated Diagnostic Environment displays a message stating that the previous device has been removed. The Storage Automated Diagnostic Environment topology will not, however, display the deleted device until you manually add and discover the device using the functionality found in "To Maintain Devices" on page 47.

To Delete a Device

You can delete a device once the device has been removed from the site or if device monitoring is no longer needed.

Note – You can delete a Slave, but the only way to delete the Master is to remove the package, which is described in "To Remove the SUNWstade Package" on page 26.

- 1. Click the Maintenance link on the Storage Automated Diagnostic Environment main window.
- 2. Click the Maintain Devices link on the Maintenance menu.

The Maintain Devices window is displayed.

3. From the Maintain Devices window, select the device you want to delete from the Existing Devices list.

The Maintain Hosts: Update Host window is displayed.

4. Click Delete.

Deleting a device from the configuration does not remove the access to the device instrumentation in the cache immediately. This will, however, be cleared the next time the agent runs. State information for that device is also maintained until the agent's next run. Therefore, a device can be removed from a configuration and still be viewed in the Topology window and Instrumentation window as a "snapshot in time" until you once again execute the Storage Automated Diagnostic Environment.

Renaming a Device

If you rename a device and then execute the agent from the command line or from the GUI, the Storage Automated Diagnostic Environment displays a message that the previous device has been removed. The Storage Automated Diagnostic Environment topology will not, however, display the renamed device until you rediscover the device.

You can avoid this problem by manually adding the world wide name (WWN) into the Storage Automated Diagnostic Environment configuration file, which is located in the file /var/opt/SUNWstade/DATA/rasagent.conf.

Note – Unless you remove the SUNWstade directory or do a clean ras_install, the configuration file remains on the system between upgrades.

Customizing Email Deliveries

You can use the Local Email/Pager Notifications window to customize the generation of emails to yourself or to other administrators at their companies. For example, if you are interested in receiving only high-priority alerts coming from Sun StorEdge T3 and T3+ arrays, you can create a specialized notification for this instance only.

This section is divided into the following subsections:

- "To Set Up Local Email and Pager Notifications" on page 53
- "To Send a Customizable Subset of the Event-Driven Messages From the Host" on page 55

▼ To Set Up Local Email and Pager Notifications

Alerts are sent only to valid email addresses that you have entered through the Email Notification function. Local notification does *not* send mail to the provider.

Local Email/Pager Notifications					[<u>Help</u>]	
This list of email addresses and filenames is used by the Storage Automated Diagnostic Environment Agent to generate local notification of Alerts.						
Existing Local Notifications						
Notif. Type	Email (Click t	o Update)	Category	EventType	Priority	
Email			All	**All Events	All	
		Add Notificatio	on			

FIGURE 3-3 Local Email/Pager Notifications window

You can customize the following local notification information:

- Notification type
- Email address
- Category
- Event Type
- Priority

Note – The local email/pager notifications feature, shown in FIGURE 3-3, is optional and does not affect the main transmission functions of the Storage Automated Diagnostic Environment. The Master instance of the Storage Automated Diagnostic Environment is the only instance generating emails based on local notifications. Slave instances of the Storage Automated Diagnostic Environment send their alerts to the Master, which filters them and forwards them to the providers, if selected, and to local system administrators, if configured.

- 1. Click an email link to update or delete an existing email address.
- 2. Click the Add Notification button to add new email addresses to the notification list.

In addition to sending the RAS information collected by the Storage Automated Diagnostic Environment, you can send a customizable subset of the event-driven messages from the Host (configured as Master) directly to local system administrators at the customer's site by email.

Local Notification: Add New Entry [Help]				
Create a notification record for each Administrator who would like to be notified of findings. You can restrict the findings by Event Priority, Device Category and Event Type. Multiple events can be selected and even sent to a pager if desired. This flexibility allows for a customized alert notification scheme .				
Enter Local Natification	n Information			
Enter Local Notificatio	n information			
Notification Type:	🖲 Email 🕠 Pager			
Email Address/ Filename:	I			
Priority:	All Priorities			
Category:	All Categories 🗆			
Event Type:	All Events AgentDeinstallEvent AlarmEvent AuditEvent CommunicationEstablishedEvent			
1	Add Back			

Note – Email may not be sent if the system is not properly configured to send mail to the recipient. This is primarily evident in service processor environments where the service processors are on a subnet and there is no gateway to the intended recipient. Please refer to the *Sun StorEdge 3900 and 6900 series Installation and Service Manual*, pn 816-3244-10 for more information.
▼ To Send a Customizable Subset of the Event-Driven Messages From the Host

- 1. Enter one or more email addresses into the Email Address/Filename textbox.
- 2. For each address, choose from:
 - The Priority list:
 - 0 = normal (green)—information only. This is the lowest priority.
 - 1 = warning (yellow)
 - 2 = error (red)
 - 3 = down (red)—the system is down. The is the highest priority.
 - The Category list—Select from the current suite of storage products that are monitored by the Storage Automated Diagnostic Environment.
 - The Event Type list—select from a list of event types, categorized by device type.
- 3. Click Add.

The Storage Automated Diagnostic Environment sends the specific event-type messages, by device type, to one or more email addresses you specified.

Using Providers

The Storage Automated Diagnostic Environment supports six providers: HTTP, NSM, NetConnect, SRS, SNMP Traps, and RSS. A provider's main function is to relay events generated by health monitors. The providers do this either by pulling the events from the Persistent Data Manager (PDM) or by registering them with the PDM for a certain type of event.

This section discusses the following provider function and providers:

- "To Set Up Providers" on page 57
- "HTTP Provider" on page 58
- "NSM Provider" on page 58
- "NetConnect Provider" on page 59
- "SRS Provider" on page 59
- "SNMP Traps" on page 60
- "RSS Provider" on page 61

▼ To Access the Notification Provider Maintenance Window

1. Click the Maintenance link in the Storage Automated Diagnostic Environment main window.

2. Click the Providers link on the Maintenance menu.

The Notification Provider Maintenance window is displayed.

▼ To Set Up Providers

The Storage Automated Diagnostic Environment supports six providers. To set up one of these providers, complete the required fields shown in TABLE 3-7.

Provider	Required Action
НТТР	 Enable the Active checkbox. Type the IP address that identifies the Host to the network. Type the Proxy address. Specify, in seconds, the duration of time before timeout is to occur.
NSM	 Enable the Active checkbox. Type the IP address that identifies the Host to the network. Type an optional timeout duration (in seconds). The default is 90 seconds. Specify in which format you would like the information transmitted: Internal or XML.
NetConnect	 Enable the Active checkbox. Type the maximum size, in Kbytes, of RAS information that the the Storage Automated Diagnostic Environment agent will collect and transport.
SRS	 Enable the Active checkbox. Type the IP address of the SRS console. Type an optional second IP address. A second IP address is allowed if events need to be sent to a second notification provider, which must also accept the SRS XML format.
SNMP Traps	 Enable the Active checkbox. Type the IP name and address that identifies the Host to the network. You can provide information for up to five IP addresses. Specify the port number. Default is 162. Specify the minimum alert level: Warnings, Error, or Down.
RSS	 Enable the Active checkbox. Select or deselect Use Proxy. The Proxy can be used when the modem is not set up.

 TABLE 3-7
 Storage Automated Diagnostic Environment Providers

HTTP Provider

The HTTP Provider sends HTTP calls to an HTTP server and transfers CIM data in the appropriate format. This is an Intranet mechanism for transporting RAS information collected by the Storage Automated Diagnostic Environment Agent and is currently internal to Sun only.

Notification	Notification Provider Maintenance						
Http (On)	NSM (Off)	<u>Net Connect (On)</u>	SRS (Off)	<u>SNMP Traps (Off)</u>	<u>RSS (Off)</u>		
The HTTP Provider Environment Agent The Active checkbo	is an INTRANET me . Currently this is ir x indicates the curr	chanism for transporting Iternal to Sun only. Tent status of the Provid	g RAS information ං er.	ollected by the Storage Aut	omated Diagnostic		
					1		
Http Provider In	formation						
Act	ive: 🗹						
IP Addre	ess: [
Proxy Addre	ess: [
Timeout (se	cs): į̇́90						
		U	odate				

NSM Provider

The Network Storage Manager (NSM) provider sends Storage Automated Diagnostic Environment events to the NSM console.

Notificatio	on Provider	Maintenance			[<u>Help</u>]
<u>Http (On)</u>	NSM (Off)	<u>Net Connect (On)</u>	<u>SRS (Off)</u>	SNMP Traps (Off)) <u>RSS (Off)</u>
Network Storage	Manager: Activate	this function to send even	ts to the NSM conso	le.	
NSM Provider	Information				
	Active: 🗆 Inact	ve			
IP	Address: I				
Timeo	Timeout (secs): [90				
Transmi	t Format: Internal	(toc) 🗖			
Update					

NetConnect Provider

The NetConnect Provider is a common information model (CIM) provider, which is an independent program. It requests information from the PDM, converts the information to the appropriate format, and relays it to NetConnect.

Notification Pro	lotification Provider Maintenance						
Http (On) NSM ((Off) Net Co	nnect (On)	<u>SRS (Off)</u>	<u>SNMP Traps (Off)</u>	RSS (Off)		
The Net Connect Provider Environment Agent. The Active checkbox indi	The Net Connect Provider is the transport mechanism for RAS information collected by the Storage Automated Diagnostic Environment Agent. The Active checkbox indicates the current status of the Provider.						
NetConnect Provider I	nformation						
Active	×						
Max Size (Kbytes): [2000							
Update							

Note – The Host with the Master agent must be the same Host that is configured as the relay in a NetConnect Provider configuration.

SRS Provider

The Sun Remote Services (SRS) Provider enables a second IP address, in case the events need to be sent to a second notification provider that also accepts the SRS xml format.

Notification Provider Maintenance					
Http (Off) NSM (Off) Net (<u>Connect (Off)</u>	SRS (Off)	<u>SNMP Traps (Off)</u>	<u>RSS (Off)</u>	
Enter the IP address of the SRS console and check 'Active' to turn on the SRS Notification Provider. A second IP is allowed if events need to be sent to a second Notification Provider that also accepts the SRS xml format.					
SRS Provider Information					
Acti	ive: 🗆 Inactive				
IP Addre	ess: I				
Second IP Address (option	al): I				
Heartbeat Frequency (hou	irs): I				
Update					

SNMP Traps

The SNMP Traps Provider enables the Storage Automated Diagnostic Environment to send traps for all actionable events that occur during monitoring, to external management systems.

No	Notification Provider Maintenance [Help]							
<u>H</u>	ttp (Off) <u>NSM (Off</u>) <u>Net Conn</u>	ect (Off) SRS (Off)	<u>SNMP Traps (Off)</u>	<u>RSS (Off)</u>			
Acti Arr Sele Onl	Activate this notification provider to send traps to external Management Systems. A maximum of 5 different IP can be entered along with a port# and a minimum alert level. Select [Warning] to send all alerts, select [Error] to only send errors, and worse. Only actionable events are sent.							
H	ttp Provider Information	on	hatiya 🗆 t					
			Active: 🗆 Inactive					
#	IP Name/Address	_	Port	Min Alert Level				
1	I		<u>1</u> 62	Warnings 🗖				
2	Ĭ		<u>1</u> 62	Warnings 🗖				
3	Y		Ĭ162	Warnings 🗖				
4	Ĭ		<u>i</u> 162	Warnings 🗖				
5	Y		Ĭ162	Warnings 🗖				
			Update					

When an alert occurs, it is sent to the SNMP transport as an SNMP trap. An SNMP trap listener can use the StorAgent.mib SNMP MIB file, which is included in the SUNWstade package to decode these alerts.

The alerts contain the following information:

- 1. Storage Automated Diagnostic Environment Agent location
- 2. Storage Automated Diagnostic Environment device to which alert pertains
- 3. Alert level
- 4. Message content

RSS Provider

The Remote Support Service (RSS) Provider uses modem technology with the UNIXto-UNIX Communication Protocol (UUCP). RSS software is required on the Host and is intended to access the modem. It must be configured accordingly.

Notification Provi	Notification Provider Maintenance					
Http (On) NSM (Of	f) <u>Net Co</u>	nnect (On)	<u>SRS (Off)</u>	SNMP Traps (Off)	RSS (Off)	
The Uucp Provider allows the	e transfer of Stora	ge Automated Dia	gnostic Environi	ment Agent Events to the	SRS infrastructure.	
RSS Provider Information Provider is Active	Inactive					
Use Proxy	*Local					
Heartbeat Frequency (hours)	Ι					
		Upd	ate			

The RSS software is delivered with the Service Processor software bundle. The intended audience for this software and for this provider are those customers who have purchased a remote support service offering and are supplying phone lines to make use of the modem phone home capability.

Note – If the Storage Automated Diagnostic Environment is run either manually or from the cron, and the RSS Provider is selected, but the RSS software is not installed or is not configured properly, the following error message appears:

***ERR: Cannot find Machine name in Permissions file.

Push Configs

You will be prompted to push configurations whenever you change a configuration. Refer to "To Review the Configuration" on page 65 for a list of settings that you might have missed or you need to double-check.

Note – The Push Configs functionality, detailed in the section "To Push Slave Configs" on page 43, is very important. It synchronizes all Slave instances under a Master instance with the instances of the Master to ensure proper monitoring and notification.

▼ To Start or Stop Agent(s)

You can control agent activity to temporarily stop the Storage Automated Diagnostic Environment from running on a selected host. You can also avoid creating email notifications on false errors when a device is being tested and faults are injected intentionally.

Note – By default, the Master Monitoring function does not automatically default to *On*. You must manually enable Monitoring to *On*.

The subsections associated with controlling agent activity are as follows:

- "To Access the Start/Stop Agents Window" on page 62
- "To Deactivate Monitoring on a Device-by-Device Basis" on page 63
- To Access the Start/Stop Agents Window
- 1. Click the Maintenance link in the Storage Automated Diagnostic Environment main window.
- 2. Click the Start/Stop Agents link on the Maintenance menu.

The Start/Stop Agents window is displayed.

Start/Stop Agents						[<u>Help</u>]
Hosts Current Monitoring Sta	ite					
HostName	Туре	HostID	Host IP	Monitoring	Start	Stop
<hostname-1></hostname-1>	Master	<hostid1-1></hostid1-1>	<ip-1></ip-1>	On		
<hostname-2></hostname-2>	Slave	<hostid1-2></hostid1-2>	<ip-2></ip-2>	On		
Update Agents State						

Using the Start/Stop Agent(s) function to avoid creating email notifications on false errors works only if you are monitoring a *single* device type and all of the devices are monitored simultaneously, as in an upgrade.

To control email notifications for specific devices, refer to "To Deactivate Monitoring on a Device-by-Device Basis" on page 63.

▼ To Deactivate Monitoring on a Device-by-Device Basis

To access the Start/Stop Device Monitoring window:

- 1. Click the Maintenance link on the Storage Automated Diagnostic Environment main window.
- 2. Click the Start/Stop Devices link from the Maintenance menu.

The Start/Stop Device Monitoring window is displayed.

Start/Stop D	Start/Stop Device Monitoring [Help]					
Start or Stop devices When monitoring is submitted, the progr	by checking the appr Off', the 'Start' checkl 'am will call each host	opriate che lox is availa : and updat	ckbox. When Monitorin ble. Multiple selection e its configuration.	ig is 'On', the 'Stop' check i in the same submission	box is av is allowe	ailable. ed. Once
Start/Stop Device	Monitoring					
Primary Host	Name	Type	Address/WWI	N Monitoring	Start	Stop
diag244	diag244-sw1a	switch		On		
diag244	diag244-sw1b	switch		On		
diag244	diag244-sw2a	switch		On		
diag244	diag244-sw2b	switch		On		
diag244	diag244-t3b0	t3		On		
diag244	diag244-t3b1	t3		On		
diag244	diag244–v1a	ve		On		
diag244	diag244-v1b	ve		On		
Submit Changes						

3. To start or stop devices, check the appropriate checkbox.

- When Monitoring is *on*, the "Stop" checkbox is available.
- When Monitoring is *off*, the "Start" checkbox is available.

4. Click Submit Changes.

The Storage Automated Diagnostic Environment must update the configuration files on all the Slave hosts configured with this Master. To accomplish this, use the Push Configs function on the Maintenance menu.

5. If the device is being tested or if faults are being injected into the device intentionally, you can temporarily stop the local notifications for a specific device by using the Start/Stop Device Monitoring window.

Note – Monitoring continues when the device is deactivated (turned off). However, email notifications will not occur for any faults that are detected while the device monitoring is in this state. Consequently, any errors that may have been detected will be logged and sent by means of the NetConnect or HTTP Providers, but not by means of email notification.

▼ To Send Test Email

Use the Test Email window to send test emails and a message, and to verify that the mailing capability of the Storage Automated Diagnostic Environment is installed and working properly.

- ▼ To Access the Test Email Window
 - 1. Click the Maintenance link in the Storage Automated Diagnostic Environment main window.
 - 2. Click the Test Email link on the Maintenance menu.

The Test Email window is displayed.

- 3. Type your email address into the Email Address textbox.
- 4. Type a brief comment into the Message textbox.

If you leave the Message text box blank, the test email contains a default message with the words *Test Message* in the subject line.

Test Email [Help]					
Enter an Email Address and	an optional Message to send an email.				
Email Address:	I				
Message:	b and				
Submit					

Note – If you place a carriage return in the Message field, you cannot enter text. To restart, click in the Message textbox and enter text.

▼ To Review the Configuration

- 1. Once you have completed your configuration and you want to verify all settings, select the Review Configuration link on the Maintenance menu.
- 2. If necessary, follow the displayed instructions for settings that you might have missed or for those that you need to double-check.

C	onfiguration Error/Warning Report [Help]	
No	Details	Click here to
1	Error : The configuration of slave diag219.central.sun.com needs to be updated, Execute <u>Push Slaves</u> to update Slaves.	to the
2	Warning: There is no category in 'diag230.central.sun.com' selected in 'Maintain Host' for this a5k deviçe: DSQA3. Change the categories in <u>Maintain Hosts</u> .	screen
3	Warning: There is no category in 'diag230.central.sun.com' selected in 'Maintain Host' for this a5k device: DSQA2. Change the categories in <u>Maintain Hosts</u> .	
4	Warning: There is no category in 'diag219.central.sun.com' selected in 'Maintain Host' for this tape device: tape-6905. Change the categories in <u>Maintain Hosts</u> .	

▼ To Change Configuration Options

The Configuration Options window, shown below, returns the options depending on the location and status of the storage device. Use the information in TABLE 3-8 to update existing configuration options.

Configuration Options	[<u>Help</u>]		
Enter Configuration Options			
Number of old (historical) topologies to save for comparison :	<u>ط</u>		
Test Manager Refresh rate:	30 (minimum is 5 secs.)		
Max Event File Size:	[2 (Meg)		
Path to Email Program:]/usr/bin/mail		
Minimum FC monitoring frequency (mins.):	Ĭ		
GUI Bandwidth:	High Bandwidth 🗆		
Screen Size:	Small (1024×1000) 🗆		
Submit			

 TABLE 3-8
 Configuration Options

Configuration Option	Description
Number of old (historical) topologies to save for comparison	Type the number of topologies to save in Topology History for comparison.
Test Manager Refresh Rate	Specify the window refresh (re-load) rate, in seconds, for test output.
Max Event File Size	Specify, in megabytes, the maximum event file size. Used to limit data packet size to notification providers.
Path to Email Program	Use this field to change the local email program.
Minimum FC monitoring frequency (in minutes)	Use to configure FC counter monitoring rate (in minutes).
GUI Bandwidth	Specify the bandwidth for low and high speed local area network (LAN). The default is Low Bandwidth.
Screen Size	Use to configure the default GUI screen size.

Topology Maintenance

The Storage Automated Diagnostic Environment's graphical storage area network (SAN) interface displays all fabric components and the state of those components. Fabric components include HBAs, switch ports, storage controllers, and disks, along with more specialized components such as fans, batteries, power, and volumes.

SAN agents collect counter information based on error messages and telemetry information. This information is then used in the topology drawing to indicate link failures.

This section includes the following SAN maintenance functions that you can perform using the Storage Automated Diagnostic Environment:

- "To Access the Maintenance Window" on page 31
- "To Create a Topology Snapshot" on page 68
- "To Merge Topologies" on page 69
- "To Display Topology History" on page 70
- "Grouping" on page 71

▼ To Access the Topology Window

1. Click the Maintenance link in the upper right corner of the Storage Automated Diagnostic Environment main window.

Maintenance is divided into two sections: General Maintenance and Topology Maintenance.



Note – In order to see a topology, you must first execute the ras_install command to start the Storage Automated Diagnostic Environment services.

▼ To Create a Topology Snapshot

Use this function to create and update the topology view from the Host or to review error details. When the individual Host topologies are updated, the merged topology views are also updated.

Note – This function requires Solaris 8 or later.

To view topologies in the Monitoring and Diagnostic windows, you must first create a snapshot.

Before you create a Topology snapshot, make sure there are no failed over Sun StorEdge T3 or T3+ array paths. If there are failed over Sun StorEdge T3 or T3+ array paths, the Topology view will not show the paths correctly.

1. From the Topology Maintenance window, select Snapshot History.

The Topology Snapshot window is displayed.

alle Sum	Storage Auto	omated Diagnostic Envi	ronment	Maintenanc	e <mark> </mark> Monito	r Diagnose	Report	Utilities I
microsystems	2.0.6.009 bradst	ter.central.sun.com		General Maint. Topo	ology Main	it.		
Topology Maint.	Торо	logy Snapshot						[<u>Help</u>]
<u>Topo. Snapshot</u> - <u>Merae Topoloaies</u> - <u>Topoloay History</u> - <u>Groupina</u>	THIS fur - Click [- Once a available - The 'Li - The 'Li - If a tir Note: TI Select T	action is used to retrieve topol Create and Retrieve Selected all topologies are retrieved, th topology into the 'MASTER' to ast Snapshot' column contains ast Master' column contains th neout-error occurs, you may r his function requires Solar opology	ogies from different hosts Topologies] to start discove e Status should say Done'. opology. the date of the last time a ne date the last Master Top need to increase the timeou is 8 or above to work.	(master and slaves) and ry on the selected hosts. Then click [Merge and P a topology was extracted ology was sent back to t its in <u>System Timeouts</u>	merge the ush Maste from this hat host.	em into a Mas er Topology] t : host.	ster Top o merge	ology. 9 all
	Select	Topology	Last Snapshot	Last Master	Status	Error Details	5 D	elete
		bradster.central.sun.com (Local)	2002-02-14 16:12:49	2002-02-22 11:04:48	Done	<u>Review</u>		
		crash3.central.sun.com	2002-02-14 16:13:01	2002-02-14 16:13:14	Done	<u>Review</u>		
	Create	and Retrieve Selected Top	ologies	Merge and Push	Master	Topology		

- 2. Retrieve the selected Master or Slave topology.
- 3. Once the individual Host snapshots have been retrieved, click Merge and Push Master Topology to synchronize the Master and Slaves.

Note – Once the topology has been created, verify the Topology view.

▼ To Merge Topologies

A Master configuration topology includes the Master and all Slaves and is automatically generated.

1. From the Topology Maintenance window, select Merge Topologies.

The SAN Merge Topologies window is displayed.

- 2. To create a combined (merged) topology, select two or more topologies and enable the corresponding Select checkboxes.
- 3. Enter a unique name for the merged topology. Click Create.

By merging topologies, multiple Host topology drawings are combined into a single Topology view.

SAN Me	SAN Merge Topologies [Help						
Merge the	topologies (san_merge)						
Combined to male size							
Delete	SanName	Created	Components				
	MASTER	2002-01-17 21:38:56	diag245.central.sun, diag244.central.sun.com				
Available Topologies							
Select	Hos	;tName	Created				
	diag244.central.sun.com		2002–01–17 21:41:26				
	diag245.central.sun.com		2002-01-17 21:41:25				
Select Agents and enter a name for this new topology: [Create							

4. To delete a merged topology, select the topology from the Combined topologies section and click Delete Selected.

Note – Topologies can be updated only by creating a new Topology snapshot, which overwrites the old snapshot.

Merged topologies automatically incorporate new snapshots.

If you wish to delete a merged topology, use the SAN Merge Topologies functionality, not the Snapshot History functionality.

▼ To Display Topology History

With this function, you can compare and delete current Topology snapshots, or you can view, compare or delete previously stored Topology snapshots.

1. From the Topology Maintenance window, select Topology History.

Snapshot History []					
Click a date link to see the topology diagram . Check 2 topologies and press [Compare Selected] to see a report of the changes between the topologies. [Delete Selected] will delete all checked topologies.					
Available topologies					
Host	Current	Date1	Date2	Date3	
bradster.central.sun.com		2002-02-04_10:40:52	2002-02-06_21:39:10	2002-02-07_18:25:37	
		□ <u>2002-02-07_23:11:42</u>			
crash3.central.sun.com		<u>2002-02-04_10:41:05</u>	<u>2002-02-06_21:46:15</u>	<u>2002-02-07_18:25:48</u>	
		□ <u>2002-02-07_23:11:55</u>			
Compare Selected		De	lete Selected		

- 2. To see a summary of the changes between two topologies, select two Hosts and enable the corresponding Current checkbox, then click Compare Selected. The Host must have at least two topologies to use the Compare Selected feature.
- **3.** To delete a topology, select one or more Hosts and enable the corresponding Current checkbox, then click Delete Selected.
- 4. To view a previously stored Topology snapshot, double click the date link from the Date1, Date2, or Date3 columns. The oldest snapshot is displayed first.
- 5. To compare or delete previously stored Topology snapshots, click the corresponding checkbox on the Date1, Date2, or Date3 columns.
 - a. To compare two or more previously stored snapshots, click Compare Selected.
 - b. To delete one or more previously stored snapshots, click Delete Selected.

Grouping

The Storage Automated Diagnostic Environment Topology Grouping function enables you to display multilevel topologies and to aggregate a large number of devices in a single view.

This section discusses the following functions that are associated with Topology grouping:

- "To Create and Maintain Groups" on page 71
- "To Add Devices to a Particular Group" on page 73

▼ To Create and Maintain Groups

1. In the Grouping window, create a group name.

To create a new group name, you must specify a hostname or merged topology name, a group code, and a description. The name and code are arbitrary, but should be chosen to accurately reflect the top level. An example of this might be a campus name (Campus1 with a code of Building 1).

Grouping [Help] This section allows the user to create multi-level grouping which can be shown in the monitoring and diagnostic topology (graph) views. An example of grouping would be to create a campus group which contains several lab groups. Each lab group can then contain several rack groups which consist of the individual devices. After a grouping category is created, select the group category and assign different group-id names to the devices. For example, under a lab grouping, some devices can have the lab-id 'lab1' and others may have a lab-id of 'lab2'.						
Maintain		MERGE-MASTER:BRM5-194				
Grouping Maintenance						
Host	Group Code	Description	Delete			
MERGE-MASTER						
			DeleteSelected			
New Grouping Category						
Host: MERGE-MASTER	□ Code: [Desc	AddCategory			

FIGURE 3-4 Storage Automated Diagnostic Environment Grouping window

2. Click Add Category to add the group name.

3. Click the Merge Master link to create the new group.

The Group Detail window is displayed, showing a list of all the devices contained in the Host or merged topology you chose, shown in FIGURE 3-5.

alle Sama	St	orage Auto	omated Diagnostic Enviro	nment	Mainte	nance Monitor Diagn	ose Report L	Jtilities Help
microsystems	2.0.	6.005 diag24	45.central.sun.com		General Maint.	Topology Maint.		
Topology Maint. - <u>Topo. Snapshot</u> - <u>Merge Topologies</u> - <u>Topology History</u>		Groupin This section views. An ex several rack different gro have a lab-i	1g allows the user to create multi- ample of grouping would be to c groups which consist of the indi- oup-id names to the devices. For d of 'lab2'. (groupingMaint)	level grouping which can reate a campus group wh vidual devices. After a gro example, under a lab gro	be shown in the n nich contains sever ouping category is ouping, some device	nonitoring and diagnos al lab groups. Each lab created, select the gro es can have the lab—ic	stic topology (group can th up category ar l 'lab1' and ot	[<u>Help</u>] graph) en contain nd assign hers may
<u>a. vapini</u>			<u>Maintain</u>		MERGE-	MASTER:SE		
		Category 'N	IERGE-MASTER:SE'	Type: *All	_ _		[<u>Details</u>	<u>Summary</u>]
		+ <u>Type</u>			<u>Name</u>		<u>SE-Id</u>	Select
		host	diag245.central.sun.com					
			diag237.central.sun.com				diag237	
			diag244.central.sun.com			l		
		switch	100000c0dd008467					
			100000c0dd008863					
			100000c0dd006487					
			100000c0dd00b170		•			
			100000c0dd008849					
			100000c0dd00b1f4		•			
			100000c0dd0057aa					
			100000c0dd009e12		·			
			100000c0dd006fd7					
		t3	slr-mi.370-3990-02-e-f2.04	1638				
			slr-mi.370-3990-01-e-f0.026	5828				
		ve	2900006022004193					
			2900006022004188					
		Page: 1 of 1				SE-Id: I		Update

FIGURE 3-5 Group Detail window

▼ To Add Devices to a Particular Group

1. Select the device's checkbox and select an ID for the group.

The Group ID enables multiple instances of similar objects. An example of this is rack1, rack2, rack3, and so on.

You can also create subgroups by using the '.' notation in the Group-ID field. By using names such as lab1.rack1 and lab1.rack2, you create a lab1 group that contains two separate rack groups. The '.' notation enables multilevel grouping.

2. To access the Group Summary window, select the Summary link.

This menu enables you to select the color and type of group icon that you want displayed. From this window, you can also add a description, which will appear next to the icon in the topology view.

3. To delete a component from a group, clear the Description field of the component you want to delete, and click UpdateIDs.

in Sum	Storage Autom	ated Diagnosti	c Enviror	nment	Mai	ntenance Monitor	Diagnose	Report I	Jtilities Help
microsystems	2.0.6.005 diag245.c	entral.sun.com			General Mair	it. Topology Maint			
Topology Maint.	Grouping								[<u>Help</u>]
- <u>Topo. Snapshot</u> - <u>Merge Topologies</u> - <u>Topology History</u> •Grouning	This section allo views. An exam several rack gro different group- have a lab-id of	ws the user to cre ple of grouping wo ups which consist -id names to the d 5 'lab2'. (groupingMai	ate multi-l uld be to c of the indiv evices. For nt)	evel grouping which can reate a campus group wh vidual devices. After a gro example, under a lab grou	be shown in th ich contains se uping category uping, some de	e monitoring and veral lab groups. E is created, select vices can have the	diagnostic ach lab gro the group o lab—id 'la	topology (oup can th category a b1' and ot	graph) en contain nd assign hers may
<u></u>		<u>Maintain</u>			MERG	E-MASTER:SE			
	Maintain Cate	goryID for 'MERG	E-MASTER	t:SE'				[<u>Details</u>	[<u>Summary</u>]
	CategoryID	#Members	Descrip	tion		Style	C	olor	
	SE.diag237	1	Ι			rack 🗖		gray 🗆	2
				Update	IDs				

FIGURE 3-6 Grouping Summary window

CHAPTER 4

Monitoring

This section discusses the following monitoring functions you can perform using the Storage Automated Diagnostic Environment:

- "To View Instrumentation Reports" on page 76
- "To Review the Most Recent Entries on a Host" on page 85
- "To View an Event Log for a Host" on page 86
- "To View Alert Logs" on page 88
- "To Display Agent Errors" on page 89

▼ To Access the Monitor Page

• Click the Monitor Devices link on the Storage Automated Diagnostic Environment main window.

The Monitor window is displayed.



Monitoring Devices

You can use the Monitor Devices (Instrumentation) window to review all FRU-level information and access components of a selected device.

Instrumentation agents are very different from one another because they are specialized modules designed to probe a specific type of device. Each instrumentation agent produces reports and, when available, reads new entries into the logs accessed by the /var/adm/messages function.

Click here for Summary options

▼ To View Instrumentation Reports

1. Click the Monitor Devices link on the Monitor main window.

or Diagnose Report Utilit<u>ies Help</u> Maintenar Storage Automated Diagnostic Environment 2.0.6.009 bradster.central.sun.com Devices | Topology | Log Monitor Devices (Instrumentation) Summarv Summ, | Alerts | Log | Report | Graph] Calling crash3.central.sun.com. Name: [All Hosts GO Host: T3 t3-104-21 WWN: Name: bradster host 2002-02-22 16:45:14 Monitored: Comm: crash3 a5k 2002-02-22 09:28:11 2002-02-22 09:28:09 Identification Rev./Info crash3 a5k Type Comp Status a5k 2002-02-22 09:28:06 Unit 1 crash3 crash3 2002-02-22 09:28:54 Disk SEAGATE/ ST318203ESUN18G readv- enabled D94A host Disk SEAGATE/ ST318203FSUN18G u1d2 ready- enabled D94A crash3 switch 2002-02-22 09:28:14 Disk SEAGATE/ ST318203FSUN18G u1d3 ready- enabled D94A 2002-02-22 09:28:14 crash3 switch D94A Disk SEAGATE/ ST318203FSUN18G u1d4 ready- enabled crash3 switch 2002-02-22 09:28:12 u1d5 crash3 switch 2002-02-22 09:28:13 Disk SEAGATE/ ST318203FSUN18G ready- enabled D94A Disk SEACATE/ ST318203ESUN18C u1d6 ready- enabled D94A crash3 t3 2002-02-22 09:28:38 SEAGATE/ ST318203FSUN18G u1d7 Disk readv- enabled D94A crash3 t3 2002-02-18 05:55:12 Disk SEAGATE/ ST318203FSUN18G u1d8 ready- enabled D94A crash3 t3 2002-02-22 09:28:50 Disk fault- disabled u1d9 Cntrl SCI-S|/ 375-0084-01-j-j6 u1.ctrl ready- enabled 0210/011800 ✓"Warning ■"Error ■"Down Midplane SCI-SJ/ 002399 u1.mpn ready- enabled 0000 Loop 008097 u1|1 ready- enabled 5.02 Elash 008250 u112 ready- enabled Loop 5.02 Flash Use the 'Name:' field to search large list of devices. Name Search Systax: Enter a few or all letters of the device name. Port 50020F230000095F u1p1 online tectrol-can- 300-1454-01(50) u1pcu1 ready-enabled 0000 Power Wildcards are not supported Example: '192' will find device 'diag–192' and device '192.20.20.1' Battery:normal, fan1:normal, fan2:normal u1pcu2 ready-enabled Power tectrol-can- 300-1454-01(50) 0000 Battery:fault, fan1:normal, fan2:normal Volume v2 (raid5) u1vol1 unmounted 143.23 GB Volume u1vol2 0.00 GB Unit 2 SEAGATE/ ST318203FSUN18G u2d1 Disk ready-enabled D94A Disk SEAGATE/ ST318203FSUN18G u2d2 ready- enabled D94A SEAGATE/ ST318203FSUN18G Disk u2d3 ready- enabled D94A Disk SEAGATE/ ST318203FSUN18G u2d4 ready- enabled D94A u2d5 ready-enabled SEAGATE/ ST318203ESUN18G D94A

FIGURE 4-1 Monitor Devices (Instrumentation) window

- 2. Enter at least a portion of the device name to display specific devices.
- 3. Choose a host from the Host pull-down menu and click GO.

A list of devices for that host appears.

- 4. Click on a device from the Name column to view a list of all of the device's components.
- 5. Click on a component's corresponding Alerts, Log, Report, or Graph link, found in the right frame.

Information that is available from the Monitor Devices (Instrumentation) window follows:

Left Frame

- A list of all monitored devices
- The severity column, which shows current errors and warnings on a selected device. To access the Alerts page, drag your mouse over a severity button and left click the mouse button. The Alerts summary is displayed in the right frame.
- The report date on which alerts, logs, and summaries will be generated.

Right Frame

- A summary report about the selected device
- An alert state page showing any problem with the selected device
- A log page showing all events generated against the selected device
- A report page showing, in detail, all monitored attributes for the selected device
- A graph summary, displaying a summary of the component in graphical format

Note – You can also access the content of the right frame, using the Topology view, by right-clicking an icon and clicking Report.

Graph Summary

When you select Graph from the Monitor Devices window, the status of the component is displayed in graphical format.

Note – The Storage Automated Diagnostic Environment currently displays only the Sun StorEdge T3 array in the graphical view, as in FIGURE 4-2.



FIGURE 4-2 Sun StorEdge T3 Array Graph Summary and Component Alert List

Devices other than the Sun StorEdge T3 array do not display in graphical format, in which case the following message appears: No Graph Available!

The Alerts list, however, does appear with every device.

Monitoring Topology

This section discusses Topology monitoring functions you can perform using the Storage Automated Diagnostic Environment.

▼ To Access the Topologies Page

1. Click the Monitor Topology link on the Storage Automated Diagnostic Environment main window.

The Topologies window is displayed.

2. Use the Topology section to display a host-based topology and SAN-based topologies.



Topologies are created by merging multiple host topologies. Basic host topologies include all devices that can be seen, on the Fibre Channel connection, from that host, including host-centric and direct attached storage (DAS) topologies. These topologies can be merged to construct a SAN-based topology (those created by combining two or more topologies).

When you select a storage device or a merged topology, the index displays all host topologies that include that device.

Note – This function requires Solaris 8 or above.

To Display Host-Based Topology

In order to see a topology, you must first do the following:

- 1. Execute the ras_install command to start the Storage Automated Diagnostic Environment services.
- 2. After device discovery, create a Topology snapshot. Refer to "To Create a Topology Snapshot" on page 68 for more information.

1. Select Monitor Topology from the Monitor Menu.

The Topology menu appears.

2. Using your left mouse button, select an individual host or a merged (Master) topology.

The index displays all host topologies that include that device, as shown in the following example.





Note – Select the <u>Clear</u> link in the Topology view to accept all errors shown in the Topology. Once you have selected the <u>Clear</u> link, *all errors*, including RED errors in the 'bad' state, are removed from Topology.

Normally, you should not select the <u>Clear</u> link until all faults have been corrected.

Topology Feature Hints

- To narrow the size of the topology, select an address from the Filter pull down menu and click GO.
- To zoom in closer, right click on the screen's white space. You can zoom to 100%, 83%, 66%, or 50% by selecting the value with your left mouse button.
- To isolate a faulty power supply, right click on the enclosure, then right click on the report. An Alert log will display all alerts for the enclosure's power supplies separately.

Note – Some enclosures might have two power supplies but only one P box. In this case, the P box will turn red, yellow, or green if *either* power supply is faulty.

- To save the layout, click the save button next to XY Layout on the footer bar.
- To view actual illustrations of the devices vs. conceptual drawings of them, click the Turn Off button next to Graphics on the footer bar.
- To change the layout of the topology from default to Sun StorEdge 3900 or 6900 series Solutions rack, click [Change] next to Layout on the footer bar.
- To change the view from Horizontal to Vertical, click the Horizontal button on the footer bar.

Topology Help





The icons shown in FIGURE 4-3 and FIGURE 4-4 symbolize the various storage devices that Storage Automated Diagnostic Environment monitors. Inside each storage device icon are boxes that represent:

- Disks inside the arrays
- Ports
- Other components

The "other" category includes a variable set of boxes that represent components of the enclosures that are monitored. See TABLE 4-1 for a description of 'other' components

Storage Device	'Other' Components Monitored
Sun StorEdge T3 and T3+ array	L=LUN, M=Midplane, P=Power, V=Volume
Sun StorEdge network FC switch-8 and switch-16 switch	F=Fan, P=Power, T=Temp
Sun StorEdge A5000 array	F=Fan, B=Backplane, P=Power
Sun StorEdge A3500FC array	L=LUN, B=Battery
Virtualization Engine	L=LUN
Host	O=Other, L=LUN

TABLE 4-1"Other" Components

Note – Links that are labeled 'mpxio' in the Topology could indicate one of the following:

If a RED "mpxio" label is displayed in the Topology, a Sun StorEdge Traffic Manager failover has been discovered on this path.

If a GREEN "mpxio" label is displayed in the Topology, a Sun StorEdge Traffic Manager has occurred on this path, but was later corrected.

Monitoring Logs

This section discusses the following log monitoring functions you can perform using Storage Automated Diagnostic Environment:

- "To Review the Most Recent Entries on a Host" on page 85
- "To View an Event Log for a Host" on page 86
- "To View Alert Logs" on page 88
- "To Display Agent Errors" on page 89

▼ To Access the Log Page

- 1. Click the Monitor link on Storage Automated Diagnostic Environment main window.
- 2. Click the Log link.

The Log window is displayed.



▼ To Review the Most Recent Entries on a Host

You can review the content of the /var/adm/messages and T3 and T3+ message log files from a host. Log entries are displayed from the end of the file going back; the most recent entries are shown first. The Storage Automated Diagnostic Environment must be functioning properly on each host for the /var/adm/messages function to work.

Host's log files HostName Type HostID Host IP Log T3 Log	/var/adm/messages [Help						
HostName Type HostID Host IP Log T3 Log	Host's log files						
	HostName	Type	HostID	Host IP	Log	T3 Log	
<pre></pre>	<hostname-1></hostname-1>	Master	<hostid-1></hostid-1>	<ip-1></ip-1>	<u>/messages</u> /messages.0	/messages.t3	

• Choose a host from the HostName column and click the corresponding /messages link.

The descending list of /var/adm/messages on the local host is displayed.

/var/adm/messages on Local Host/diag245.central.sun.com (size: 2K) [<u>Display 40K</u>]	[<u>Refresh</u> <u>List</u>]
Jan 21 17:34:58 diag245.Central.Sun.COM /usr/dt/bin/ttsession[3036]: [ID 862433 daemon.error] child (3240) exited due to signal 1	
Jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.notice] ASC: 0x29 (power on, reset, or bus reset occurred), ASCQ: 0x0, I	FRU: 0x0
Jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.notice] Sense Key: Unit Attention	
Jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.notice] Vendor: SUN Serial Number: 0001610111	
Jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.notice] Requested Block: 0 Error Block: 0	
Jan 21 08:45:04 diag245.Central.Sun.COM Error for Command: Error Level: Informational	
Jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.warning] WARNING: /pci@f,2000/pci@2/SUNW,qlc@4/fp@0,0/ssd@w50	020f2300003ee5,1 (ssd581):
Jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.notice] ASC: 0x29 (power on, reset, or bus reset occurred), ASCQ: 0x0, I	FRU: 0x0
Jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.notice] Sense Key: Unit Attention	
Jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.notice] Vendor: SUN Serial Number: 0001610110	
jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.notice] Requested Block: 0 Error Block: 0	
jan 21 08:45:04 diag245.Central.Sun.COM Error for Command: Error Level: Informational	
jan 21 08:45:04 diag245.Central.Sun.COM scsi: [ID 107833 kern.warning] WARNING: /pci@f,2000/pci@2/SUNW,qlc@4/fp@0,0/ssd@w50	020f2300003ee5,0 (ssd582):
Jan 21 08:45:02 diag245.Central.Sun.COM fp: [ID 517869 kern.warning] WARNING: fp(4): N_x Port with D_ID=114800, PWWN=210000e0	8b02732a reappeared in
fabric	
Jan 21 08:45:01 diag245.Central.Sun.COM fp: [ID 517869 kern.warning] WARNING: fp(4): N_x Port with D_ID=114800, PWWN=210000e0	8b02732a disappeared from
fabric	
Jan 21 08:45:01 diag245.Central.Sun.COM qlc: [ID 686697 kern.info] NOTICE: Qlogic qlc(4): Link ONLINE	
Ian 21 08/44/52 diag245 Central Sun COM dic: [ID 585597 kerninfo] NOTICE: Ologic dic(4): Loon OFFLINE	

▼ To View an Event Log for a Host

1. From the Monitoring Log menu, select Event Log.

Event Log [Help]					
Category		Event	Filter	·1	and Filter2
A	II 🗆	All	• I		Ĭ
Select a H	lost				
Select		HostName	Type	HostID	Host IP
×	<hostname></hostname>		Master	<hostid-1></hostid-1>	<ip-1></ip-1>
		Sea	rch Log		

2. To search for events on a given device type, choose the device from the Category pull-down menu.

All is the default.

Pull-down menu options include Sun StorEdge T3 array Events, Sun StorEdge T3 array Message Events, Sun StorEdge A5000 array, Sun StorEdge A3500FC array, Sun StorEdge network FC switch-8 and switch-16 switch, Virtualization Engine, FC Tape, Agent, Message Events, Host Events, and Topology Events.

3. Choose an Event from the Event pull-down menu.

All is the default.

Pull-down menu options include Communication Events, Discovery, Alarm Event, Removal Event, Diagnostic Test, Topology Event, Audit Event, State Change Event, Agent Event, Backup Event, Patch Event, Link Event, Location Change Event, and Statistical Event.

- 4. To narrow the size of the Event Log, type the IP address for which you want to view the Event Log into the Filter1 and/or Filter2 text box.
- 5. Select one or more HostNames and click Search Log.

The corresponding event log window is displayed.

EventLog Local Host (diag	245.	.central.sun.com)	[<u>Back</u>] [<u>Help</u>]
Time	Sev	Event	Description
2002-01-22 00:30:09		host.backup	NS Agent backup
2002-01-21 08:45:03		message.LogEvent driver. LOOP_OFFLINE	Found 1 'driver.LOOP_OFFLINE' error(s) in logfile: /var/adm/messages on diag245.central.sun.com (id=80bd1196):
2002-01-21 08:45:03		message.LogEvent driver. LOOP_ONLINE	Found 1 'driver.LOOP_ONLINE' error(s) in logfile: /var/adm/messages on diag245.central.sun.com (id=80bd1196):
2002-01-21 08:45:02	2	message.LogEvent driver. Fabric_Warning	Found 2 'driver.Fabric_Warning' warning(s) in logfile: /var/adm/messages on diag245.central.sun.com (id=80bd1196):
2002-01-21 00:30:09		host.backup	NS Agent backup
2002-01-20 00:30:10		host.backup	NS Agent backup
2002-01-19 00:40:24		t3message.LogEvent MessageLog	Error(s) found in logfile: /var/adm/messages.t3 on diag245.central.sun.com (id=80bd1196):
2002-01-19 00:40:24	2	t3message.LogEvent MessageLog	Warning(s) found in logfile: /var/adm/messages.t3 on diag245.central.sun.com (id=80bd1196):
2002-01-19 00:29:59		t3message.LogEvent MessageLog	Error(s) found in logfile: /var/adm/messages.t3 on diag245.central.sun.com (id=80bd1196):
2002-01-19 00:26:26		host.PatchInfo	New Patch and Package Information generated
2002-01-19 00:26:26		host.backup	NS Agent backup
2002-01-19 00:26:19		topo.DiscoveryEvent SAN_Topology	Topology event with 1 host(s), 2 switch(es)/VEs and 1 Storage Array(s)
2002-01-19 00:26:04		agent.AgentInstallEvent host	Agent on host diag244.central.sun.com 83061b3a was Added to the Config File
2002-01-19 00:26:03		<u>agent.AgentInstallEvent host</u>	Agent on host diag245.central.sun.com 80bd1196 was Added to the Config File
Done			· · · · · · · · · · · · · · · · · · ·

FIGURE 4-5 Example of Event Log Messages on Local Host

▼ To View Alert Logs

1. From the Monitoring Log window, select Alert Log.

Category		Event	Event			Severity		
A			All			All		
elect a F	lost			-				
Select	HOS	tivame		Type	HOSTID		HOSTIP	
V	<hostname-1></hostname-1>			Master	<hostid-1></hostid-1>	<ip-1></ip-1>		

- 2. Customize the report by using the following pull-down menus:
 - Category—All is the default.

The device types supported by the current Storage Automated Diagnostic Environment Agent version are displayed in the pull-down menu.

• Event—All is the default.

The Event types supported by the current Storage Automated Diagnostic Environment Agent version are displayed in the pull-down menu.

■ Severity—All is the default.

Options that are displayed in the pull-down menu include System Down, Critical (Error), Alert (Warning), and Caution (Information).

Note – The intent of this log is not to view the content of the Alert, but rather to view the list of Alert types that have been generated. You can obtain the actual content by scanning through the appropriate message logs or through the email that was sent for each notification.

▼ To Display Agent Errors

The Storage Automated Diagnostic Environment System Errors window displays system errors that have occurred on a given host.

1. To access the Storage Automated Diagnostic Environment System Errors window, click Agent Errors on the Monitor Log window.

System Errors [Help]										
This section displays system errors generated by the Storage Automated Diagnostic Environment Agent during execution. Select a host to see the system error log of the agent on that host. System Errors are usually related to a file security problem, perl execution errors and other system errors. The agent can usually recover from these errors but will have to skip a part of its normal execution.										
Select an existing Host										
Host	Hostid	Location								
Master	808acd2e									
crash3.central.sun.com	80c9a11a	crash3.central.sun.com								

2. Select an existing host.

The corresponding Storage Automated Diagnostic Environment Agent Error Logfiles window is displayed.
Diagnostics

Selecting a User Interface

You can run the Storage Automated Diagnostic Environment tests either from the Storage Automated Diagnostic Environment graphical user interface (GUI) or from the command line.

TABLE 5-1 describes the basic differences between the user interfaces.

Interfaces	Description
GUI Window	 You can select tests and test options inside the Storage Automated Diagnostic Environment GUI's Topology section in one of two ways: While in the Topology view, point to a device or Host and click the right mouse button. Select a test from the Test from List window.
Command line	You run each test individually from a shell tool command line. Each test description in this book contains the corresponding command-line syntax.
	Note: You must log into the appropriate Host or Slave for testing.

TABLE 5-1 Storage Automated Diagnostic Environment Diagnostics User Interface Differences Differences

Diagnostics Test Rules

Certain tests have limitations and cannot be run with other tests. The following rules exist with Storage Automated Diagnostic Environment diagnostic tests:

- All tests are offline tests. All devices and paths must be quiesced prior to invocation of diagnostic tests.
- a5ksestest cannot be run concurrently with the qlctest or the socaltest.
- qlctest (which tests the Qlogic 2200 HBA) cannot be run concurrently with any child test (while any attached device is running).
- Switchtest can only run on one port on a single switch instance at a time.

Note – Any attempts to circumvent diagnostic test rules will cause pop-up warnings to be displayed.

Running Diagnostic Tests From the Command Line

In some cases it is more convenient to run a single Sun StorEdge diagnostic test from the command line rather than through a Storage Automated Diagnostic Environment interface.

When running a test from the command line, you must specify all test options in the form of command-line arguments. Standard arguments are common to all tests. Refer to TABLE 5-2 for details.

The standard syntax for all tests is as follows :

```
% testname [-scruvdtelnf] [-i number] [-w number][-o test-specific-arguments]
```

Standard Command-Line Arguments

TABLE 5-2 describes the standard command-line arguments.

Argument	Description
-u	Displays command-line Usage information.
-v	Runs the test in Verbose mode and displays messages with more detailed information about the testing process. The default is False.
-f	Runs the test in full Functional test mode. This mode assumes that the test has complete control of the device being tested. The default is False.
-0	Indicates that the Options and arguments that follow are test-specific.

 TABLE 5-2
 Standard Command-Line Arguments

Note – All options must be listed in a quoted list following the standard $-\circ$ argument. The options must be separated by a pipe (|); for example:

command_name -o "dev=*path* | ..."

Running Diagnostic Tests From the GUI Window

If you run the diagnostic test from the Storage Automated Diagnostic Environment main window, you can easily access test configuration, control, and results using the buttons in the dialog boxes. The test parameter options, however, are unique for each test and are illustrated in the individual sections with each test in this chapter.

- ▼ To Access the Diagnostic Tests
 - 1. Click the Diagnose link in the Storage Automated Diagnostic Environment main window.
- 2. Click the Test link in the Diagnose page.

SUN.	Storage Automated Diagnostic Environ 2.0.6.001 diag245.central.sun.com	ment Maintenance Monitor Diagnose Report Utilitie Tests Tools
Diagnostics - <u>Test from Topology</u>		Diagnostics
- <u>Test from List</u> - Test Manager	Diagnostics	
- <u>Test Mqr (popup)</u> - <u>Set Defaults</u>	< Use the buttons on the left to go	to the appropriate "Test" function.
– <u>Test Archives</u>	[Test from Topology]	Execute test from the Site Topology
	[Test from List]	Execute test from a list of physical devices.
	[Test Manager]	View and control test execution.
	[Test Defaults]	Set the default options for each available test.
	[Test Archives]	Review the results of archived tests.

You can run Storage Automated Diagnostic Environment diagnostic tests from the test list or from a topology. The functional tests are designed to test the target FRU and operate on in-band or out-of-band data paths. The Storage Automated Diagnostic Environment will cause the test to be run on the appropriate Host.

Note – You can invoke the Link Test by right-clicking on the link displayed in the topology; however, the Link Test functionality is not available by using the Test from List functionality.



FIGURE 5-1 Link Test

While the Link Test is running, you can access Test Manager to view the status of the test.

-			Netscape: Test Manage	er		
Test Ma	nager				[<u>Running</u> <u>All</u>	Tests Close]
Host	Process	<u>Command</u>	Target	– <u>Status</u>	<u>Start</u>	Select
diag245	<u>ST-19425</u>	linktest	switch:100000c0dd00b170:2	Running	01-23 10:40:46	
Page: 1 of	1			Archive-9	Selected Delet	e-Selected

FIGURE 5-2 Link Test Running in Test Manager

Once the test has completed its run, an email message, similar to the following message, will be sent to the Email recipient that was specified in Link Test.

```
Date: Wed, 23 Jan 2002 10:44:49 -0700 (MST)
From: StorageAutomatedDiagnosticEnvironment@diagxxx.Central.Sun.COM
Subject: linktest on diagxxx.central.sun.com
To: undisclosed-recipients:;
Test : linktest
Host : diagxxx.central.sun.com
Target : switch:100000c0dd00b170:2
Status : PASSED
Device(s) :
#OUTPUT
running on diagxxx.central.sun.com
linktest started on Fibre Channel interconnect: switch to t3
switchtest started on switch 100000c0dd00b170 port 2
Estimated test time 14 minute(s)
switchtest completed successfully
Skipping t3.
Required Fibre Channel loopback test capabilities are not supported by this
device. linktest completed on Fibre Channel interconnect: switch to t3
```

Test from Topology

Storage Automated Diagnostic Environment's implementation of diagnostic tests verify the operation of all the user-selected components. Tests are selected from a graphical view of the system's topology. The Storage Automated Diagnostic Environment Graph view shows the physical topology of a system or merged system. Using the Topology view, you can select specific subtests and test options. The monitoring status of devices and links appears both in the test topology view and in the list view.

To View the Topology

- 1. Select a Host from the Host pull-down menu.
- 2. To narrow the size of the topology, select an address from the Filter pull-down menu and click GO.
- 3. To zoom in closer, right click on the screen's white space. You can zoom to 100%, 83%, 66%, or 50% by selecting the value with your left mouse button.



4. Using your right mouse button, select the device.

A list of test options appears.

5. With your left mouse button, select the test you want to run on the device. The applicable test is displayed in the right pane.

Notes

- 1. The Storage Automated Diagnostic Environment Link Test enables FRU isolation for Fibre Channel devices. You can invoke the Link Test from the Topology view by selecting the link.
- 2. In-band tests can be run from the Test from Topology view with no restrictions.
- 3. An out-of-band test (that is, t3ofdg, t3volverify, switchtest, or vediag) can be run from the Test from Topology view under the following conditions:
 - It is run from the monitoring Host view
 - It is run from the merged topology view
- 4. If you need help understanding the various link colors, icons, or symbols found in the Topology, please refer to "Topology Help" on page 82.
- 5. To specify email addresses to where test results are sent, use the Set Defaults screen. Refer to "Test Defaults" on page 103.

Test from List

Storage Automated Diagnostic Environment's implementation of diagnostic tests verifies the operation of all the user-selected components. Tests are selected from a list of physical devices.

The Test from List view shows the devices and their associated tests. This list also includes warnings and errors reported by the monitoring agents. You can sort by Host, device type, test type, and device status. You can select options for a specific device or choose multiple devices or all devices. In addition, you can select the number of passes for each test.

▼ To Invoke the Test Option Pane for a Particular Device

Using the Test from List window shown in FIGURE 5-3, select the diagnostic test.

- 1. Select a Host from the Host pull-down menu.
- 2. From the Type pull-down menu, select a specific device, multiple devices, or click All to display all devices.
- 3. Click GO.

A list that is customized to your criteria is displayed.

4. To run diagnostics, select the test name link in the Run Test column.

The appropriate test window is displayed in the right frame.

ale Sun	Storage Automated Di	agnostic Environme	ent	Maint	enance Monitor Diagnose F
microsystems	2.0.06.010 bradster.central.s	un.com			Tests Tools
Diagnostics	Test from List			[<u>Help</u>]	
- <u>Test from Topoloq</u> ▶ <u>Test from List</u> -Test Manager	Host: san:MASTER 🗖	Type: All 🗖	System: <mark>I</mark>	GO	
– Test Mar (nonun)	+ <u>System</u>	Component	<u>Status</u>	Run Test	
	a5k:A	port1		<u>A5000-SES</u>	
- <u>Set Defaults</u>		All disks	[<mark>□_comp</mark>	<u>A5000–Disk</u>	
– <u>Test Archives</u>	a5k:Bart	port0	_	<u>A5000-SES</u>	
		All disks	[<mark>L comp</mark>	<u>A5000–Disk</u>	
	a5k:c	port1		A5000-SES	
		All disks	[<mark>L comp</mark>	<u>A5000–Disk</u>	
	host:crash3	hba0 / soc0_0		SOCAL	
		hba1 / soc0_1		SOCAL	
		hba2 / socU_U		SOCAL	
		hba4 (soc0_0		SOCAL	
		hha5 / soc0_0		SOCAL	
		hba6 / soc0_0		SOCAL	
		hba7 / soc0_1		SOCAL	
		hba8 / qlc119	<u> Iink</u>	<u>QLC</u>	
		hba9 / qlc119	<u> link</u>	QLC	
		hba10 / qlc119	<u> Iink</u>	QLC	
		hba11 / qlc119		QLC	
	switch:172.20.104.11	port4	<u>link</u>	Sun-Switch	
		i port6	i <mark>U</mark> link	Sun-Switch	
	Page: 1 of 2 [First Prev	Next Last]			

FIGURE 5-3 Test from List window

Test Manager

Storage Automated Diagnostic Environment's integration with diagnostic tests implements Test Manager to track and control the progress of the user-selected tests.

The test output view allows you to view test output for active and completed tests in the test monitor view.

Note – All tests are keyed by the unique process identifier (PID) that Test Manager assigns during test invocation.

Test Ma	inager				[]	Running <mark>All Te</mark> s	<u>sts</u>]
<u>Host</u>	Process	<u>Command</u>	<u>Target</u>	– <u>Status</u>	<u>Start</u>	Select	
crash3	<u>ST-6527</u>	t3ofdg	t3-104-10	Done:ERR	12-05 21:45:11		
crash3	<u>ST-6216</u>	qlctest	crash3.central.sun.com	Done:ERR	12-05 21:26:15		
crash3	<u>5T-5992</u>	qlctest	crash3.central.sun.com	Done:ERR	12-05 21:14:58		
crash3	<u>5T-8232</u>	linktest	switch:100000c0dd0085c3:8	Done:ERR	12-05 23:47:47		
crash3	<u>5T-7931</u>	t3ofdg	t3-104-10	Done:OK	12-05 23:25:25		
crash3	<u>ST-6729</u>	t3volverify –v	t3-104-10	Done:OK	12-05 21:57:23		
crash3	<u>ST-6388</u>	t3test	t3-104-10	Done:OK	12-05 21:37:28		
crash3	<u>5T-6260</u>	switchtest	172.20.104.11	Done:OK	12-05 21:27:41		
crash3	<u>ST-5962</u>	socaltest	crash3.central.sun.com	Done:OK	12-05 21:13:58		
crash3	<u>ST-8616</u>	linktest	switch:100000c0dd008869:16	Done:OK	12-06 00:01:09		
crash3	<u>5T-8377</u>	linktest	switch:100000c0dd008869:16	Done:OK	12-05 23:52:13		
crash3	<u>ST-5914</u>	a5ktest	Bart	Done:OK	12-05 21:11:34		
crash3	<u>ST-5867</u>	a5ksestest	A	Done:OK	12-05 21:09:46		
Dogo 1 of	: 4			Arc	hive-Selected	Delete-Selecte	

Page: 1 of 1

Storage Automated Diagnostic Environment runs the device tests in a distributed fashion. The Storage Automated Diagnostic Environment Master calls the proper Slave Host to begin tests.

For test processes, the states can be:

- Waiting—you must answer a question for the process to continue. Click the PID link.
- Done:OK—the process ran with no errors.
- Done:ERR—the process failed but continues to run.
- Done: Aborted

Note – When the link test is in the *Waiting* state, user intervention is required, which may involve replacing the cables or the gigabit interface converters (GBICs).

▼ To Archive or Delete Tests

Storage Automated Diagnostic Environment's implementation of diagnostic tests enables you to view archived diagnostic logs saved by the Test Manager.

1. Click the Select checkbox for the test from the Test Manager window.

2. Click the Archive-Selected (to archive) or Delete-Selected (to delete) checkbox.

The Archived Tests view lists all archived tests and their associated output based on the criteria you specify.

Diagnostic Tests Available in the Storage Automated Diagnostic Environment

Test options exist for each individual test. The Set Test Defaults options screen enables you to change the default options for the following diagnostic tests:

- "Sun StorEdge A3500FC Array Test (a3500fctest)" on page 104
- "Sun StorEdge A5000 and A5200 Array Enclosure Test (a5ksestest)" on page 107
- "Sun StorEdge A5000 and A5200 Array Test (a5ktest)" on page 109
- "Sun StorEdge FC Tape Test (fctapetest)" on page 111
- "Sun StorEdge PCI Fibre Channel-100 Host Adapter Test (ifptest)" on page 113
- "Sun StorEdge PCI Dual Fibre Channel Host Adapter Board Test (qlctest)" on page 115
- "Sun StorEdge SBus Fibre Channel-100 Host Adapter Board Test (socaltest)" on page 118
- "Sun StorEdge Network FC Switch-8 and Switch-16 Switch Test (switchtest)" on page 123
- "Brocade Silkworm Test (brocadetest)" on page 121
- "Sun StorEdge T3 and T3+ Array Test (t3ofdg)" on page 125
- "Sun StorEdge T3 and T3+ Array Test (t3test)" on page 127
- "Sun StorEdge T3 and T3+ Array Test (t3volverify)" on page 130
- "Virtualization Engine Diagnostic Test (vediag)" on page 132
- "Sun StorEdge VELUN Test (veluntest)" on page 134

Test Defaults

Storage Automated Diagnostic Environment's implementation of diagnostic tests enables you to set default options to be applied to user-selected tests.

Note – The Verbose Mode	option is o	common t	o all tests.
-------------------------	-------------	----------	--------------

Set Test De	efaults		[<u>Help</u>]
Defaults	A3500FC-LUN	A5000-SES A5000-Disk Brocade-S	<u>Switch FC-Tape IFP QLC </u>
<u>SOCAL</u>	Sun-Switch	<u>T3-ofda T3-LUN T3-VolVerif</u>	y <u>VE-test</u> <u>VE-LUN</u>
Defai	ılt Options		
	Verbose Mode:	X	-v
	Send Results to Email:	Ι	#EMAIL
L			
		Update	

FIGURE 5-4 Storage Automated Diagnostic Environment Diagnostics Set Test Defaults

Default Options

 TABLE 5-3
 Storage Automated Diagnostic Environment Diagnostic Tests Default Option

ltem	Description
Verbose Mode -v Runs the test in verbose mode and displays messages detailed information about the testing process. The d ON.	
Send Results to Email	Enter email addresses to where the test results need to be sent. An entry in Send Results to Email from the Default Options screen sends the results of all tests. To send results to email recipients on an individual test, access that test's Default Option screen.

Note – Email may not be sent if the system is not properly configured to send mail to the recipient. This is primarily evident in service processor environments where the service processors are on a subnet and there is no gateway to the intended recipient. Please refer to the *Sun StorEdge 3900 and 6900 Series Installation and Service Manual* for more information.

Storage Automated Diagnostic Environment Tests

This section describes the Storage Automated Diagnostic Environment Diagnostic tests that are available from the Storage Automated Diagnostic Environment GUI.

Sun StorEdge A3500FC Array Test (a3500fctest)

The a3500fctest(1M) test verifies the functionality of the Sun StorEdge A3500FC array using four subtests: Media, File System, Asynchronous I/O, and Write/Read Device Buffer, that are described in TABLE 5-4.

Subtest	Description
Media	Verifies LUN media by reading data from the LUN. The Media subtest treats a LUN as one large chunk of contiguous data.
File System	Verifies the LUN system's integrity. The File System subtest exercises the partition being tested to determine if it is mounted. If the partition is not already mounted or premounted, the test is blocked. The test opens two temporary files (of the size specified on File System File Size) and performs a read/write test.
Asynchronous I/O	Uses the asynchronous read/write feature of the Solaris LUN driver to exercise the LUN. In read-only mode, the test sends a maximum of four asynchronous read packets, each with a random size and a random offset, into the selected partition. The test then waits for all outstanding I/O activity to complete before issuing another round of packets. This process continues until the whole area is tested. In read/write mode, one write packet is issued in every four read packets as a spot check of the write operation.
Write/Read Device Buffer	This test verifies the Fibre Channel loop by performing a pattern test. If the Write/Read Device Buffer subtest fails on a particular device, there is a problem with an upstream Fibre Channel component, that might not be on the actual device where the test failed.

TABLE 5-4a3500fctest Subtests

Note – An instance of a3500fctest is present for each LUN. The File System subtest can be run only if the selected partition is mounted.

a3500fctest Options		
Select Pattern Type:	critical 🗆	selectpattern
User Defined Pattern:	jūx7e7e7e7e	userpattern
Check RDLS Counts:	C:Enable 🖲:Disable	checkrdls
W/R Device Buffer Test:	€:Enable ⊂:Disable	wrdevbuf
W/R Device Buffer Iterations:	Ĭ1000	wrdevbufiterations
Test Media:	€:Enable ⊂:Disable	rawsub
Partition:	0 🗖	partition
Media Test Method:	SynclO+AsynclO 🗖	method
Media Coverage(%):)	rawcover
Media Transfer Size:	32KB 🗆	rawiosize
Test File System:	C:Enable :Disable	fstest
File System File Size:	512KB 🗖	fssize
File System Transfer Size:	512B 🗆	fsiosize
Number of passes:	1 🗆	#PASSES
General Options:		
Verbose Mode:	2	-v
Send Results to Email:	Ι	#EMAIL
Update Reset		

FIGURE 5-5 a3500fctest Test Options Dialog Box

a3500fctest Test Options

TABLE 5-5 describes the test options for the a3500fctest test.

TABLE 5-5a3500fctest Options

Option	Description
Select Pattern Type	Enables you to choose user, critical, or all pattern options
User Defined Pattern	Defines the pattern specified by the user
Check RDLS Counts	Monitors the read link status (RDLS) counts
W/R Device Buffer Test	Enables or disables the write/read actions of the Device Buffer Test Note: The Sun StorEdge A3500FC array firmware level must be at 3010360 or greater to support the W/R Device Buffer Test.
W/R Device Buffer Iterations	Specifies the number of times W/R are done to the internal buffer of the Sun StorEdge A3500FC array
Test Media	Enables or disables the Media subtest
Partition	Specifies the partition for the Media subtest. If a partition is mounted, its mount point is appended after the partition number, such as $1(/usr)$, where 1 is the partition number and $/usr$ is the mount point.
Media Test Method	Enables or disables the Media Test Method (SyncIO or AsyncIO)
Media Coverage (%)	Tests all or part of a partition (in percentages)
Media Transfer Size	Specifies the transfer size of the Media subtest
Test File System	Enables or disables the File System subtest
File System File Size	Creates a file twice the size of what is specified
File System Transfer Size	Specifies the transfer size of the File System subtest

Sun StorEdge A5000 and A5200 Array Enclosure Test (a5ksestest)

The a5ksestest(1M) test provides configuration verification, fault isolation, and repair validation of the disks in the Sun StorEdge A5000 array. It tests both Sun StorEdge A5000 14- and 22-slot disk enclosures.

The a5ksestest test detects all Sun StorEdge A5000 arrays that are connected to the HBA and collects relevant configuration information.

Note – The Sun StorEdge A5000 array was formerly known as the Sun Enterprise Network Array[™]. The a5ksestest test tests both versions of this disk array subsystem.

Start Test [Help]	
	A5K-SES
a5ksestest a5k 'A'	Start Test
Run on Host	crash3.central.sun.com
Name	: a5k:5080020000084f40
Box	: A
Product	: SUN StorEdge-A5200
Select :	ses—B (p1) ses22 📥 ses—B (p1) ses23 🕎
Disk Access:	(«Enable (:Disable
Pass Delay:	<u>j</u> 30
General Opt	ions:
Verbose Mode:	2
Send Results to Email:	þrad.derolf@sun.com
Start Test	Display Command&args

FIGURE 5-6 a5ksestest Test Options Dialog Box

The diagnostic software attaches at least one instance of a5ksestest whenever a Sun StorEdge A5000 SCSI enclosure services (SES) device is found. Normally, two instances occur for each path to a Sun StorEdge A5000 array.

Note – To prevent test failures, do not run the a5ksestest and socaltest tests at the same time.

a5ksestest Test Options

TABLE 5-6 describes the test options for the a5ksestest test.

st Options

Option	Description
Disk Access	During testing, each disk is accessed through each active connection leading to that disk. The a5ksestest test opens partition 2 on the disk and reads 512 bytes of raw data.
Pass Delay	Specifies the number of seconds between passes. The default is 30 seconds.

Sun StorEdge A5000 and A5200 Array Test (a5ktest)

The <code>a5ktest(1M)</code> test verifies the functionality of the Sun StorEdge A5000 array using five subtests: Media, File System, Asynchronous I/O, Write/Read Device Buffer, and Self Test.

a5ktest Subtests

The a5ktest subtests are listed in TABLE 5-7.

Subtest	Description
Media	Verifies disk media by reading data from the disk. The Media subtest treats a disk as one large chunk of contiguous data.
File System	Verifies the file system's integrity. The File System subtest exercises the partition being tested to determine if it is mounted. If the partition is not already mounted or pre-mounted, then the test is blocked. The test opens two temporary files (of the size specified on File System File Size) and performs a read/write test.
Asynchronous I/O	Uses the asynchronous read feature of the Solaris disk driver to exercise the disk. In read-only mode, the test sends a maximum of four asynchronous read packets, each with a random size and a random offset into the selected partition. The test then waits for all outstanding I/O activity to complete before issuing another round of packets. This process continues until the whole area is tested.
Write/Read Device Buffer	Verifies the Fibre Channel loop by performing a pattern test. If the Write/Read Device Buffer subtest fails on a particular device, there is a problem with an upstream Fibre Channel component that might not be on the actual device where the test failed.
Disk Self Test	Instructs a device to run its internal diagnostics. If the device fails this test, check the error message for a more detailed description of the error.

TABLE 5-7a5ktest

The a5ktest Test Options dialog box shows all the partitions that are available for testing. The File System subtest can be run only if the selected partition is mounted (described in TABLE 5-7). An instance of a5ktest is present for each disk in a Sun StorEdge A5000 array.

a5ktest a5k 'Carson'	Start Test
Run on Host: dia	g176.central.sun.com
Name: a5k:	508002000001e618
Box: Cars	ion StorEdge AE000
PIOUUCU SON	Storeuge-ASOOO
Select :	f0 c9t64d0s2 f1 c9t65d0s2 f2 c9t66d0s2
Select Pattern Type:	critical 🗖
User Defined Pattern:	j0x7e7e7e7e
Disk Self Test:	⊂:Enable €:Disable
Check RDLS Counts:	⊂:Enable €:Disable
W/R Device Buffer Test:	€:Enable ⊂:Disable
W/R Device Buffer Iterations:	<u>1000</u>
Test Media:	€:Enable _:Disable
Partition:	2 🗆
Media Test Method:	SynclO+AsynclO 🗖
Media Coverage(%):	¥1
Media Transfer Size:	32КВ 🗆
Test File System:	⊂:Enable
File System File Size:	512KB 🗆
File System Transfer Size:	5128 🗖
Number of passes:	1 💷
General Options:	
Verbose Mode:	×
Send Results to Email:	Ι

FIGURE 5-7 a5ktest Test Options Dialog Box

Sun StorEdge FC Tape Test (fctapetest)

The fctapetest(1M) writes a pattern on the tape. fctapetest then rewinds the tape and reads and compares the data just written. The fctapetest file test writes four files to the tape and then reads them back, comparing the data.

The fctapetest diagnostic provides a variety of tests for Sun-supported Fibre Channel tape drives. The fctapetest does not test the tape library. It presumes that the user of the diagnostic either uses tape-library management software or manually inserts tapes into the drives.

fctapetest Test Requirements

If you have a Sun Fibre Channel tape drive in your system, load a blank writable tape (scratch tape) before you start the diagnostic application.



Caution – If you mount a tape containing valid data, that data will be overwritten by the fctapetest diagnostic.

fctapetest Options		
Mode:	C:Readonly :Write/Read m	
Number of Blocks:	Short 🗖	1
Specified Number of Blocks:	25300 s	
Tape Data Density:	Compression 🗖	d
Tape File Mark Test:	C:Enable	ft
General Options:		
Verbose Mode:	M	-v
Send Results to Email:	[brad.derolf@sun.com #EMAIL	
Update		

FIGURE 5-8 fctapetest Test Options Dialog Box

fctapetest Test Options

TABLE 5-8 describes the test options for the fctapetest test.

Option	Description
Mode	If you enable write/read mode, the test first writes to the tape and then reads it back to compare. If you enable read-only mode, the test assumes the tape has already been written and merely reads it. Read-only mode is useful for checking proper head alignment.
Number of Blocks	 The amount of the tape to be tested. The choices are: EOT: The default; Tests the entire tape Long: Tests 70,000 blocks of the tape Short: Tests only the first 1000 blocks Specified: Type the number of blocks to be tested in the number of blocks field.
Specified Number of Blocks	If you select Specified under the Number of Blocks option, you must type the number of blocks you want to test.
Tape Data Density	If you do not want the fctapetest test to run in the compression mode (the default), specify low, for <i>low</i> compression.
Tape File Mark Test	Verifies that the tape drive is able to correctly write and navigate file marks on the tape.

TABLE 5-8 fctapetest Options

Sun StorEdge PCI Fibre Channel-100 Host Adapter Test (ifptest)

The ifptest(1M) test verifies the functionality of the Sun StorEdge PCI Fibre Channel-100 host adapter, which is a single-loop Fibre Channel card with an onboard Gigabit Interface Converter (GBIC).

The ifptest tests the functionality when there are no devices attached to the loop. The driver checks for devices on the Fibre Channel loop. If any devices are detected, the driver blocks any diagnostic commands.

An error message is displayed if the device is attached to storage. If devices are attached to the loop, do not run ifptest. Instead, run the t3test(1M), a3500fctest(1M), a5ktest(1M), or fctapetest(1M) test on the individual devices.

The ifptest test uses the "mailbox" interface to the card, which enables certain firmware operations to be performed that normally would not be available to the application layer.

Start Test [Help]		
<u>QLC</u>	SOCAL	IFP
ifptest	14 control sup cor	
Host dis	a244 central sun co	u vm
Model: SII	NW HitraAX-i2	200
Proden 50		
No	Port/Path availab	ile!
Mailbox		
Loopback	.€:Enable ⊂:Disa	ble
Firmware		
Revision	●:Enable ⊂:Disa	ble
Check:	· · · ·	
Firmware	Grande Obla	6.1.
Test:	Ciblisa	DIE
Check		
Module	.€:Enable ⊂:Disa	ble
Revisions:		
Number of		
passes:		
General Opt	ions:	
Verbose Mode:	¥	
Moue.	[
to Email:	I	
	L'	
Di	splay Command&ar	gs

FIGURE 5-9 IFP Test Options Dialog Box

ifptest Test Options

TABLE 5-9 describes the ifptest Test Options dialog box for different test modes.

TABLE 5-9	ifptest	Options
-----------	---------	---------

Option	Description
Mailbox Loopback Test	Enables or disables the mailbox loopback command. This test writes data patterns into the mailboxes and then reads them back from the output mailboxes and verifies that the data is correct. It is run by default, but it can be disabled.
Firmware Revision Check	Enables or disables the firmware revision check command. This test extracts the firmware revision from the RISC firmware code and verifies it against expected values. It is run by default, but it can be disabled.
Firmware Checksum Test	Enables or disables the firmware checksum command. This command instructs the interface's RISC processor to calculate the current checksum on the microcode and then compare it to the checksum that was loaded in with the microcode. It is run by default, but it can be disabled.
Check Module Revisions	Enables or disables the firmware check module command. This command returns the revision level of several modules on the interface card. Although this test is executed when enabled, the module revision levels are displayed only in verbose mode. It is run by default, but it can be disabled.

Sun StorEdge PCI Dual Fibre Channel Host Adapter Board Test (qlctest)

The <code>qlctest(1M)</code> test tests the functions of the Sun StorEdge PCI dual Fibre Channel host adapter board. The PCI dual fibre board is an HBA that has diagnostic support. This diagnostic test is not scalable.

Note – To run the glctest when connected to storage, enable the Yes option in the "Test if Connected to Storage" field in the glctest dialog box, as shown in FIGURE 5-10.

Start Test	[<u>Help</u>]
	QLC
qlctest host 'diag245.ce	ntral.sun.com' Start Test
Host: diag24 Model: SUNW,	ł5.central.sun.com Ultra–Enterprise
Select :	Board7, Slot0, Port2 qlc3 (17) Board4, Slot0, Port1 qlc6 (10) Board4, Slot0, Port2 qlc7 (11)
Test If Connected to Storage:	C:Yes ●:No
Online Selftest:	€:Enable ⊂:Disable
Mailbox Loopback Test:	€:Enable ⊂:Disable
Firmware Checksum Test:	€:Enable ⊂:Disable
Test 10 bit :	€:Enable ⊂:Disable
Test 1 bit :	●:Enable ○:Disable
Loopback Test:	C:Enable
Loopback Transfer Count:	0x10000 🗆
Loopback Iteration Count:	100 🗆
User Defined Pattern:	j̇́0×7e7e7e7e
Select Pattern Type:	critical 🗖
Number of passes:	1 🗆
General Options	-
Verbose Mode:	×
Send Results to Email: I	
Start Test	Display Command&args

FIGURE 5-10 glctest Test Options Dialog Box

Note the following:

- Do not run customer applications while running glctest, as the test will take priority over customer data requests. Data cannot be accessed while the glctest test is running.
- Do not run other tests while the qlctest test is running. The qlctest test might cause other tests to fail.
- Running the qlctest test can affect the switch counters along with the operation of the Storage Automated Diagnostic Environment Agent.

TABLE 5-10 describes the glctest Test Options dialog box.

Option	Description
Test If Connected to Storage	Runs glotest while connected to storage
Online Selftest	Evaluates the functionality of ISP hardware by performing the following tests: • Transmit FIFO test • Receive FIFO test • SRAM test • Miscellaneous Register tests It runs by default, but it can be disabled.
Mailbox Loopback Test	Loads a series of registers into the input mailboxes on the card and then reads the output mailboxes and compares the results. Verifies that the system side of the card is operating correctly and that the internal data paths are correct. It runs by default, but it can be disabled.
Firmware Checksum Test	Runs an internal checksum test on the installed firmware. This test verifies that the RISC RAM on the card is fully functional and that the installed firmware is still intact. This test also serves as a quick RAM check of the RISC RAM. It runs by default, but it can be disabled.
Internal Loopback Test 10-bit	Performs an internal loopback test within the host adapter ISP hardware at the 10-bit interface. This test is done with data sourcing from the system memory. You select the desired data pattern, transfer count, and iteration count from the Test Options dialog box. It runs by default, but it can be disabled.
Internal Loopback Test 1-bit	Performs an internal loopback test within the host adapter ISP hardware at the 1-bit interface. This test is done with data sourcing from the system memory. You select the data pattern, transfer count, and iteration count from the Test Options dialog box. It runs by default, but it can be disabled.

TABLE 5-10glctest Options

Option (Continued)	Description
External Loopback Test	Performs an external loopback test. This test is done with data sourcing from the system memory and going to the system memory. You select the data pattern, transfer count, and iteration count from the Test Options dialog box. This is an intervention test, because a loopback cable from the transceiver to the QLC receiver of the QLC port must be inserted when testing this port by itself. This subtest can also test the entire Fibre Channel loop when the loop is connected to the storage to be tested. It does not run by default, but it can be enabled.
Loopback Transfer Count	Controls the packet size used in the internal 10-bit, internal 1-bit, and external loopback tests. The default value is 0×10000 .
Loopback Iteration Count	Sets the number of times to loop the internal 10-bit, internal 1-bit, and external loopback tests. The default value is 10.
User Defined Pattern	Uses the user-entered data pattern to loop for the internal 10-bit, internal 1-bit, and external loopback tests. The default value is $0x7e7e7e7e$.
Select Pattern Type	Selects which data pattern to loop for the internal 10-bit, internal 1-bit, and external loopback tests. The default value is critical.

TABLE 5-10glctest Options (Continued)

Sun StorEdge SBus Fibre Channel-100 Host Adapter Board Test (socaltest)

The socaltest(1M) test aids the validation and fault isolation of the Sun StorEdge SBus Fibre Channel-100 host adapter board. In the case of a faulty board, the test tries to isolate the fault to the card, the GBIC module, or the direct memory access (DMA) between the host adapter card and the Host memory.

Start Test	:	[<u>Help</u>]	
<u>SOCAL</u>			
socaltest host 'diag245.central.sun.com' Start Test			
Host: diag245.central.sun.com Model: SUNW,Ultra–Enterprise			
Select :	Board1, Slotd, Port Board1, Slotd, Port Board1, Slot1, Port	0 soc0_0 (0) 1 soc0_1 (1) 0 soc1_0 (2)	
Internal Loopback Test:	⊂:Enable	le	
External Loopback Test:	⊂:Enable	le	
Loopback Frame Test:	€:Enable ⊂:Disab	le	
User Defined Pattern:	IDx7e7e7e7e		
Select Pattern Type:	critical 🗆		
Loopback Iteration Count:	1000 🗆		
Loopback Transfer Count:	0x10000 🗖		
Number of passes:	1 🗆		
General Options:			
Verbose Mode:	V		
Send Results to Email:	Ĭ		
Start Test	Display Comma	und&args	

FIGURE 5-11 socaltest Test Options Dialog Box

Note – Do not run socaltest and a5ksestest at the same time. Otherwise, test failures might occur. Do not run socaltest with a high system load.

In addition to the tests described above, the socaltest test also tests the basic functions of the SOC+ chip, the on-board XRAM, and the Host control buffer by invoking the appropriate tests implemented in firmware.

Note – You cannot run the Internal or External Loopback tests if the port is connected to a disk array.

TABLE 5-11 describes the socaltest Test Options.

Option	Description	
Internal Loopback Test (with no storage attached)	Checks the host adapter card and the DMA with the Host system, as follows:	
	 A frame is created in the host adapter local memory, sent out through the SOC+ transmitter, and internally looped back to the SOC+ receiver. The received data is compared to the original data. A frame is created in the host adapter local memory, sent out through the SOC+ transmitter, and looped back through the SERDES (serializer-deserializer) chip on the host adapter card. The received data is compared to the original data. A frame is created in the Host main memory, transferred through the DMA to the host adapter transmitter, looped back within the SOC+ chip, and transferred from the receiver to the host main memory through the DMA. The received frame is compared to the original transmitted frame, which tests the host memory to the host adapter DMA path. If the board is not connected to storage, the Internal Loopback test is selected by default. External Loopback and Loopback Frame tests are disabled. 	
External Loopback Test (with no storage attached)	Verifies the proper functioning of the GBIC module. A frame is created in the host adapter local memory and is sent out and looped back through the external loopback connector attached to the port. If the External Loopback test is run together with the Internal Loopback test, the DMA path is also tested by creating a frame in host main memory, transferring it to the host adapter through the DMA, looping it back through the external loopback connector, and transferring the received frame back to the host main memory by DMA. By default, this is always disabled.	
Loopback Frame Test (with storage attached)	Sends out a buffer initialized with the selected pattern and compares it to the looped-back frame. It passes if the two match and fails if they do not. If the board is connected to storage, the Loopback Frame test is selected by default. Internal and External loopback tests are disabled.	

TABLE 5-11socaltest Options

Option	Description	
User Defined Pattern	User specified pattern in hexadecimal. The default is 0x7e7e7e7e.	
Select Pattern Type	Applies only to Loopback Frame test. user uses the pattern entered by user. critical runs the 10 most critical patterns for fault detection. all runs the complete list of hexadecimal patterns for fault detection. The all pattern includes the critical pattern. The default is critical, which applies only to Loopback Frame Pattern.	
Loopback Iteration Count	Sets the number of times to loop the internal 10-bit, internal 1-bit, and external loopback tests. The default value is 10.	
Loopback Transfer Count	Controls the packet size used in the internal 10-bit, internal 1-bit, and external loopback tests. The default value is 0x10000.	

TABLE 5-11	socaltest	Options
------------	-----------	---------

In addition to the tests described above, the socaltest test also tests the basic functions of the SOC+ chip, the on-board XRAM, and the Host control buffer by invoking the appropriate tests implemented in firmware.

Brocade Silkworm Test (brocadetest)

The brocadetest(1M) test is used to diagnose Brocade switch devices. The brocadetest process also provides command line access to Brocade Silkworm switch diagnostics. Brocadetest supports testing on all Brocade Silkworm switches that have network access from the testing host.

Brocadetest runs the port diagnostic on connected switch ports. While brocadetest is running, the port statistics are monitored for errors.

Start Test [Hel		[<u>Help</u>]	
Brocade-Switch			
brocadetest '/172.20.67.1	switch 182'	Start Test	
Host: di	ag176.central.sun.	com	
Product: Bi	rocade-8		
WWN: 1)	000006069221e70)	
Select: p p	ort0/g-port ort1/g-port ort2/g-port		
loop iterations: 1	000		
Password: I			
Number of passes:	1 🗆		
General Opti	General Options:		
Verbose Mode:	1		
Send Results to Ĭ Email:			
Start Test Display Command&args			

The brocadetest(1M) options are shown in FIGURE 5-12.

FIGURE 5-12 brocadetest Test Options Dialog Box

Note – The brocadetest(1M) uses a telnet session from which to run the diagnostics. In order for the diagnostics to run correctly, no users can have a telnet session open to the switch while the diagnostics are running. If there is a telnet session open, the brocadetest will fail, indicating that another user may be logged into the switch.

brocadetest (1M) Test Options

TABLE 5-12 describes the test mode options for the brocadetest test.

 TABLE 5-12
 brocadetest
 Test
 Options

Option	Description
Iterations	Specifies the number of iterations the port test should run, between 0 and 1,000,000.
dev	Specifies the port and IP address of the switch to be tested.
ip	Specifies the IP address of the switch to be tested.
port	Specifies the port of the switch to be tested.
password	Specifies the password used by the admin user of the switch. There is no default value.

Sun StorEdge Network FC Switch-8 and Switch-16 Switch Test (switchtest)

The switchtest(1M) test is used to diagnose the Sun StorEdge network FC switch-8 and switch-16 switches. The switchtest process also provides command-line access to switch diagnostics. switchtest supports testing on local and remote switches.

switchtest runs the port diagnostic on connected switch ports. While switchtest is running, the port statistics are monitored for errors.

Start Tes	t [<u>Help</u>]	
FC-Switch		
switchtest switch 'sw-104-11/172.20.104.11' Host: bradster central sup com		
Product: Sw	itch-8	
Select :	port1 / T_Port port2 / F_Port port3 / T_Port	
Transfer Size:	2000 🗆	
Iterations:	100000 🗆	
User Defined Pattern:	IDx7e7e7e7e	
Select Pattern Type:	critical 🗆	
General Options:		
Verbose Mode: 💌		
Send Results to Email:		
Start Test	Display Command&args	

The switchtest options are shown in FIGURE 5-13.

FIGURE 5-13 switchtest Test Options Dialog Box

switchtest Test Options

TABLE 5-13 describes the test mode options for the switchtest test.

 TABLE 5-13
 switchtest Test Options

Option	Description
Transfer Size	Specifies the transfer count for the port test, between 200 and 2000
Iterations	Specifies the number of iterations the port test should run, between 0 and 1,000,000.
User Defined Pattern	Specifies the default pattern, in hexadecimal format, to be used for the port test. You can also enter the hexadecimal pattern to run for the test.
Select Pattern Type	Gives the user the choice of running the one user pattern, critical patterns (10 of the most critical patterns), or all patterns (a complete list of test patterns)

Sun StorEdge T3 and T3+ Array Test (t3ofdg)

The t3ofdg(1M) test runs the internal diagnostics of the Sun StorEdge T3 and T3+ array.

Note – Before you run the t3ofdg test, you must first generate a report for the device against which you are running the test (for example, Sun StorEdge T3 and T3+ arrays). If the number of existing volumes do not match, an error message is displayed.

Start Te	st		[<u>Help</u>]
T3-ofdg		<u>T3-V</u>	olVerify
Warning No password defined for this device. (slr-mi.370-3990-01-e-e1.002974)			
t3ofdg t3't3-104-10' Start Test			Test
Runon Host: b	radste	er.central.su	in.com
Name: t3	8:slr–m	i.370-3990-	01-e-e1.002974
Box: t3 Product: SI	3-104- IN Stor	-10 rEdae—T3	
	5	2490 10	1
T3_Address:	172.3	20.104.10 🗆	
Syslog Dump:	Disal	ble 🗆	
Fast_Test:	Disal	ble 🗆	
Health_Check :	Enab	ole 🗖	
Fast_Find:	Disal	ble 🗆	
Find:	Disal	ble 🗆	
Loop:	all 🗆	1	
User:	Iroot		
Password:	Ĭ		
General Opti Verbose	ons:		
Mode:	×		
Send Results to Email:	I		
Start Tes	t D	isplay Com	mand&args[

FIGURE 5-14 t3ofdg Test Options Dialog Box

Note – This test requires the user to enter the user ID and password options, which are the user ID and password for the Sun StorEdge T3 or T3+ array that is being tested.

t3ofdg Test Options

TABLE 5-14t3ofdg Test Options

Syslog Dump	<i>Enable</i> will dump all syslog entries that have been added, while the test is running, to the screen. <i>ofdg</i> will dump only OFDG lines, added to the syslog, to the screen. The default is <i>Disable</i> .
Fast_Test	<i>Fast_Test</i> performs a fast Go/No Go test of the selected enclosure and loop. It does not attempt to detect bad FRUs. The default is <i>Enable</i> .
Health_Check	<i>Health_Check</i> runs <i>Fast_Test</i> multiple times, one time for each loop per unit. The default is <i>Disable</i> .
Fast_Find	<i>Fast_Find</i> can be used to detect bad loop cards, interconnect cables, and controllers. <i>Fast_Find</i> does not try to isolate down to a single disk port. Run <i>Fast_Find</i> before <i>Find</i> to eliminate loop cards, interconnect cables, and controllers as bad FRUs before the midplane or disks are suspended (which are checked using <i>Find</i>). The default is <i>Disable</i> .
Find	<i>Find</i> performs an extensive Go/No Go test. If loop failures are detected, it automatically initiates the full-loop-fault-isolation diagnostic. The loop fault diagnostic has the capability to detect and isolate a single disk port but is very time-consuming. The default is <i>Disable</i> .
Loop	Loop specifies which loop to test. All tests both loops. The default is All.
Sun StorEdge T3 and T3+ Array Test (t3test)

The t3test(1M) test verifies the functionality of Sun StorEdge T3 and T3+ array logical unit numbers (LUNs) using three subtests: Media, File System, and Asynchronous I/O.

The file system subtest can be run only if the selected partition is mounted. An instance of the t3test test is present for each LUN.

t3test Subtests

TABLE 5-15 describes the t3test subtests.

TABLE 5-15t3testSubtests

Subtest	Description				
Media	Verifies LUN media by reading data from the LUN. The Media subtest treats a LUN as one large chunk of contiguous data.				
File System	Verifies the LUN system's integrity. The File System subtest exercises the partition being tested to determine if it is mounted. If the partition is not already mounted or premounted, the test is blocked. The test opens two temporary files (of the size specified on File System File Size) and performs a read/write test.				
Asynchronous I/O	Uses the asynchronous read/write feature of the Solaris LUN driver to exercise the LUN. In read-only mode, the test sends a maximum of four asynchronous read packets, each with a random size and a random offset into the selected partition. The test then waits for all outstanding I/O activity to complete before issuing another round of packets. This process continues until the whole area is tested. In read/write mode, one write packet is issued in every four read packets as a spot check of the write operation.				

Start Te	st <u>T3-ofda</u>	[<u>Help</u>] <u>T3-VolVerify</u>
t3test t3 '50020f20	100003ee5'	Start Test
Run on H	lost: diag245.cen	tral.sun.com
Na Proc	ame: t3:50020f230 luct: SUN StorEdge	00003ee5 -T3
Select :	c14t50020F2300 c14t50020F2300	003EE5d0s2:Primary 003EE5d1s2:Alternate
Select Pattern Type:	critical 🗖	
User Defined Pattern:	j0x7e7e7e7e	
Check RDLS Counts:	C:Enable 🖲:Disa	able
Check T3 Log File:	€:Enable ⊂:Disa	able
Test Media:	●:Enable ○:Disa	able
Partition:	2 🗆	
Test Method:	SynclO+AsynclO	
Media Coverage(%):	<u>ĭ</u> 1	
Media Transfer Size:	32КВ 🗆	
Test File System:	C:Enable ():Disa	able
File System File Size:	512KB 🗖	
File System Transfer Size:	512B 🗆	
Number of passes:	1 🗆	
General Opt	ions:	
Verbose Mode:	x	
Send Results to Email:	I	

FIGURE 5-15 t3test Test Options Dialog Box

Caution – Before using the diagnostics package to monitor messages from a Sun StorEdge T3 or T3+ array, you must set up the array to mirror its /syslog messages to the Host that is running the diagnostics package. The array messages may be mirrored to the /var/adm/messages.t3 file on the Host.

t3test Test Options

TABLE 5-16 describes the test mode options for the t3test test.

Option	Description
Select Pattern Type	Selects which pattern to run. You will choose the user, critical pattern, or all patterns.
User Defined Pattern	User defined test pattern
Check RDLS Counts	Monitors the read link status (RDLS) counts
Check T3 Log File	Scans Sun StorEdge T3 and T3+ array log file for errors during test
Test Media	Enables or disables the Media subtest
Partition	The partition for the Media subtest. If a partition is mounted, its mount point is appended after the partition number, such as $1(/usr)$, where 1 is the partition number and /usr is the mount point.
Test Method	Enables or disables the Test Method (SyncIO and AsyncIO)
Media Coverage (%)	Tests all or part of a partition (in percentages)
Media Transfer Size	The transfer size of the Media subtest
Test File System	Enables or disables the File System subtest
File System File Size	Creates two files, half the size of what is specified
File System Transfer Size	The transfer size of the File System subtest

TABLE 5-16t3test Options

Sun StorEdge T3 and T3+ Array Test (t3volverify)

The t3volverify(1M) test enables array administrators to execute manual parity checks on existing volumes. Parity checking applies only to RAID 1 and RAID 5 volumes. Check data parity using the t3volverify test before performing tape backup overwrite cycles, approximately once every 30 days.

Before you run the t3volverify test, you must first generate a report for the device against which you are running the test (for example, Sun StorEdge T3 and T3+ arrays). If the number of existing volumes do not match, an error message will be displayed.

Caution – Ensure that system health is in optimal condition before running t3volverify. For example, make sure that no LUNs are under reconstruction, the status of all disks is zero, and other similar conditions are resolved before performing this procedure.

Start Test []						
<u>T3-ofd</u>	9	Į	<u>T3-VolVerif</u>	<u>v</u>		
Warning No password (slr-mi.370-	defir 3990	ned for t −01−e−	his device. f0.003001)			
t3volverify t3 'msp0-t3	b2'		Start Test			
Run on Host:	mspC).centra	l.sun.com			
Name: t	3:slr-	mi.370-	-3990-01-e-	-f0.003001		
Product: S	UN SI	-c30∠ torEdge-	-T3			
T3_Address:	192	2.168.0.4	12 🗆			
VolumeName:	vol	1 🗆				
Fix_Errors:	no					
Rate:	16					
User:	Iroo	t				
Password:	I					
General Opt	ons:					
verbose Mode:	V					
Send Results to Email:	Ι					
Start Te	st	Display	/ Command	&args		

FIGURE 5-16 t3volverify Test Options Dialog Box

Note – This test requires the user to enter the user id and password options, which are the user id and password for the Sun StorEdge T3 and T3+ array that is being tested. The Storage Automated Diagnostic Environment test will use the user id and password that were set up using the Maintain Devices -- > Add T3 window.

t3volverify Test Options

TABLE 5-17	t3vol	verify	Test	Options
------------	-------	--------	------	---------

Option	Description
VolumeName	<i>VolumeName</i> is the name of the volume to verify. Note that the volume name is a name internal to the array and is not seen by the Host.
Fix_Errors	The <i>Fix_Errors</i> option corrects parity errors on RAID 5 volumes and corrects mirrored data errors on RAID 1 volumes. If <i>Fix_Errors</i> is <i>not</i> specified, then t3volverify will report errors but not correct them. If the <i>Fix_Errors</i> option is specified and an error is detected, the t3volverify command will regenerate parity from the existing data on the volume.
Rate	<i>Rate</i> refers to the speed at which the t3volverify is run. The verification rate is n , where n equals any number from 1 to 16. The default rate is 1, which has the minimum performance impact on the Data Host. <i>16</i> has the highest performance impact on the Data Host.

Virtualization Engine Diagnostic Test (vediag)

The vediag(1M) test enables testing of the virtualization engine. TABLE 5-17 describes the test mode options for the vediag test.

Note – Before you run the vediag test, you must first generate a report for the device against which you are running the test (for example, the virtualization engines). If the number of existing volumes do not match, an error message will be displayed.



FIGURE 5-17 vediag Test Options

TABLE 5-18 vediag Options

Option	Description					
target number	Select a target from the target number pull-down menu. The target number enables testing between the virtualization engine and the target device.					
sudiag	The sudiag test checks internal components of the virtualization engine. The default is Enable.					
	Warning: This option reboots the virtualization engine.					
sddiag The sddiag test enables testing between the virtualization engine and a ta device. The default is disable.						
Note: Running the sudia: active before you run thes	g test and sddiag test disables the in-band path. Verify that the in-band path is not e tests.					

Sun StorEdge VELUN Test (veluntest)

In a Sun StorEdge 3900 and 6900 Series Solution configuration, the veluntest(1M) replaces the t3test (1M). Valid LUNs must exist for the veluntest(1M) test to run.

Start Te	est		[<u>Help</u>]
<u>VE-</u>	<u>test</u>	VE-L	<u>UN</u>
veluntest ve 'diag244	-v1 b/2900	006022004193	, Start Test
Runon	Host: diag2	45.central.sun	.com
N	ame: ve:290 Boy: diag24	000602200419 4-v1h	13
Pro	duct: VE		
Select :	c10t2B000 c10t2B000 c10t2B000	006022004193 006022004193 006022004193	d0s2:Alternate d1s2:Primary d2s2:Primary
Select Pattern Type:	critical 🗆	1	
User Defined Pattern:	jOx7e7e7e7	'e	
Test Media:	(•Enable)	:Disable	
Partition:	2 🗆		
Test Method:	SynclO+A	synclO 🗖	
Media Coverage(%):]1		
Media Transfer Size:	32КВ 🗆		
Test File System:	(:Enable	(e:Disable	
File System File Size:	512KB ⊏	1	
File System Transfer Size:	512B 🗆		
Number of passes:	1 0	2	
General Opt	ions:		
Mode:	R		
Send Results to Email:	I		
Start	Test Dis	play Comman	d&args

FIGURE 5-18 veluntest Test Options

veluntest Test Options

TABLE 5-19 describes the test options for the veluntest test.

Option	Description
Select Pattern Type	Selects which pattern to run. You will choose the user, critical pattern, or all patterns.
User Defined Pattern	User defined test pattern.
Test Media	Enables or disables the Media subtest.
Partition	The partition for the Media subtest. If a partition is mounted, its mount point is appended after the partition number, such as $1(/usr)$, where 1 is the partition number and $/usr$ is the mount point.
Test Method	Enables or disables the Media Test Methods (SyncIO and AsyncIO)
Media Coverage (%)	Tests all or part of a partition (in percentages)
Media Transfer Size	The transfer size of the Media subtest
Test File System	Enables or disables the File System subtest
File System File Size	Creates two files, half the size of what is specified
File System Transfer Size	The transfer size of the File System subtest
Number of passes	Specifies the number of times the test will run. The default is 1.

TABLE 5-19veluntest Options

Storage Automated Diagnostic Environment Tools

- ▼ To Access the Storage Automated Diagnostic Environment Diagnostics Tools
- 1. Click the Diagnose link on the Storage Automated Diagnostic Environment main window.
- 2. Click the Tools link on the Diagnose page.



Revision Checking

Storage Automated Diagnostic Environment's implementation of Revision Checking checks the software, hardware, and firmware revisions of all user-selected components.

Revision checking is based on an updatable matrix, which is generated using the PatchPro database. The matrix provides distributed revision checking across the system. Solaris versions and component firmware versions are verified against the matrix.

▼ To Run Revision Checking Report

- 1. Click Start Revision Checking.
- 2. Select a Host from the pulldown menu.
- 3. Select a device associated with the Host from the list of modules.
- 4. In the Email textbox, type an email address or addresses to where Revision Checking results will be sent.
- 5. Click Run Revision Checking.

in Sum	Storage Automated Diagnostic Env	ironment		Maintenance Monit	or Diagnose Repo	rt Utilities He		
microsystems	2.0.06.009 bradster.central.sun.com				Tests Tools			
Diagnostic	Revision Checking					[<u>Hel</u> j		
Tools Revision Check	Select from the list of modules to run Re	vision checking	Go to [Display Repo	rt List] to see the Revisio	n Reports.			
- <u>FC Check</u>	Start Revision Checkin	9	<u>Display</u>	Report List	Current	Report		
	Select options to run Revision Checl	ing						
	Select Host:	Loca						
	Select Matrix:	config-matrix	_3.12 Revision: 3.	12 , created time: 01/22	/2002 🗆			
		All Comp	onents	· - · · _•				
		Sun StorEdge A3500FC Patches and Disk Firmware						
		Sun StorEdge A5000 array Patches and Disk Firmware						
		Host Patches						
	Select Components:	Sun StorEdge HBA Patches						
		🗆 Sun StorE	dge network FC sw	itch–8 and switch–16				
		🗆 Sun StorE	dge T3 and T3+ arr	ay firmware/disk/loo	p/pcu			
		🗆 Sun FC-T	ape Firmware					
		🗆 Virtualiza	tion Engine Firmw	are				
	Email:]brad.derolf@s	un.com					
		R	un Revision Check	sing				

- ▼ To View the Revision Checking Report
 - 1. Click Display Report List.
 - 2. Select a Host from the pulldown menu.
 - 3. Select a device associated with the Host from the list of modules.
 - 4. In the Email textbox, type an email address or addresses to where Revision Checking Report will be sent.
 - 5. Click Run Revision Checking.

Fibre Channel Counters Check

Use the Fibre Channel Counters Check functionality to perform a quick check of the system's transient noise. Run the Fibre Channel Counters Check only on an idle system.

The Fibre Channel Counters Check functionality verifies the health of physical Fibre Channel links by extracting and comparing link status information from all attached storage devices and switches and then displaying error counters.

Fibre Channel Counters Check retrieves the fibre channel Read Link Status counter information from the available devices. This counter information is stored while the Fibre Channel Check program begins the second pass. The second pass begins two minutes after the first pass completes its run. The results of both passes are then compared and changes are displayed in the resulting table.

Note – Fibre Channel Check does not validate connections. Use this functionality only to test for transient noise on the physical Fibre Channel connectors and components. If a system exhibits excessive transient noise, it should be checked for proper cable connections. Diagnostics should then be run on the suspect components.

To Check Fibre Channel Counters

• To run the Fibre Channel Counters check, select a Host from the Select host to check pull-down menu and click START FC Check.



Reports

The Reports section summarizes system information, including agent statistics and system thresholds, on every device that is currently supported by the Storage Automated Diagnostic Environment.

This section also displays device-specific report information on the Sun StorEdge network FC switch-8 and switch-16 switches.

▼ To Access the Reports Page

1. Click the Reports link in the Storage Automated Diagnostic Environment main window.



FIGURE 6-1 Reports window

Agent Statistics

Using the Agent Statistics functionality, you can determine the average time required to run the modules. The information is generated on every run of Storage Automated Diagnostic Environment's host.

- ▼ To Check Storage Automated Diagnostic Environment Statistics
 - 1. Click Agent Stats in the Reports window.

Agent Statistics								[<u>Help</u>]
This report displays, for each agent, the average time required to run the main modules. It can help find bottlenecks in the agent functions. This information is generated and accumulated on every run of the agents on each host.								
Agent Execution time by Host / Module (Mins:Secs)								
Host	::A5K	::HOST	::MESSAGE	::SAN	::S₩ITCH	::T3	T3MESSAGE	Total
crash3.central.sun.com	00:34	00:10	00:00	00:01	00:03	00:46	00:00	01:34
bradster.central.sun.com	00:00	00:00	00:00	00:00	00:00	00:00	00:00	00:00
Average:	00:17	00:05	00:00	00:00	00:01	00:23	00:00	00:47

2. Click an existing host.

The execution time for that host, measured in minutes and seconds, is displayed.

Thresholds List

The Thresholds window displays the thresholds that are used to monitor entries related to I/O interfaces in the /var/adm/messages file.

- Frequency is the number of alerts and hours required to generate a new message.
- Quiet is the quiet time in between messages, which is used to avoid sending too many messages at once.

Thresholds

[<u>Help</u>]

This page lists all threshold rules. The 'Frequency' is the number of alerts/hours required to generate a new message/email. The 'Quiet' value refers to the quiet time in between messages, used to avoid sending too many emails. Text in blue gives information about last message date or about the current number of alerts found against a specific threshold key. This text will appear only after StorADE ran for a while.

Thresholds				
Name	Frequency	Quiet	Type	Desc
fibre.SFOFFL	10 / 24 hours	Ohours	Warning	socal/ifp Offline
fibre.SFOFFLALERT	15 / 24hours	Ohours	Error	socal/ifp Offline
fibre.SCSI_TRAN_FAILED	10 / 4hours	Ohours	Warning	scsi
fibre.SCSI_ASC	10 / 4hours	Ohours	Warning	scsi
fibre.WARNSSD	5 / 24hours	Ohours	Warning	SSD Warning
fibre.ALERTSSD	20 / 24hours	1 hours	Error	SSD Alert
fibre.PFA	1 / 24hours	1 hours	Error	Predictive Failure
fibre.SFCRCWARN	10 / 24hours	1 hours	Warning	CRC Warning
fibre.SFCRCALERT	15 / 24hours	1 hours	Error	CRC Alert
fibre.SFOFFTOWARN	5 / 24hours	1 hours	Warning	Offline Timeouts
fibre.SFDMAWARN	1 / 24 hours	1 hours	Warning	SF DMA Warning
fibre.SFRESET	10 / 24hours	1 hours	Warning	SF Reset
fibre.RETRYELS	10 / 24hours	1 hours	Warning	ESL retries
fibre.RETRYSF	10 / 24hours	1 hours	Warning	SF Retries
fibre.TOELS	10 / 24hours	1 hours	Warning	ELS Timeouts
fibre.SFTOELS	10 / 24hours	1 hours	Warning	SFTOELS Timeouts
fibre.DDOFFL	10 / 24hours	1 hours	Warning	Offlines
fibre.QOFFLINE	1 / 5mins	Omins	Error	Loop Offline
fibre.QONLINE	1 / 5mins	Omins	Error	Loop Online
a3500.CTRL_FIRM	1 / 24hours	24 hours	Warning	Controller firmware version error
vicom.crc	200 / 50mins	10 mins	Error	
vicom.itw	200 / 50mins	10mins	Error	
vicom.link	200 / 50mins	10mins	Error	
vicom.proto	200 / 50mins	10mins	Error	
vicom.signal	200 / 50mins	10mins	Error	
vicom.sync	200 / 50mins	10 mins	Error	
switch.LinkFails	200 / 50mins	10mins	Error	
switch.Total_LIP_Rcvd	200 / 50mins	10mins	Error	
switch.InvalidTxWds	200 / 50mins	10mins	Error	
switch.SyncLosses	200 / 50mins	10 mins	Error	
switch.CRC_Errs	200 / 50mins	10mins	Error	
switch.Prim_Sea_Errs	200 / 50mins	10mins	Error	

Switch Traffic

Use the Switch Traffic functionality to display inframes and outframes for each port of each switch. SAN monitoring must be active and the Storage Automated Diagnostic Environment must have run at least once for this functionality to work properly.

Switch FC Graph

Use the Switch FC Graph functionality to view the values of Fibre Channel counters in graph format.



1. Click Set-Baseline to reset all counters by saving them and displaying only the delta between the new values and the saved values.

After you have set the Set-Baseline field, the graph report only displays counter increments. The report also displays the start time and the duration of the baseline.

2. Click Total to erase the values saved by Set-Baseline and to display the total Fibre Channel counters.

Switch FC Data

Use the Switch FC Data functionality to view the values of Fibre Channel counters.

1. Select a Switch from the FC Counters for Switch pull-down menu to see the values of all six Fibre Channel counters.

After you have set the Set-Baseline field, the Switch FC Data report will display counter increments and the start time and the duration of the baseline, as shown in FIGURE 6-2.

SWITCH Select a Swite [Total] to go I bold any incr	FC Data ch to see the v back to the act ease since tha	alues of all 6 ual values. W t moment (10	Fibre Chanr hen [Set–Bas /100).	nel Counters. seline] is sele	Use [Set–f cted , the	Baseline] t report wil	o reset the I display th	[<u>Help</u>] counters (software reset only). Use e starting date/time and will display in
FC Counters	for Switch:	sw-104-11	/crash3 □	Refresh	10000	0c0dd008	5c3	Set-Baseline Total
All Switch E	rror Counter	s by Port						
Name	Component	Link	Sig	Seq	CRC	Sync	InvTxW	
sw-104-11 sw-104-11 sw-104-11 sw-104-11 sw-104-11 sw-104-11 sw-104-11 sw-104-11	port.01 port.03 port.04 port.05 port.06 port.07 port.08	49 88252 3 0 0 2 0 2	23998 59356 0 0 0 0 0 0	2 30 0 0 0 0 0	1 0 0 0 0 0 0	0 0 3 0 0 2 0	909493 6056191 358 4 4 184 4	

FIGURE 6-2 Switch FC Data Report

2. Click Total to erase the values saved by Set-Baseline and to display the total Fibre Channel counters.

Utilities

The Utilities section contains optional tools you can use for Storage Automated Diagnostic Environment administration.

System Utilities

The sections that follow explain how to perform the administrative tasks.

- "System Time-Outs" on page 147
- "Erase Cache" on page 148
- "Run Agent" on page 149
- "Email Configuration" on page 150
- "Recover Configuration" on page 151
- "Update User Roles" on page 153
- "Clear Login" on page 154

▼ To Access the System Utilities Page

1. Click the Utilities link in the Storage Automated Diagnostic Environment main window.



2. Click the System link.

The System window is displayed.

Be Sum	Storage Automated Diagnostic Environment Maintenance Monitor Diagnose Report Ut					
microsystems	2.0.06.009 bradster.central.sun.cor	n Tests Tools				
System Utilities		System				
- <u>System Time-outs</u> -Frace Cache	The System section contain	is those tools specific to doing administration of Storage Automated Diagnostic Environment.				
- <u>Run Agent</u>	For these and all functions	additional information can be found in the Help section.				
- <u>Email Confiq.</u> - <u>Recover Confiq</u> - <u>Ras Password</u>	< Use the links on the	<<- Use the links on the left to go to the appropriate "System" function.				
- <u>User/Roles</u> - <u>Clear Login</u>	[System Time-outs]	These are the values Storage Automated Diagnostic Environment uses to determine when it has waited long enough for responses from various utilities.				
	[Erase Cache]	This function will erase the last report generated by instrumentation modules for the listed devices. There are NO safeguards for deleting this information.				
	[Run Agent]	Primarily used in test enviornments, this function has the capability to execute the Storage Automated Diagnostic Environment Agent on any specified host from the pull-down.				
	[Email-Configuration]	The Email Configuration option will prompt for a valid email address and send a detailed report of the entire Storage Automated Diagnostic Environment configuration. It will include things like device categories and slave hosts configured to monitor those devices.				
	[Recover Config]	Used to recover the Configuration from a Slave Agent.				
	[Ras Password]	This function allows the administrator of Storage Automated Diagnostic Environment to modify the main password. Refer to the man page <u>save_password(1M)</u> for additional information.				
	[User / Roles]	This section allows the administrator to create new user logins and assign roles to each user.				
	[Clear Login]	Clears the current user and allows the user to re–login as somebody else without having to close the browser. User must press [Cancel] on the "Question Authorization failed retry?" and Reload the browser page to return to the login screen.				



System Time-Outs

System time-outs are the values the Storage Automated Diagnostic Environment Agent uses to ensure it does not spend too much time waiting on a response for commands to return.

▼ To Change System Time-Out Settings

1. Click System Time-outs in the System window.

Agent System Time-outs				
System Time-outs (Seconds)				
Ping : 10	second(s)			
Sun T3 tokens (Http) : [60	second(s)			
SNMP (Sun Switch) : [20	second(s)			
Luxadm (Sun A5000) : [600	second(s)			
RM6 (Sun A3500FC) : 200	second(s)			
Discman (Discovery) : 1200	second(s)			
Slicd (VE) : 180	second(s)			
	Submit Defaults			

- 2. Change the default settings for scheduled time-outs and click Submit.
- 3. To return to the default settings, click Defaults.

Erase Cache

When you select an existing device, the last report in the cache for that device is erased. This forces the Storage Automated Diagnostic Environment Agent to regenerate discovery events.

A Samo	Storage Auto	mated Diagnostic Environment	Maintenance Monitor Diagnos	e Report Utilities Help
	2.0.6.008 bradste	er.central.sun.com	Devices Topology Log	
System Utilities - System Time-outs > Erase Cache - RUN Agent - Email Config. - Recover Config - Ras Password - User/Roles	Use this Doing th compare [Erase Al NOTE: T	Device Cache section to erase the last report generated b is forces rediscovery of the device's instrum to. The device will still be monitored. I Files] deletes the last report generated by there is NO WARNING and NO VERIFICATI	ny instrumentation for a particular device and/or fu nentation upon the next execution, since it will have i instrumentation for ALL listed Devices. ION for these actions. (cache)	[<u>Help</u>] unction. ve no old report to
– <u>Clear Login</u> Scripts – <u>3900/6900 Confi</u> s	Devices Type	Select Host: crash3.central.sun.com crash3.central.sun.com bevice ID (Click to erase)	Display Files	Delete
	a5k	5080020000000188	c c	
	a5k	5080020000084f40	A	
	backup	crash3.central.sun.com		
	switch	100000c0dd006489		
	switch	100000c0dd006fb3	Click the device's	
	switch	100000c0dd0085c3	checkbox to erase the	
	switch	100000c0dd008869	device's cache.	
	t3	slr-mi.370-3990-01-e-e1.002399		
	t3	slr-mi.370-3990-01-e-e1.002974		
	t3	slr-mi.370-3990-01-e-f0.011773		

▼ To Erase a Device's Cache

1. Select a host from the Select Host pull-down menu and click Display Files. A list of devices for the selected host is displayed.

2. Click the device's corresponding Delete checkbox to erase the device's cache. The device is removed from the list.



Caution – There are no safeguard messages for this function. Once you have selected a device, the instrumentation report for that device is immediately erased.

The Erase Device Cache function erases the cache of all the selected devices and the list is removed. Although the device's cache is erased, however, the device will continue to be monitored.

Run Agent

Although the Storage Automated Diagnostic Environment is normally run from the cron facility, the Run Agent function enables you to run Storage Automated Diagnostic Environment manually.

▼ To Run the Agent Manually

Run Agent manually	[<u>Help</u>]				
This option will run the agent program immediately unless it is already running. The agent is normally executed from the cron facility but this option allows you to run it manually. Selected 'Audit all devices' to force the agent to do an extensive probe of all devices. On some devices such as a Sun StorEdge A5000 array, revision information is only extracted once a week to save time. Warning : Diagnostics were running in the last 120 minutes. The agent may return unpredictable results. - Use [RUN] to restrict the agent to message file parsing (recommended). - Use [RUN FORCE] to run all modules anyway (not recommended).					
Select Host: Local 🗖					
Audit all devices:					
	RUN FORCE				

1. Select a host from the Select Host pull-down menu and click RUN.

A summary report displays the status of all the components that are running.

Audit All Devices

On some devices, such as the Sun StorEdge A5000 device, the Storage Automated Diagnostic Environment automatically extracts revision information once a week.

If you click Audit All Devices, you manually force the Storage Automated Diagnostic Environment to run an extensive probe of all devices, rather than having to wait until the next scheduled run.

Email Configuration

▼ To Email Configuration Information

You can forward a detailed report of the Storage Automated Diagnostic Environment configuration by means of email that includes a list of all devices being monitored and the last instrumentation report available for each host. You can also forward topology information.

Email might not be sent if the system is not properly configured to send mail to the recipient. This is primarily evident in service processor environments where the service processors are on a subnet and there is no gateway to the intended recipient. Refer to the *Sun StorEdge 3900 and 6900 Series Hardware Installation and Service Manual* for more information.

Ettale 1

1. Click Email-Configuration in the System window.

Empil Configuration

Eman Configuratio	
This function will email a det along with the last instrumen	alled report of the configuration including a list of all devices being monitored, tation report available on each host.
Email Configuration	
Email Address:	I
Comments:	¥
Select Host Topology	
bradster.central.sun.com:	C
crash3.central.sun.com:	C
	EMAIL

2. Type an email address into the Email Address text box and click EMAIL.

Note – The Email Configuration function enables Sun personnel to view customer information. Note, however, that in order to view the topology in an emailed configuration report, the browser must have access to Sun's Internal Wide Area Network (SWAN).

Recover Configuration

The Recover Configuration function enables you to recover the configuration from the Alternate Master. The configuration includes identification information, along with device, host, and email notification information.

Recover Config	[<u>Help</u>]					
This function allows to recover your of The agent configuration include ident	This function allows to recover your configuration by copying the one from another agent. The agent configuration include identification information along with devices, hosts and email notification.					
Enter Information						
Enter IP Name/Addr of slave:	I					
Recover Config:	X					
Reset slave to this master: 🖬						
	Recover					

Once the system has completed running, a message displays confirming recovery of the configuration file. A message is also sent to the email address specified in the Configuration file.

Recovered the Config file from diag176.central.sun.com

_name=host1 active=N admin_email=yourself@yoursite categories=a5k contract= cust_no=1 customer=christian email=rasagent@nscc.central.sun.com frequency=10 host_adapter=S hostid=80b43b26 hostname=diag245.central.sun.com last_push=10-04 22:42:26 local_event_log=Y log_months= logfile=/var/adm/messages mailer=/usr/bin/mail max_event_file_size=4 ping= pro.http.ip=http://system:2000/nscc30/InsertService pro.http.timeout=90 pro.netconnect.active=N pro.netconnect.maxsize=2000 pro.srs.active=N pro.srs.ip=192.9.228.81 pro.srs.ip2= production=Y

Password Maintenance

Use the Password Maintenance window to change the security password for the Storage Automated Diagnostic Environment GUI.

• Click Ras password in the System window.

Password Maintenance	
User : ras	
Enter OLD password : 🎽	
Enter NEW password (min 4 chars.) :	
	Update

- ▼ To Change the Security Password
 - 1. You must know the current password. If you do not know the current password, use the save_password(1M) utility to create a new password.

Note – The default login and password after initial installation are ras/agent (all lowercase).

- 2. Enter the OLD password.
- 3. Enter a new password, with a minimum of four characters.
- 4. Click Update.

The security password is changed.

Update User Roles

The Storage Automated Diagnostic Environment administrator can assign permission privileges for the categories listed in TABLE 7-1. In addition to assigning privileges, the administrator can add a new user, maintain passwords, and update or remove existing users.

Update User Roles

[<u>Help</u>]

Use this screen to add new user to your system,check at least one privilege for each user. Remove the password and click [Add/Update] to remove that user.

Privilege
Description

user Cannot make changes in the maintenance functions.

admin this privilege is required to add/delete/update in the Maintenance section.

test required to run tests.

script This privilege is required to run scripts. Sun StorEdge 3900/6900 Series configuration scripts are available. expert required to use the FC Analysis section.

User/	ser/Role maintenance							
#	User	Name	Password	user	admin	script	test	expert
1	dan	Dan Atherton	Į*****				¥	
2	david	David Lloyd				v		
з	dottie	Dottie Rawsky			V			
4	gavin	Gavin Gibson						•
5	glen	Glen Grantham	Į*****	¥				
6	todd	Todd McKenney	Į****	•	V	•	R	V
add	Ι	Ĭ	Ĭ					
add	I	Ĭ	Ĭ					
add	I	Ĭ	I					
add	I	Ĭ	I					
add	Ĭ	Ĭ	I					

Add/Update

TABLE 7-1 User Privilege	es
--------------------------	----

Privilege	Description	
user	Cannot make changes in the maintenance functions	
admin	Required to add, delete, or update in the Maintenance section	
test	Required to run tests	
script	Required to run scripts.	
expert	Required to use the FC Analysis section.	

Note – To remove a user, the administrator must delete the encrypted password and click Add/Update.

Clear Login

The Clear Login screen, which requires authorization, enables the user to login as a different user without having to exit from the browser.

Note – You must press Cancel to the "Question Authorization Failed Retry?" question and reload the browser to return to the login screen.

Sun StorEdge™ 3900 and 6900 Series Configuration

The hardware components for the Sun StorEdge 3900 and 6900 series include the Storage Service Processor, customer management connection, Sun StorEdge T3+ arrays, Fibre Channel switches, and a Sun StorEdge expansion cabinet. In addition, the Sun StorEdge 6900 series has a virtualization engine. These components can be displayed, configured, verified, and managed using the Sun StorEdge Configuration functionality.

The configuration options for the various components of the Sun StorEdge 3900 and 6900 series are defined in the following pages. For more information on the SUNWsefcg utility and how to use the options, please refer to the *Sun StorEdge 3900 and 6900 Series Reference Manual*.

Note – To utilize the Sun StorEdge 3900 and 6900 series configuration functionality, the machine must be registered in the Storage Automated Diagnostic Environment as a Sun StorEdge 3900 and 6900 series solution.

▼ To Access the Sun StorEdge 3900 and 6900 Series Configuration Window

1. Click the Utilities link in the Storage Automated Diagnostic Environment main window.



2. Click the Sun StorEdge 3900 and 6900 Series Configuration link.

The Sun StorEdge 3900 and 6900 Series Configuration window is displayed.

Sun StorEdge 3900/6900 Series Configuration					
Warni	ng: This machine does not appear to be a Sun StorEdge 3900/6900 Series.				
Select Host:	bradster.central.sun.com (Local) 💷 Change				
Sun StorEdge (Unknowr	n Model) System				
Sun StorEdge T3+ Array	/ View/Configure				
Display Config	Display Sun StorEdge T3+ Array Configuration				
Save Config	Save Sun StorEdge T3+ Array Configuration				
<u>Configure</u>	Configure Sun StorEdge T3+ Array(s)				
<u>Modify Parameters</u>	Modify Sun StorEdge T3+ Array System Parameters				
Restore	Restore Sun StorEdge T3+ Array Configuration				
Verify	Verify Sun StorEdge T3+ Array Configuration				
Configuration Status	Display Sun StorEdge T3+ Array Configuration Status				
<u>Manage Password</u>	Store/Clear Persisted Sun StorEdge T3+ Array Password				
Sun StorEdge network F	C switch-8 and switch-16 View/Configure				
Display Config	Display Sun StorEdge network FC switch Configuration				
<u>Verify</u>	Verify Sun StorEdge network FC switch Configuration				
<u>Set Confiq</u>	Set Sun StorEdge network FC switch Configuration				
Set switch Port SL M	ode Set Sun StorEdge network FC switch Port SL Mode Per Sun StorEdge T3+ Array (Sun StorEdge 3900 Series Only)				
Download Firmware	Download Sun StorEdge network FC switch Firmware				
Virtualization Engines					
Manage Disk Pools	Manage Disk Pools				
Manage VLUNs	Manage Virtual LUNs				
Manage Zones	Manage Virtualization Engine Zones				
Manage ConfigFiles	Manage Configuration Files				
<u>Manage Hosts</u>	Manage Virtualization Engine Hosts				
<u>View Map</u>	View Virtualization Engine Map				
Manage slicd Daemo	<u>n</u> Manage Virtualization Engine Daemon				
Manage Password	Store/Clear Persisted Virtualization Engine Password				
Log Functions and Sun	StorEdge 3600/6900 Series Maintenance				
View Log	View Sun StorEdge 3900/6900 Series Log File				
View Errors	View Sun StorEdge 3900/6900 Series Errors				
Remove Locks	Remove Lock Files				

FIGURE 7-2 Sun StorEdge 3900 and 6900 Series Configuration

Sun StorEdge 3900 and 6900 Series Configuration Options

Sun StorEdge T3+ Array View/Configure

The Sun StorEdge T3+ array menu presents the following options:

Display Config Reads and displays configuration information for the specified Sun StorEdge T3+ array partner group. It includes output from the following Sun StorEdge T3+ array commands: • version • sys list • vol list • vol stat • port list port listmap Save Config Updates the Sun StorEdge T3+ array snapshot file after reading the array configuration. This information is saved to be reused if needed. Configure Steps you through a series of submenus that enable you to reconfigure Sun StorEdge T3+ arrays. After successful update of the configuration, the information is saved to the snapshot file. Supported RAID types include: • RAID 1 • RAID 5 • RAID 1 + RAID 5 In the RAID 1 + RAID 5 mixed-RAID types, each Sun StorEdge T3+ array has one RAID 1 volume consisting of three physical drives and a RAID 5 device consisting of five physical drives, plus a spare. **Modify Parameters** Modifies a Sun StorEdge T3+ array's system parameters. Restore Restores the Sun StorEdge T3+ array configuration with the static saved configuration from the snapshot file. This is a function that can be used in the event you lose a LUN or you must replace a Sun StorEdge T3+ array. Verify Checks and verifies that the current Sun StorEdge T3+ array configuration is the same as the most recently-saved version of the configuration in the snapshot file. Configuration Status Displays the Sun StorEdge T3+ array(s) that are being configured. Manage Password Store or clears a persistent Sun StorEdge T3+ array telnet password.

 TABLE 7-2
 Sun StorEdge T3+ Array Configuration Options

```
/etc/syslog.conf
```

```
# messages to syslogd on another host
#*.info @stradic
# *.warn @192.29.207.216
# *.warn @192.29.207.241
# *.warn @192.29.207.242
# *.warn @192.29.207.208
# messages sent as SNMP traps
# *.warn | snmp_trap tomm
```

FIGURE 7-3 Example #1 of a Sun StorEdge T3+ Array Configuration

```
/etc//hosts
# Internet host table
#
# 127.0.0.1 localhost
# 10.4.46.224 isa-224 loghost
```

FIGURE 7-4 Example #2 of a Sun StorEdge T3+ Array Configuration

```
# /etc/syslog.conf
```

```
local7.notice /var/adm/messages.t3
```

```
# touch /var/adm/messages.t3
```

```
# ps -ef | grep syslog
```

kill -HUP (pid)

```
# edit /etc/fcswitch.conf
# Sample configuration file for Qlogic Switches
#
# Enter switch IP address and name
# Note: The name is optional
#
# 10.0.0.1 switch1
# 10.0.0.2 switch2
# 10.4.46.229 isa-229
# 10.4.46.230 isa-230
# 10.4.46.231 isa-231
# 10.4.46.232 isa-232
# 10.4.46.233 isa-233
# 10.4.46.234 isa-234
```

Sun StorEdge Network FC Switch-8 and Switch-16 Switch View/Configure

The Switch menu presents the following options:

 TABLE 7-3
 Sun StorEdge Network FC Switch-8 and Switch-16 Configuration Options

Display Config	 Displays the specified switch's configuration, including the following: IP address configuration Number of ports Version information Port status and type Zones
Verify	Verifies that the configuration of the specified switch matches the predefined static configuration for that switch, based on the cabinet type (Sun StorEdge 3900 series or Sun StorEdge 6900 series).
Set Config	Configures the specified switch to a predefined static configuration, based on cabinet type and location (top or bottom switch for Sun StorEdge 6900 series).
Set Switch Port SL Mode	Sets the switch SL mode for a Sun StorEdge T3+ array (the StorEdge 3900 Series only).
Download Firmware	Downloads the flash code to the Sun StorEdge network FC switch-8 or switch-16 switch and resets it. The flash code is automatically loaded from the appropriate file from the /usr/opt/SUNWsmgr/flash directory.

Virtualization Engines

The virtualization engines configuration menu presents the following options:

TABLE 7-4	Virtualization	Engine	Configuration	Options
-----------	----------------	--------	---------------	---------

Manage Disk Pools	 Enables you to create and remove disk pools that map to physical LUNs on the specified Sun StorEdge T3+ array. These are the physical LUNs that can be used to creat VLUNs. The options on the submenus include: Create Disk Pools Remove Disk Pools
Manage VLUNs	Enables you to create and remove VLUNs. The options on the submenus include: • Create VLUNs • Remove VLUNs
Manage Zones	 Presents the following options: Create virtualization engine zones Add components to virtualization engine zones Delete components from virtualization engine zones Remove virtualization engine zones
Manage Config Files	 Presents the following options: View virtualization engine map Save virtualization engine map Restore virtualization engine map Verify virtualization engine map
Manage Hosts	Manages the virtualization engine host
View Map	Displays the current configuration of the specified virtualization engine
Manage slicd Daemon	Manages the virtualization engine daemon
Manage Password	Stores or clears the virtualization engine telnet password

Log Functions and Sun StorEdge 3900 and 6900 Series Maintenance

You can review the content of the Sun StorEdge 3900 and 6900 Series Configuration log files. Log entries are displayed from the end of the file going back; the most recent entries are displayed last.

View Log

The View Log option enables you to look at all the logs containing messages about the configuration actions taken.

Sun StorEdge Series Config (on msp0.central.sun.com)		
<u>Sun StorEdge Series Config</u> > View Logs		
StorEdge 6900 Logs		
Tue Jan 29 15:45:59 MST 2002 postinstall all INFO Package SUNWsecfg 1.1.10 has been installed.		
Tue Jan 29 15:47:30 MST 2002 getcabinet: all INFO found VEs and 8 port switch, type is 6910.		
Tue Jan 29 15:47:33 MST 2002 getcabinet: all INFO found VEs and 8 port switch, type is 6910.		
Tue Jan 29 15:47:39 MST 2002 getcabinet: all INFO found VEs and 8 port switch, type is 6910.		
Tue Jan 29 15:47:40 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd Telnet New .		
Tue Jan 29 15:47:40 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd telnet open .		
Tue Jan 29 15:47:40 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd login .		
Tue Jan 29 15:47:40 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.		
Tue Jan 29 15:47:41 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd Telnet New .		
Tue Jan 29 15:47:41 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd telnet open .		
Tue Jan 29 15:47:41 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd login .		
Tue Jan 29 15:47:41 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd Telnet close .		
Tue Jan 29 15:47:51 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd Telnet New .		
Tue Jan 29 15:47:52 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd telnet open .		
Tue Jan 29 15:47:52 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd login .		
Tue Jan 29 15:47:52 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.		
Tue Jan 29 15:47:53 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd Telnet New .		
Tue Jan 29 15:47:53 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd telnet open .		
Tue Jan 29 15:47:53 MST 2002 checkt3pw: t3b0 INFO : Executing T3+ cmd login .		
View Errors

The View Errors option enables you to look at all the errors that have occurred on a given host. The errors are sorted by date, the most recent are displayed last.

Sun StorEdge Series Config (on msp0.central.sun.com)	[<u>Help</u>]
Sun StorEdge Series Config > View Errors	
StorEdge 6900 Errors	
Tue Jan 29 15:47:40 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Tue Jan 29 15:47:52 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Fri Feb 8 16:49:40 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Fri Feb 8 16:49:48 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Fri Feb 8 16:50:23 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Fri Feb 8 16:50:27 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Fri Feb 8 17:51:27 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Fri Feb 8 17:51:35 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Fri Feb 8 17:52:04 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Fri Feb 8 17:52:08 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Mon Feb 11 10:38:44 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Mon Feb 11 10:39:02 MST 2002 checkt3pw: t3b0 login ERROR : login failed: bad name or password.	
Mon Feb 11 10:39:35 MST 2002 showt3: t3b0 login ERROR : login failed: bad name or password.	

Remove Locks

Using the Remove Locks functionality, you can remove lock files on the Sun StorEdge network FC switch-8 and switch-16 switch, the virtualization engines, and/or the Sun StorEdge T3+ array.



Note – Use the Remove Locks function only if you know the device *is not* in use, but the Sun StorEdge 3900 and 6900 Series Configuration shows that the device *is* in use.

Storage Automated Diagnostic Environment Help

This chapter explains the GUI online help and the command-line utilities help associated with the Storage Automated Diagnostic Environment.

Site Map

The following site map shows the Storage Automated Diagnostic Environment and its functionality at a glance. Online Help is available for each topic.



FIGURE 8-1 Storage Automated Diagnostic Environment Site Map

▼ To Access the Online Help

1. Click the Help link in the Storage Automated Diagnostic Environment main window.



2. Click the Help Page link on the Help menu in the left frame.

The Storage Automated Diagnostic Environment Help window is displayed.

Storage Au	Itomated Diagnostic Environment Help	[<u>Help</u> <u>Toc</u> <u>Index</u>]
This is the Storag	e Automated Diagnostic Environment Help Page, select from one of the following sectio	ns.
Help Sections		
Maintenance		
– <u>General</u> – <u>Topology</u>	Used primarily to maintain site, host, device, email and notification providers informat Create, Merge and Review Topologies.	tion.
Monitor		
- <u>Devices</u> - <u>Topology</u> - <u>Log</u>	Review monitored devices, recent alerts, report and Log information. Monitor Topologies and review device state. Review the Event, Alert and System log	
Diagnose		
- <u>Tests</u> - <u>Tools</u>	Run Diagnostic Tests Run Revision Checking and FC Check	
Reports		
- <u>General</u>	Run Reports.	
Utilities		
– <u>System</u> – <u>Scripts</u>	General Utilities functions. Run Scripts	

FIGURE 8-2 Storage Automated Diagnostic Environment GUI Online Help

3. Click the topic for which you need information from the Help Sections list.

Storage Automated Diagnostic Environment Event Grid

The Event Grid shows all the actionable and non-actionable events Storage Automated Diagnostic Environment can generate.

- ▼ To Customize an Event Report
 - 1. Click the Event Grid link on the Help menu.
 - 2. Select the criteria from the Storage Automated Diagnostic Environment event grid, like the one shown in FIGURE 8-3.

Event	Grid						[<u>Help</u>]
Select a C [ReportFo	Category/Con ormat1 to disp	nponent/Event llav a Report fo	Type and type [GO] to I ormat. Click Info/Action	imit the	e report. iew. (Ever	Click on the Columns hea tGrid)	ders to change the sort. Check
Categor	Category: Component:			Event	Type:		
All			All 🗆			All 🗆	ReportFormat 🗌 GO
+ <u>Cat</u>	<u>Comp.</u>		<u>EventType</u>	Sev.	Action	Description	
a3500fc	enclosure		Audit		•	[<u>Info</u>] Auditing a new a ID:a3500fc.da106_004	3500fc with
a3500fc	ib		Comm_Established			Communication regained	with A3500fc called DPL4
a3500fc	ib		Comm_Lost	U	Y	[<u>Info/Action</u>] Lost com DPL4	munication with A3500FC called
a3500fc	a3500fctes	t	DiagnosticTest-			a3500fctest (diag240) or	n a3500fc.da106_004 failed
a3500fc	enclosure		Discovery			[<u>Info</u>]Discovered a new ID:a3500fc.da106_004	v a3500fc with
a3500fc	controller		InsertComponent			[<u>Info</u>]Component 'cont da106_004 'a3500fc.da1	roller.b:XX' was added to 06_004
a3500fc	disk		InsertComponent			[<u>Info</u>]Component 'disk added to da106_004 'a35	.1:SEAGATE.LT02926900' was 500fc.da106_004'
a3500fc	enclosure		LocationChange			Location of a3500fc calle	d DPL4 was changed
a3500fc	controller		RemoveComponent		Y	[<u>Info/Action</u>] Compon from da106_004 'a3500f	ent 'controller.b:XX' was removed c.da106_004
a3500fc	disk		RemoveComponent		Y	[Info/Action] Compone was removed from da106	ent 'disk.1:SEAGATE.LT02926900' 5_004 'a3500fc.da106_004'
a3500fc	disk		StateChange+		• •	disk.0 (SEAGATE.LT04183 (a3500fc.da106_004) is r changed from 'Degraded'	3900) in A3500FC 'da106_004' now Available (status–state to 'Optimal')
a3500fc	controller		StateChange-		Y	[<u>Info/Action</u>] controlle A3500FC 'da106_004' (a3 Not–Available (status–sta	r.a (Symbios.1T93200466) in 3500fc.da106_004) is now ite changed from 'Active' to 'Error')
a3500fc	disk		StateChange-		Y	[<u>Info/Action</u>] disk.0 (SI 'da106_004' (a3500fc.da (status-state changed fro	EAGATE.LTO4183900) in A3500FC 106_004) is now Not–Available m 'Optimal' to 'Degraded')

FIGURE 8-3 Storage Automated Diagnostic Environment Event Grid

3. For more diagnostic information, click the <u>Info/Action</u> link from the Description field.

A pop-up menu appears, displaying the information for that event and the action recommended for problem resolution.

Note – The Storage Automated Diagnostic Environment Event Grid is intended to be used interactively, but it is also printable. Before printing the Event Grid, run the page with the ReportFormat checkbox enabled; to see all the details about each event. Otherwise, you must click the **[details]** link to see details, one event at a time.

Storage Automated Diagnostic Environment Architecture

1. Click Architecture on the Help menu.

The Storage Automated Diagnostic Environment Architecture Diagram, as shown in FIGURE 8-4, is displayed.

2. For details of the component, move your mouse over the section within the diagram, or refer to TABLE 8-1.



FIGURE 8-4 Storage Automated Diagnostic Environment Architecture Diagram

Slave Agent	 Each Slave Agent includes the same functionality as the Master Agent: Tests Instrumentation Event Generation Slave Agents are scheduled from the cron. They generate CIM events that are transmitted to the Master.
Notification Providers	 HTTP—Sends HTTP calls to an HTTP server and transfers CIM data in the appropriate format. Internal to Sun only. NSM—The Network Storage Manager (NSM) provider sends events to the NSM console. NetConnect—A common information model (CIM) provider that requests information, converts the information to the appropriate format, and relays it to NetConnect. SRS—The Sun Remote Services (SRS) provider accepts and sends information in xml format. SNMP Traps—Enables the Storage Automated Diagnostic Environment to send traps for all actionable events that occur during monitoring to external management systems. RSS—Uses modem technology with Unix-to-Unix Communication Protocol (UUCP). RSS software is required on the host and must be configured accordingly. <i>Local Provider (Email)</i>—The local provider can email events to administrators. Events can be filtered per administrator using event- type, severity level, or device-type filters.
Browser UI	The user interface (UI) uses HTML browsers. Using the UI, administrators can configure the agents, monitor storage devices, review the topology, execute diagnostic tests, and verify the configuration.
Storage Automated Diagnostic Environment Services	 The services are the core of the framework. They provide logic and persistence to all agents, monitors, and user interface functions. The services include: A database of current instrumentation reports All CIM schemas (mof files) required to generate events The current configuration of all agents Topology functions The current state of each storage device A database of diagnostic processes Logic and persistence for timers, threshold, transitions, and revision matrix

 TABLE 8-1
 Storage Automated Diagnostic Environment Component Definitions

Health Monitors	Health monitors read instrumentation reports and generate CIM events that are stored and sent to Notification providers. Events are generated using a cache database that stores previous reports, timers, and thresholds.	
Diagnostic Modules	Diagnostic tests can be executed locally or remotely to test different components.	
Instrumentation Agents	Instrumentation agents probe storage devices and monitor log files to generate detailed reports on the state of each component of the device. Agents are scheduled to execute by crons. Instrumentation reports are stored and compared by the health monitors to generate CIM events.	

TABLE 8-1	Storage Autor	nated Diagno	stic Environm	ient Componei	nt Definitions

Utilities

The explanations of the various diagnostics tests associated with the Storage Automated Diagnostic Environment, shown in FIGURE 8-5, are available from the command line. These utilities also have man(1M) pages.

Storage Automated Diagnostic Environment Diagnostics Tests

The Storage Automated Diagnostic Environment diagnostics tests are defined in TABLE 8-2 and are located in /opt/SUNWstade/Diags/bin. Refer to the man pages for more detail.

Storage Automated Diagnostic Environment Diagnostics / Exercisors				
All diagnostics are located in /opt/SUNWstade/Diags/bin, refer to the man pages for more details.				
Diagnostic	Description			
a3500fctest	In-band diagnostic for Sun StorEdge A3500FC LUNs attached via the host data-path.			
a5ksestest	In-band diagnostic for Sun StorEdge A5000 Enclosures attached via the host data-path.			
a5ktest	In-band diagnostic for Sun StorEdge A5000 disk drives attached via the host data-path.			
fctapetest	In-band diagnostic for FC tape drives attached via the host data-path.			
ifptest	Sun StorEdge QLC 2100 HBA diagnostic for HBAs using the ifp driver			
qlctest	Sun StorEdge QLC 2200 HBA diagnostic for HBAs using the qlc driver.			
socaltest	Sun StorEdge Soc+ HBA diagnostic for HBAs using the socal driver.			
linktest	The linktest is designed to test two end points of a FC link segment and assist the user in FRU isolation of the link segement. This test is typically run when the FCAL topology indicates a link problem. i.e. A red link.			
switchtest	Out-of-band diagnostic for testing ports on Sun StorEdge Network FC switch-8 and switch-16.			
t3test	In—band diagnostic for Sun StorEdge T3 and T3+ LUNs attached via the host data—path.			
t3ofdg	Out-of-band diagnostic for Sun StorEdge T3 and T3+ LUNs attached via the an enet connection. t3ofdg will test all of the FRUs within the T3 enclosure			
t3volverify	Out-of-band diagnostic for Sun StorEdge T3 and T3+ LUNs attached via the an enet connection. t3volverify will execute the volume verify function on the selected T3.			
vediag	Out—of—band diagnostic for Virtualization Engine (VE) attached via the an enet connection. Instantiates and monitors diagnostics on the virtualization engine.			
veluntest	In—band diagnostic for Virtualization Engine VE attached storage. Verifies the functionality of the in—band virtualization engine Sun StorEdge 3900/6900 Series LUN(s).			
Exercisors				
dex	In-band exercisor for all disk drives attached via the host data-path.			
lbf	In-band exercisor for FC loops(media) utilizied by the host Sun StorEdge HBAs soc+ and QLC 2200.			

FIGURE 8-5 Storage Automated Diagnostic Environment Commands

Diagnostics Utilities

TABLE 8-2 Storage Automated Diagnostic Environment Diagnostic Commands

a3500fctest(1M)	Verifies the functionality of the Sun StorEdge A3500FC array using the two subtests provided
a5ksestest(1M)	Provides configuration verification, fault isolation, and repair validation of Sun StorEdge A5200 array
a5ktest(1M)	Verifies the functionality of the Sun StorEdge A5200 array
brocadetest(1M)	Verifies the functionality of the Brocade Silkworm switch ports
fctapetest(1M)	Tests the Fibre Channel tape drive
ifptest(1M)	Tests the functionality of the Sun StorEdge PCI FC-100 host adapter board.
qlctest(1M)	Tests the functions of the Sun StorEdge PCI Dual Fibre Channel host adapter board.
socaltest(1M)	Validates and performs fault isolation on the Sun StorEdge SBus FC-100 host adapter board.
linktest(1M)	Tests two end points of a Fibre Channel link segment and assists in FRU isolation of the link segment. This test is typically run when the FCAL topology indicates a problem.
switchtest(1M)	Diagnoses the Sun StorEdge network FC switch-8 and switch-16 switches
t3test(1M)	Verifies the functionality of the Sun StorEdge T3 and T3+ array
t3ofdg(1M)	Tests all of the FRUs within the Sun StorEdge T3 and T3+ array enclosure. The $t3ofdg$ is an out-of-band diagnostic test for Sun StorEdge T3 and T3+ array LUNs attached through an Ethernet connection.
t3volverify(1M)	Out-of-band diagnostic for T3 and T3+ LUNs attached through an Ethernet connection. t3volverify executes the volume verify function on the selected Sun StorEdge T3 and T3+ array.
vediag(1M)	Out-of-band diagnostic for the virtualization engine attached through an Ethernet connection. Instantiates and monitors diagnostics on the virtualization engine.
veluntest(1M)	In-band diagnostic for virtualization engine attached storage. Verifies the functionality of the in-band virtualization engine LUN(s).

Storage Automated Diagnostic Environment Agent Utilities

The Storage Automated Diagnostic Environment Agent utilities that are defined in TABLE 8-3 are located in the /opt/SUNWstade/bin directory. Refer to the man pages for more detail.

Storage Autom	ated Diagnostic Environment utilities
All utilities are located i	in /opt/SUNWstade, see the man pages for more details
Program	Description
bin/checkcron	If executed from the command line, this executable program verifies that the executable program "rasagent" is entered in the crontab(1M) file. Note: You must be root to run this program.
bin/clearcache	If executed from the command line, this executable program will remove the instrumentation information for all the configured devices from the CACHE. This is typically used during testing purposes so that a DiscoveryEvent can be forced to occur upon the next execution of the Storage Automated Diagnostic Environment Agent.
disk_inquiry	Used by Storage Automated Diagnostic Environment to discover devices on the data–path that are using SCSI commands.
bin/ras_install	This program will set—up the HTTP service, add a cron and register with the master Storage Automated Diagnostic Environment Agent in the case of a slave install. It must be run manually upon completing the 'pkgadd'. This program can be executed anytime after the initial execution and setup when a modification to the cron is desired or to change the Master/Slave title of a Host agent. Refer to the User's Guide before making changes.
bin/rasagent	This is the executable program that calls the modules for the different Network Storage Devices supported by this monitoring agent. It is executed automatically by cron or can be run manually from the command line. Keep in mid however, if run from the command line, rasagent will first check that another process for rasagent is not already running.
bin/save_password	Utility used by the system administrator to change the password to Storage Automated Diagnostic Environment's graphical user interface. Changing the password can also be performed in the GUI itself. However this utility would be used from the command line when the administrator forgot the password.
	Usage: save_passworu < passworu >
DINYSANDOX	Utility used by Storage Automated Diagnostic Environment to gather Fibre Channel Switch Information.
bin/testt3	verify that the tokens of a Specific 13 are working. Is token files etemprop.html and sysprop.html must be available for the Storage Automated Diagnostic Environment probing capabilities to work properly. Running this utility from the command will require that the IP Address of the T3 being probed is known.
	Usage: testt3 <ip_address></ip_address>

FIGURE 8-6 Storage Automated Diagnostic Environment Agent Commands

TABLE 8-3 provides descriptions for commands available for the Storage Automated Diagnostic Environment Agent functions.

checkcron(1M)	Verifies whether the Storage Automated Diagnostic Environment main program is entered in the crontab(1M) file
clearcache(1M)	Clears the Storage Automated Diagnostic Environment cache files that contain the current report for each device being monitored
disk_inquiry(1M)	Identifies devices on the data path that are using SCSI commands
rasagent(1M)	Calls the modules for network storage devices supported by Storage Automated Diagnostic Environment. It is automatically executed by cron, or it can be run manually from the command line.
ras_install(1M)	Sets up the HTTP service, adds a cron, and registers with the master agent during a slave install. It must be run manually upon executing the pkgadd command.
sanbox(1M)	Displays Fibre Channel switch information
save_password(1M)	Changes the password to the Storage Automated Diagnostic Environment GUI. This function can also be performed in the GUI itself. save_password is usually used if the system administrator forgets the password.
testt3(1M)	Retrieves tokens from a Sun StorEdge T3 and T3+ array. It verifies whether the IP address used is correct and whether the IP address points to a Sun StorEdge T3 and T3+ array that can provide tokens.

 TABLE 8-3
 Storage Automated Diagnostic Environment Agent Commands

Index

SYMBOLS

/opt/SUNWrasag directory, 12 /opt/SUNWrasag/DATA/rasagent.conf, 27 /var/adm/messages, 12 on local host, 85

Α

a3500fctest command, 104 a5ksestest command, 107 a5ktest command, 109 test modes, 109 add new hosts required fields, 35

С

cache erasing, 148 checkcron(1M), 175 checklist installation, 10 clearcache(1M), 175 configuration updating master, 38 configuration recovery, 151 customer information maintenance of, 7

D

device maintenance of device-specific information, 7 disk_inquiry(1M), 175 documentation related, xv documentation conventions typographic, xiii

Е

email configuration, 150 erase cache, 148 event report how to customize, 167 events, categorization of, 7 executing StorTools tests, 91

Η

host maintaining, 36 updating, 38

I

ifptest, PCI FC_AL card test
 options, 114
installation

checklist, 10 on multiple hosts, 12 requirements, 12 verification of, 16

L

local notification, 7 login screen clearing of, 154

Μ

maintain hosts functions of push slave configurations, 37 slaves, add new required fields, 42

Ν

Network Storage Agent configuration file, 27 distributed slave(s), 24 graphical user interface (GUI), 7 starting, 28 Network Storage Agent User's Guide description of, xi notifications add notification, 54 create local email, 54 local, 7 local email/pager, 53

0

online help how to access, 166 use of, 139, 145, 165 organization of book, xii

Ρ

password security, 152 passwords how to change, 152 PCI FC-100 test ifptest, 113 permission privileges assignment of, 153 pkgadd -d command, 14

Q

glctest command, 115

R

ras_install(1M), 175 rasagent(1M), 175 requirements, installation, 12

S

sanbox(1M), 175 save_password(1M), 152, 175 **SE-series** log functions, 162 virtualization engine (VE) configuration, 161 shell prompts, xiv slave adding, 40 push configs, 43 slave setup, 25 socaltest command, 118 socaltest, Sun StorEdge SBus FC-100 Host Adapter board test options, 118 statistics how to check, 140 **StorADE** agent commands, 174 architecture diagram, 169 component definitions, 170 diagnostic commands, 173 event grid, 167

reports, 139 running manually, 149 StorTools user interfaces, 91 Sun StorEdge 3900 and 6900 series hardware components, 155 Sun StorEdge RASAgent installation, 9 Sun StorEdge SBus FC-100 host adapter card test, socaltest, 118 Sun StorEdge T3 array messages monitoring, 46 switch data to view FC counters, 143 graph, 142 switchtest command, 121, 123 options, 121, 123 system time-outs, 147

Т

t3test command, 127 subtests, 127 tests StorADE diagnostics, 172 testt3(1M), 175 thresholds viewing, 141 time-out settings how to change, 147

U

user privileges, 153 utilities system, 145

W

window event log messages on local host, 87 maintain devices, 47 maintain hosts, 36